Western University Scholarship@Western

Electronic Thesis and Dissertation Repository

8-1-2018 2:00 PM

Examining geographic variation in children's perceived barriers to physical activity and the implications on behaviour

Leah Gabrielle Taylor, The University of Western Ontario

Supervisor: Gilliland, Jason, *The University of Western Ontario* A thesis submitted in partial fulfillment of the requirements for the Master of Science degree in Geography © Leah Gabrielle Taylor 2018

Follow this and additional works at: https://ir.lib.uwo.ca/etd

Part of the Human Geography Commons, and the Public Health Education and Promotion Commons

Recommended Citation

Taylor, Leah Gabrielle, "Examining geographic variation in children's perceived barriers to physical activity and the implications on behaviour" (2018). *Electronic Thesis and Dissertation Repository*. 5489. https://ir.lib.uwo.ca/etd/5489

This Dissertation/Thesis is brought to you for free and open access by Scholarship@Western. It has been accepted for inclusion in Electronic Thesis and Dissertation Repository by an authorized administrator of Scholarship@Western. For more information, please contact wlswadmin@uwo.ca.

Abstract

Low levels of physical activity among Canadian children has become a national public health issue. Recent research has suggested that children's physical activity levels are associated with their perceptions of their everyday environments. A better understanding of the formation of these perceptions within different contexts is needed to explain the extent of the relationship. Using a multi-tool quantitative protocol, this thesis examines geographic variation in socio-ecological factors influencing children's perceptions of barriers to PA, and the extent to which perceptions mediate the relationship of the environment and PA. Results indicate that perceptions form within contexts, and have an influence on PA. The studies take place in Northwestern and Southwestern Ontario. This research provides a starting point for future research, policy, and practice to consider how structures of children's environments determine experiences of PA, suggesting a new way to conceptualize behaviour to determine effective strategies for improving children's PA and overall health.

Keywords

Children; physical activity; urbanicity; rural; accelerometer; socio-ecological model; structural equation modeling; geography

Co-Authorship Statement

Each integrated article within this thesis will be submitted for publication in peerreviewed journals. Chapters 4 and 5 were written by Leah Taylor as the primary author in both cases, performing data collection in Northwestern Ontario, statistical analysis, and writing in each article. In both cases, Dr. Jason Gilliland designed the original STEAM study. In Chapter 4, Dr. Jason Gilliland and Dr. Andrew Clark are co-authors and were involved in the development of analysis procedures. In Chapter 5, Dr. Jason Gilliland, Dr. Andrew Clark, Dr. Piotr Wilk, and Brenton Button are co-authors. Dr. Jason Gilliland and Dr. Andrew Clark were involved in the development and implementation of analysis procedures. Dr. Piotr Wilk was involved in the development and implementation of the structural equation modeling for this paper. Brenton Button was the team leader on data collection in Northwestern Ontario for the 2016 STEAM project and provided expertise on accelerometer data analysis methods and literature for this paper. Below are the journal targets for both integrated articles.

Chapter 4: Taylor, L.G., Clark, A.F., & Gilliland, J.A. Context Matters: Examining children's perceived barriers to physical activity across varying Canadian environments. Submitted to *Health and Place*.

Chapter 5: Taylor, L.G., Clark, A.F., Wilk, P., Button, B., & Gilliland, J.A. Exploring the effect of perceptions on children's physical activity in varying geographic contexts: a structural equation modelling approach. Prepared for *Social Science and Medicine*.

Acknowledgments

This research would not have been possible without the financial support provided by the HEAL, the Department of Geography, the Children's Health Research Institute, and Children's Health Foundation. I would also like to extend my gratitude to the residents of Northern Ontario, for welcoming us into your homes and communities.

To Dr. Jason Gilliland, I am so grateful our paths converged, leading me to become a geographer. The opportunities I have been afforded, and everything I have learned over the past two years would not have been possible without your hard work, community involvement, positivity, and ongoing encouragement.

To Dr. Andrew Clark, this thesis would not be what it is without your hard work, countless late nights with our data, and answers to my endless questions. I cannot express how thankful I am for your patience, knowledge and guidance.

My friends at the HEAL, you have all been instrumental in my success, and I will always cherish the memories of our time together. Thank you to Dr. Piotr Wilk, Dr. Stephanie Coen, and Brenton Button, who shared their wealth of knowledge, providing me with direction and assistance throughout my thesis. Bridget, I'm so glad I found an amazing friendship with you. Thank you for our laughs, lunch dates, and all your help with stats.

To my friends, especially Jenna, Emily, Brandon, Taylor, Sam, and Shannen. You have all been there for me every step of the way. I couldn't have done this without each of you.

Mum and Dad (my #1 cheerleaders), Rachiepants, A\$, and E-Dog. If not for you unending love and encouragement, I would never have reached this achievement. Karen and Al Roos, thank you for your ongoing support and always making time to listen. I am so lucky to call all of you my family.

Rob: I owe all my success over the last two years to the love and support of my best friend. I can't express how thankful I am for everything you do for me. Little did I know on my first day of school 6 years ago, I'd encounter the most important person in my life in Bio 1201A. So, thank you also to Western, for your first year requisite courses.

Table of	Contents
----------	----------

Abstracti
Co-Authorship Statementii
Acknowledgmentsiii
Table of Contents iv
List of Tables
List of Figuresix
List of Appendices
Chapter 1 1
1 Introduction
1.1 Research Context
1.2 Theoretical Context
1.3 Research Objectives and Questions
1.4 Thesis Format7
1.5 References
Chapter 2
2 Literature Review
2.1 Search Strategy
2.2 Physical Activity and the 24-Hour Movement Guidelines
2.2.1 Total Physical Activity17
2.2.2 Light Intensity Physical Activity
2.2.3 Moderate to Vigorous Physical Activity
2.2.4 Sedentary Behaviour
2.2.5 Outdoor Time
2.3 Socio-Ecological Perspective to Physical Activity Environments

2.4	Defining Environments	20
	2.4.1 Understanding "Rural"	20
	2.4.2 Defining Urbanicity and the Population Density Spectrum	21
2.5	Factors Influencing Physical Activity across Environments	25
	2.5.1 Barriers and Facilitators to Physical Activity	25
2.6	Evidence found within the Socio-ecological Framework of Recreational Physical Activity	28
	2.6.1 Individual Environment	28
	2.6.2 Social Environment	29
	2.6.3 Physical Environment	29
2.7	Gaps Filled and Conclusions	32
2.8	References	34
Chapte	er 3	45
3 Dat	ta Collection; Tools and Measurement	45
3.1	Research Setting	45
	3.1.1 Sample	46
3.2	Data Collection	47
	3.2.1 Survey Tools	47
	3.2.2 Accelerometers	48
	3.2.3 GPS Devices	49
	3.2.4 Median Household Income	50
3.3	Justification for Analysis of Variables	50
	3.3.1 Procedure for Analyzing Children's Perceptions	50
	3.3.2 Changes in Analysis of Activity	51
	3.3.3 Measuring Urbanicity and the Spectrum Tools	52
3.4	Measuring the Data	54

	3.5	Refere	ences	55
Chapter 4				59
4	4 Context Matters: Measuring children's perceived barriers to physical activity across varying Canadian environments			59
	4.1	Introd	uction	59
		4.1.1	Literature Review	59
	4.2	Metho	ds	61
		4.2.1	Data Source and Population	61
		4.2.2	Dependent Variables	62
		4.2.3	Independent Variables	62
		4.2.4	Data Analysis	63
	4.3	Result	S	64
		4.3.1	Intrapersonal Factors	67
		4.3.2	Interpersonal Factors	67
		4.3.3	Physical Environment	68
	4.4	Discus	ssion	72
		4.4.1	Strengths and Limitations	75
	4.5	Conclu	usion	76
	4.6	Refere	ences	77
C	hapte	er 5		81
5	Exp geo	oloring graphic	the effect of perceptions on children's physical activity in varying contexts: a structural equation modelling approach.	81
	5.1	Backg	round	81
	5.2	Metho	ds	85
		5.2.1	Measurement Instruments	86
		5.2.2	Statistical Analysis	90

	5.3	Results	S	91
		5.3.1	Girls	
		5.3.2	Boys	95
	5.4	Discus	sion	96
		5.4.1	Strengths and Limitations	
	5.5	Conclu	isions	
	5.6	Refere	nces	101
Cł	napte	r 6		110
6	Syn	thesis		110
	6.1	Summa	ary of Studies	110
	6.2	Resear	ch Contributions	113
	6.3	Limita	tions	115
	6.4	Implica	ations for Practice	117
		6.4.1	Neighbourhood Environment Barriers to Physical Activity	119
		6.4.2	Social Barriers to Physical Activity	121
		6.4.3	Safety Barriers to Physical Activity	
	6.5	Recom	mendations for Future Research	123
	6.6	Conclu	usion and Implications	125
	6.7	Refere	nces	127
Aj	open	dices		
Cı	urricu	ulum Vi	itae	149

List of Tables

Table 2.1 Number of findings demonstrating barriers and facilitators discussed by	
urbanicity, available in the literature2	26
Table 3.1 Measuring Variables 5	54
Table 4.1 Survey questions measuring barriers, and sample distribution of responses6	55
Table 4.2 Descriptive characteristics of independent variables	56
Table 4.3 Logistic regression models examining factors related to children perceiving	
safety barriers to physical activity	59
Table 4.4 Logistic regression models examining factors related to children perceiving	
social barriers to physical activity7	0'
Table 4.5 Logistic regression models examining factors related to children perceiving	
neighbourhood environment barriers to physical activity7	/1
Table 5.1 Survey questions measuring barriers, and the corresponding themed groups8	38
Table 5.2 Results of the relationship between all variables within the model)2
Table 5.3 Results of the mediating effect and direct relationships of urbanicity and	
perceptions on MVPA9)3

List of Figures

Figure 1.1 Socio-ecological model of influences on children's physical activity levels in
rural areas4
Figure 1.2 Socio-Ecological Model of Recreational Physical Activity, Adapted from
Giles-Corti and Donovan (2002)5
Figure 2.1 Activity Continuum adapted from Tremblay et al. (2010)16
Figure 3.1 Map of STEAM Study Areas in Northwestern and Southwestern Ontario
including levels of urbanicity47
Figure 5.1 Conceptual model90
Figure 5.2 Significant positive/negative association results for model, grouped by girls 04
rigure 5.2 Significant positive/negative association results for model, grouped by gills. 94
Figure 5.3 Significant positive/negative association results for model, grouped by boys. 96

List of Appendices

Appendix A: Literature Review Documents with Corresponding Number Assigned (3
pages)
Appendix B: Literature Review Results Demonstrating Similarities and Differences
Children's Perceptions of Barriers and Facilitators to PA between the Urban and Rural
Children (2 pages)
Appendix C: Research Ethics Approval Forms for Use of Human Participants STEAM
Southwestern Ontario (Redacted)
Appendix D: Research Ethics Approval Forms for Use of Human Participants STEAM
Northwestern Ontario (Redacted)
Appendix E Research Ethics Letter of Information for Parents (3 pages) and Research
Ethics Parent Consent Form (1 page)
Appendix F: Research Ethics Child Assent Form146
Appendix G: Relevant Survey Questions from STEAM Project Assessing Perceptions
used in this Research (2 pages)

Chapter 1

1 Introduction

1.1 Research Context

The benefits of physical activity, especially moderate to vigorous physical activity (MVPA), and the physical, mental, cognitive and social health contributions for children and youth are well established (Poitras et al., 2016). Despite these known benefits of physical activity, most Canadian children are not active enough to reap the health, well-being and development benefits of participation (Colley et al., 2017). The overall physical activity levels of Canadian children were issued a grade of D+ by the *2018 Participaction Report Card on Physical Activity for Children and Youth*, because less than 39% of Canadian children and youth aged 3-17 are meeting the recommendation of an average of 60 minutes of MVPA each day (Colley et al., 2017; Larouche, Garriguet, & Tremblay, 2017; ParticipACTION, 2018).

This is of further concern because physical inactivity in childhood has been associated with long lasting chronic health risks that track into adulthood. Warburton, Nicol, and Bredin (2006) have described the available research as "irrefutable evidence of the effectiveness of regular physical activity in the prevention of chronic diseases" (p.801). These associated health risks include decreases in overall health status, cardiovascular fitness, strength and bone density; and increased risk of health concerns including chronic diseases (such as some cancers, Type 2 Diabetes, and heart disease), premature death, and all-cause mortality (Bauman, 2003; Bruner, Lawson, Pickett, Boyce, & Janssen, 2008; Galloway, 2006; Pate et al., 1999; Smith, Troped, McDonough, & DeFreese, 2015; Warburton et al., 2006). Janssen (2012) has estimated a total cost of physical inactivity in Canada at \$6.2 Billion, or 3.7% of the total national health care costs in 2009. Krueger, Turner, Krueger, and Ready (2014) suggest that a modest decrease of 1% in the number of Canadians who are inactive, can have a substantial economic (\$20.3 cumulative reduction in burden) and population health impacts. The complexity and interrelatedness

of this issue, however, has led to the need for an important change in the way we must conceptualize physical activity and the health of Canadians.

The recently released *Report on the State of Public Health in Canada 2017* has drawn attention to this change. In her opening statement, Canada's Chief Public Health Officer Dr. Theresa Tam states: "Without being aware of it, our neighbourhoods and how they are built influence how healthy we are" (P.III). This report sought to encourage dialogue in community planning and health promotion for many reasons related to the health of Canadians, one of them being physical activity. Canadian public health leaders are identifying the importance of considering the environmental determinants of health, and the opportunities related to exposure and engagement with the environment for children's physical activity (Tremblay et al., 2015). Associations between children's physical activity and their surrounding physical environments have been widely discussed. Systematic reviews examining the relationship between the physical environment and children's physical activity, indicate relationships with physical environment factors such as residential density, access to recreation facilities, land-use mix, walkability, safety structures, and neighbourhood aesthetics (Ding, Sallis, Kerr, Lee, & Rosenberg, 2011; Oliveira, Moreira, Abreu, Mota, & Santos, 2014; Martins et al., 2017).

Research suggests, however, that physical activity behaviour may not be only based on the objective environment measures, but also depends on differentiating between the individual's cognitive representation, or *perception*, of their environment (Giles-Corti & Donovan, 2002; Hume, Salmon, & Ball, 2004; Orstad, McDonough, Stapleton, Altincekic, & Troped, 2017). Perceptions are defined as "an awareness through the senses", and "the way in which something is regarded, understood or interpreted" (Oxford Dictionary, 2018). A child's perception of their everyday environments is formed through the experience of their context: the interplay of the physical, social, cultural and structural forces they are exposed to, and develop through a cyclical process that is interactive with social, cognitive and affective experiences (Orton et al., 2017; Williams, 2003). One way to conceptualize contexts is through the use of the socio-ecological model.

Giles-Corti and Donovan (2002) have suggested a socio-ecological model of recreational physical activity that considers intrapersonal, interpersonal, and physical environment

level determinants of physical activity behaviour. These different levels of environment operate through reciprocal relationship whereby the individual's physical activity behaviour is affected by multiple levels of environmental influences, and physical activity behaviour shapes the surrounding social environment (Townsend & Foster, 2013). Using this theoretical framework, the present thesis will investigate geographic variation in the determinants of children's physical activity and the influence their environments will have on perceiving barriers to activity. This research will then assess the influence of these perceptions as a mitigating factor in achievement of MVPA guidelines. This thesis argues that context of physical activity environments vary across Ontario, and the importance of understanding these differences for changing behaviour. This research provides insight for future research, policy, and practice to consider how structures of children's environments determine their experiences of physical activity for health. This thesis suggests a new way to conceptualize behaviour to determine effective strategies for affecting children's health.

1.2 Theoretical Context

To meet a gap identified in the literature review, the epistemological approach that will guide the primary stage of my research is the conceptualization of the socio-ecological model. The socio-ecological model suggests individual health is shaped in two ways: "(i) behaviour affects and is affected by multiple levels of influence; and (ii) individual behaviour shapes and is shaped by the social environment (reciprocal causation)" (Townsend & Foster, 2013, p. 1101). This approach theorizes that social and physical environments are interdependent; therefore, health is an outcome of the quality of the person-environment fit (Grzywacz & Fuqua, 2000). There are five levels of influences on health in the socio-ecological model, which have both individual and interrelated effects. The conceptualizing model suggests that an individual's health is shaped by (i) intra- and (ii) interpersonal factors, (iii) community and (iv) organizational factors (or institutional), and (v) public policies (Robinson, 2008) (see Figure 1). The socio-ecological model differs from other epistemologies for dealing with public health issues because of the use of more comprehensive multilevel analyses rather than single level (Robinson, 2008). This is important because it considers the complex web of factors influencing public

health challenges. This view will be applied to understand variations in geographic influences of residence in rural communities on the perceived barriers and facilitators to physical activity



Figure 1.1 Socio-ecological model of influences on children's physical activity levels in rural areas.

Adapted from Townsend and Foster (2013), and Public Health Ontario, (2013), this is an example of how the socio-ecologic epistemology can be utilized to consider the potential influences on children's physical activity in their communities. Sherar et al.'s (2009) categories of influence can be aligned with the intra-personal through community levels within the socio-ecological model.

Giles-Corti and Donovan (2002) suggested a more specific socio-ecological model of recreational physical activity (an adaptation of the above socio-ecological model), which is relevant to this thesis. This model considers three levels of influence: intrapersonal, interpersonal and physical environment, that work within reciprocal causation to form determinants of physical activity behaviour. This model is a more appropriate method of measurement for the children's environments in this study because of the varying policy environments across and within the study areas. Based on an individual's environment level and how they experience the interpersonal and physical environment, their perceptions are formed.



Figure 1.2 Socio-Ecological Model of Recreational Physical Activity, Adapted from Giles-Corti and Donovan (2002).

This chart demonstrates some examples of features that fall within each level, which research has demonstrated to have influence on children's physical activity.

The socio-ecological perspective is the most appropriate epistemology for guiding my research because of the need to consider how individual behaviours and interactions are formed within the larger social and environmental context, when considering the determinants of children's physical activity (PHO, 2013). This strategy for understanding health influencing factors has been used in a variety of similar children's health topics, including fruit and vegetable intake in marginalized groups (Robinson, 2008), children's sport participation (Eime et al., 2013), healthy eating in schools (Townsend & Foster, 2013), and addressing childhood obesity (PHO, 2013). However, this perspective's epistemology of health shaping influences has not yet been applied to understand the influence of, and differences in, barriers to children's physical activity in different rural and urban areas in Ontario, Canada. Therefore, this approach will help to increase the understanding of environmental determinants on the experiences of children, allowing for advocacy and implementation of policy and infrastructure within the communities studied.

1.3 Research Objectives and Questions

The overarching objective of this research is to contribute to the understanding of children's physical activity levels and the relationships with their environment. This thesis was completed using data generated through research projects conducted by Dr. Jason Gilliland and his associates within the Human Environments Analysis Laboratory (HEAL). An overall goal of this work is to encourage acknowledgement of children's voices in physical activity research that seeks to improve their health. Furthermore, this thesis seeks to provide a new perspective for examining children's voices with a health equity lens, by considering unique factors that make up a child's context for health. This includes a definition of rurality that exists as more than a dichotomy of rural versus urban. Guided by a socio-ecological framework, the primary research question this thesis addresses is: what role do children's perception of their geographic environments play in engaging in physical activity? Establishing a better understanding of the formation and effect of perceptions related to geographic variation and physical activity has important implications for health research, policy and programming related to health care, infrastructure, and social programs.

In order to meet these objectives, this thesis will answer the following research questions:

- How do factors of children's intrapersonal, interpersonal, and physical environments influence their perceptions of barriers to physical activity?
- 2) What are the similarities and differences in children's perceptions of barriers to physical activity in relation to the level of urbanicity and geographic variation of their home location?
- 3) Do children's perceptions of barriers to physical activity mediate the relationship between their physical environment (urbanicity) and their MVPA levels?

For the purposes of this thesis, I define urbanicity using Vlahov and Galea's (2002) definition which is the "impact of living in an urban area at a given time", and more specifically "the conditions present in urban areas to a much greater extent than non-urban areas" (p.55). To address these questions, this research will use quantitative data

from the Spatial Temporal Environment and Activity Monitoring (STEAM) project. This project took place in 2010-2013 in 33 schools in Southwestern Ontario, and in 2016 in 4 schools in Northwestern Ontario. Examining how and why children hold perceptions of their environment is complex, however this research seeks to form a starting point for future work to consider determinants within the socio-ecological model, and the impact they will have on how children perceive and engage with their environments in pursuit of physical activity. This research provides a starting point for future work to consider *how* structures of children's environments determine their experiences of physical activity for health by providing objective evidence of behaviour and subjective evidence of children's perceptions. This will help to support development of policy, programs, and practices that incorporate, encourage, and facilitate the use of children's perspectives to determine in practice effective strategies for impacting children's health.

1.4 Thesis Format

This thesis follows an integrated article format and includes two separate but related studies. Each of the two studies aim to understand how children's environments influences their physical activity perceptions and behaviour. Each study has the same objective to examine barriers to children's physical activity in their daily environments, in order to suggest methods for alleviating these barriers to promote activity. The first study does this by examining what children perceive as barriers to gain their point of view. The second study examines the mediating effect of these perceptions in the relationship between the physical environment and objective measures of activity. Through these studies, this thesis aims to suggest the importance of considering geography when conducting research on children's physical activity. The main theme present across both studies is the importance of the environmental context in which children experience and engage in physical activity in their everyday lives. The thesis outline is as follows:

Chapter 2 reviews the existing literature on physical activity and outdoor play in children's various environments to identify gaps, methodological limitations and justifies the need for future research.

Chapter 3 discusses the data collection strategies, tools and analysis rationale to provide the reader with a comprehensive understanding of the research methods.

Chapter 4 examines children's perceived barriers to physical activity across varying Canadian environments in Northwestern and Southwestern Ontario, through a socioecological model.

Chapter 5 investigates the relative influence of barriers mediating the relationship between environments and moderate-to-vigorous physical activity levels of Northwestern and Southwestern Ontario children.

Chapter 6 summarizes the findings and relates the integrated articles to critically analyze how future policy and practice can benefit from this work. This chapter will discuss implications of the work, limitations, and areas for future research.

- Bauman, A. (2003). Updating the evidence that physical activity is good for health: an epidemiological review 2000-2003. *Journal of Science and Medicine in Sport*, 7(1), 6–19.
- Bruner, M. W., Lawson, J., Pickett, W., Boyce, W., & Janssen, I. (2008). Rural Canadian adolescents are more likely to be obese compared with urban adolescents. *International Journal of Pediatric Obesity*, 3(4), 205–211. https://doi.org/10.1080/17477160802158477
- Colley, R. C., Carson, V., Garriguet, D., Janssen, I., Roberts, K. C., & Tremblay, M. S. (2017). Physical activity of Canadian children and youth. *Health Reports*, 28(10), 8–16. https://doi.org/October 2017
- Ding, D., Sallis, J. F., Kerr, J., Lee, S., & Rosenberg, D. E. (2011). Neighborhood environment and physical activity among youth: a review. *American journal of preventive medicine*, 41(4), 442-455.
- Eime, R. M., Young, J. A., Harvey, J. T., Charity, M. J., & Payne, W. R. (2013). A systematic review of the psychological and social benefits of participation in sport for children and adolescents: informing development of a conceptual model of health through sport. *International Journal of Behavioral Nutrition and Physical Activity*, 10(1), 98.
- Galloway, T. (2006). Obesity rates among rural Ontario schoolchildren. *Canadian Journal of Public Health*, 353-356. https://doi.org/10.2307/41995739
- Giles-Corti, B., & Donovan, R. J. (2002). The relative influence of individual, social and physical environment determinants of physical activity. *Social Science & Medicine*, 54(12), 1793–1812.
- Government of Canada (2017). *The Chief Public Health Officer's Report on the State of Public Health in Canada 2017 – Designing Healthy Living*. Ottawa. Retrieved from https://www.canada.ca/content/dam/phac-

aspc/documents/services/publications/chief-public-health-officer-reports-statepublic-health-canada/2017-designing-healthy-living/2017-designing-healthy-livingeng.pdf

- Grzywacz, J. G., & Fuqua, J. (2000). The Social Ecology of Health: Leverage Points and Linkages. *Behavioral Medicine*, 26(3), 101–115. https://doi.org/10.1080/08964280009595758
- Hume, C., Salmon, J., & Ball, K. (2004). Children's perceptions of their home and neighborhood environments, and their association with objectively measured physical activity: a qualitative and quantitative study. *Health Education Research*, 20(1). https://doi.org/10.1093/her/cyg095
- Janssen, I. (2012). Health care costs of physical inactivity in Canadian adults. Applied Physiology, Nutrition, and Metabolism, 37(4), 803–806. https://doi.org/10.1139/h2012-061
- Krueger, H., Turner, D., Krueger, J., & Ready, A. E. (2014). The economic benefits of risk factor reduction in Canada: tobacco smoking, excess weight and physical inactivity. *Can J Public Health*, 105(1), e69-e78. https://doi.org/10.2307/canajpublheal.105.1.0e69
- Larouche, R., Garriguet, D., & Tremblay, M. S. (2017). Outdoor time, physical activity and sedentary time among young children: The 2012–2013 Canadian Health Measures Survey. *Can J Public Health*, 107(6), 500. https://doi.org/10.17269/cjph.107.5700
- Martins, J., Marques, A., Peralta, M., Palmeira, A., & Carreiro da Costa, F. (2017).
 Correlates of physical activity in young people: A narrative review of reviews.
 Implications for physical education based on a socio-ecological approach. *Retos, Nuevas tendencias en Educación Física, Deporte y Recreación, 31*, 292-299.
- Oliveira, A. F., Moreira, C., Abreu, S., Mota, J., & Santos, R. (2013). Environmental determinants of physical activity in children: A systematic review. *Archives of*

Exercise in Health and Disease, 4(2), 254-261.

- Orstad, S. L., McDonough, M. H., Stapleton, S., Altincekic, C., & Troped, P. J. (2017). A Systematic Review of Agreement Between Perceived and Objective Neighborhood Environment Measures and Associations With Physical Activity Outcomes. *Environment and Behavior*, 49(8), 904–932. https://doi.org/10.1177/0013916516670982
- Orton, L., Halliday, E., Collins, M., Egan, M., Lewis, S., Ponsford, R., ... Popay, J. (2017). Putting context centre stage: evidence from a systems evaluation of an area based empowerment initiative in England. *Critical Public Health*, 27(4), 477–489. https://doi.org/10.1080/09581596.2016.1250868
- Oxford Dictionary. (2018). Definition of perception in English by Oxford Dictionaries. Retrieved June 14, 2018, from https://en.oxforddictionaries.com/definition/perception
- ParticipACTION. (2018). The Brain + Body Equation: Canadian kids need active bodies to build their best brains. The 2018 ParticipACTION Report Card on Physical Activity for Children and Youth. Toronto, ON. Retrieved from https://www.participaction.com/en-ca/thought-leadership/report-card/2018
- Pate, R. R., Trost, S. G., Dowda, M., Ott, A. E., Ward, D. S., Saunders, R., & Felton, G. (1999). Tracking of Physical Activity, Physical Inactivity, and Health-Related
 Physical Fitness in Rural Youth. *Pediatric Exercise Science*, *11*, 364–376.
- Poitras, V. J., Gray, C. E., Borghese, M. M., Carson, V., Chaput, J. P., Janssen, I., ... & Sampson, M. (2016). Systematic review of the relationships between objectively measured physical activity and health indicators in school-aged children and youth. *Applied Physiology, Nutrition, and Metabolism, 41*(6), S197-S239. https://doi.org/10.1139/apnm-2015-0663
- Public Health Ontario. (2013). Addressing obesity obesity in children and youth: evidence to guide action for Ontario. Toronto, ON: Queen's Printer for Ontario.

- Robinson, T. (2008). Applying the Socio-ecological Model to Improving Fruit and Vegetable Intake Among Low-Income African Americans. *Journal of Community Health*, 33(6), 395–406. https://doi.org/10.1007/s10900-008-9109-5
- Sherar, L. B., Gyurcsik, N. C., Humbert, M. L., Dyck, R. F., Fowler-Kerry, S., Baxter-Jones, A. D. G., ... Baxter-Jones, A. D. G. (2009). Activity and Barriers in Girls (8– 16 yr) Based on Grade and Maturity Status. *Med. Sci. Sports Exerc*, 41(1), 87–95. https://doi.org/10.1249/MSS.0b013e31818457e6
- Smith, A. L., Troped, P. J., McDonough, M. H., & DeFreese, J. D. (2015). Youth perceptions of how neighborhood physical environment and peers affect physical activity: a focus group study. *International Journal of Behavioral Nutrition and Physical Activity*, 12(1), 80. https://doi.org/10.1186/s12966-015-0246-9
- Townsend, N., & Foster, C. (2013). Developing and applying a socio-ecological model to the promotion of healthy eating in the school. *Public Health Nutrition*, 16(6), 1101– 1108. https://doi.org/10.1017/S1368980011002655
- Tremblay, M., Gray, C., Babcock, S., Barnes, J., Bradstreet, C., Carr, D., ... Brussoni, M. (2015). Position Statement on Active Outdoor Play. *International Journal of Environmental Research and Public Health*, 12(6), 6475–6505. https://doi.org/10.3390/ijerph120606475
- Vlahov, D., & Galea, S. (2002). Urbanization, Urbanicity, and Health. Journal of Urban Health: Bulletin of the New York Academy of Medicine, 79(90001), 1S–12. https://doi.org/10.1093/jurban/79.suppl_1.S1
- Warburton, D. E. R., Nicol, C. W., & Bredin, S. S. D. (2006). Health benefits of physical activity: the evidence. *Canadian Medical Association Journal*, 174(6), 801–809. https://doi.org/10.1503/cmaj.051351
- Williams, G. H. (2003). The determinants of health: structure, context and agency. Sociology of Health & Illness, 25, 131–154. https://doi.org/10.1111/1467-9566.00344

Chapter 2

2 Literature Review

The chapter is divided into eight main sections. Section 2.1 provides the reader with context on the search strategy for the literature review to ensure a comprehensive background. Section 2.2 examines the literature on physical activity, and the current Canadian policy on children's physical activity levels. Section 2.3 discusses the ecological perspective for physical activity research. Section 2.4 provides definitions of the environments analyzed within this thesis, as well as the rationale behind them. Section 2.5 provides results of a comprehensive review of the current literature surrounding children's perceptions of barriers and facilitators to physical activity in their environments. Section 2.6 summarizes the evidence regarding previous interpretation of the barriers within the socio-ecological framework of recreational physical activity. Section 2.7 reviews the gaps in the literature that this thesis fills and provides a brief conclusion. Section 2.8 lists the references used within this section.

2.1 Search Strategy

To undergo the literature review, the main search terms used were perceptions, children, physical activity and urbanicity. To obtain a variety of articles related to these terms, synonyms were used throughout the search strings. Additional terms used to locate research were related to methods used by the researchers, including systematic, scoping, and literature reviews, as well as meta-analyses. These types of studies are important because they synthesize large bodies of literature into manageable reading and identify good starting points for mapping out areas of research uncertainty (Petticrew & Roberts, 2006). Quantitative and qualitative studies were used to gain a comprehensive understanding of how children's physical activity patterns are formed in relation to barriers in their communities.

The literature review took place over three main data bases: Web of Science, Scopus, and Google Scholar. Additional literature was collected from the Western University Electronic Thesis and Dissertation Repository, and through searching the reference pages of articles collected. For articles to be considered for the review, the research had to be available in English and focused on children's physical activity. Many studies were excluded due to a focus on nutrition, obesity, or adult's physical activity.

Approximately sixty papers were collected for the pre-literature review. Studies eliminated from the literature review had a lack of relevance after reading through the abstract and methods. Initially, fifteen articles were drawn on for the pre-literature review for two main reasons: (1) the articles were systematic, scoping, and literature reviews, and synthesized a large majority of the additional articles, and (2) the studies offered specific answers to the research question for children in other settings, and were therefore similar to the proposed study. These studies were used to steer research for the present literature review to ensure comprehensiveness. The following sections present the findings of the complete literature review, based on the initial search.

2.2 Physical Activity and the 24-Hour Movement Guidelines

Physical activity (PA) is defined by the Canadian Society for Exercise Physiology [CSEP] (2017) as "bodily movement produced by skeletal muscles that requires energy expenditure," and this activity "increases heart rate and breathing" (p.1). Researchers have widely examined the benefit of physical activity for children's health (Janssen & LeBlanc, 2010; Poitras et al., 2016). Broadly, physical activity is associated with physical, mental and social health indicators (Poitras et al., 2016). Due to the negative health risks and associated economic health burden of not being physically active (Krueger et al., 2014), policy makers have emphasized the importance of developing these behaviours from a young age. The Canadian Physical Activity Guidelines were developed to help families and caregivers ensure children are meeting the necessary requirements of physical activity to achieve health benefits.

In the past ten years researchers in Canada have produced a number of physical activity guidelines, that have grown to include recommendations on sedentary behaviour, outdoor time, and sleep (Tremblay et al., 2007, 2011a, 2011b, 2012, 2016). The most recent 2016 update to these guidelines is called the *24-Hour Movement Guidelines for Children and Youth*. These guidelines integrate movement across the whole day, with a shift in thinking that conceptualizes physical movement to exist on a continuum. The major difference in

the new guidelines is not the behaviour components, but rather the integration of activity over a 24-hour period. The new guidelines also incorporated recommendations related to light-intensity physical activity (LPA), sleep, and play-based activity (Tremblay et al., 2016). These changes were informed by more recent evidence related to the potential health benefits of incorporating the new recommendations, though more research is needed (Tremblay et al., 2016).

One significant development in conceptualizing physical activity behaviour in the 24-Hour Guidelines, is using a movement continuum to consider levels of activity and their impact on health, based on time spent engaging. This is a way to implement the concept of the whole day matters (Tremblay et al., 2016). According to the CSEP (2017), the continuum (Figure 3) helps to conceptualize that there are distinct differences between physiological systems involved in sedentary behaviour and physical activity, and the two should not be considered opposite. For example, a synthesis of the literature by Tremblay, Colley, Saunders, Healy, and Owen (2010), indicates sedentary behaviour is associated with deleterious health outcomes related to metabolism and physical functioning, that differ from those that can be attributed to a lack of being physically active. This perspective suggests that movement and non-movement behaviours should be considered together when assessing healthy living. As explained by Tremblay et al. (2010), achieving high levels of intense physical activity does not necessarily displace minutes spent being sedentary. As a result, we must consider each item's impact in a total activity achievement throughout the course of a day. This was identified as an important consideration by Tremblay et al. (2010) because this could lead to new approaches for minimizing, surveilling and analyzing sedentary behaviour that are different than those that aim to increase physical activity.



*One metabolic equivalent (MET) is defined as the amount of oxygen consumed while sitting at rest and is equal to 3.5 ml O2 per kg body weight per min.

Figure 2.1 Activity Continuum adapted from Tremblay et al. (2010).

As children's activity behaviour is measured on the continuum, there are different physiological resoponses. As such, sedentary behaviour is not synonomous with physical inactivity

Based on this conception then, it is important we draw a distinction between sedentary behaviour and physical inactivity. Sedentary behaviour is classified by little movement and low energy expenditure (\leq 1.5 METs) (Tremblay et al., 2010). Examples in children include watching television on a couch, travel via car or bus, and sitting at the computer for homework or browsing. Tremblay et al. (2010) define physical inactivity as the absence of participating in physical activity (at any intensity), and insufficient activity behaviour to meet activity recommendations. This is reflected as the total time or amount being physically inactive. For example, in children and youth this would mean not

achieving 60 minutes of MVPA each day (CSEP, 2017). This draws attention to the concept of total physical activity (TPA), as indicated in the *24-Hour Guidelines*.

2.2.1 Total Physical Activity

The 24-Hour Guidelines were the first physical activity guidelines in Canada to include recommendations related to total physical activity (TPA). TPA is the overall amount of minutes engaging in physical activity at all intensities (light, moderate, moderate-to-vigorous, vigorous) added together. The goal is to address time spent in physical activity and inactivity across the whole day. This is because research demonstrated evidence of a variety of physiological benefits with total daily physical activity levels. A systematic review conducted by Poitras et al. (2016) synthesized strong positive associations between children's TPA and adiposity levels, several cardio-metabolic biomarkers, physical fitness, and bone health. Furthermore, there was associations locating favourable relationships between TPA and quality of life, motor skill development, and psychological distress. Evidence related to other health indicators such as fat free mass, pro-social behaviour, academic achievement, and self-esteem, were positive but limited (Poitras et al., 2016).

While there is no specific indicator in the 24-Hour Guidelines recommending children achieve high levels of TPA, this recommendation is phrased within a practical and applicable recommendation. The Guidelines suggest specific indicators of moderate-to-vigorous physical activity (MVPA) and vigorous physical activity, "Several hours of a variety of structured and unstructured light physical activities" and "Limited sitting for extended periods". Changing the language of the current Guidelines to incorporate TPA throughout the day, will encourage more research related to understanding the impact of total accumulation time of physical activity, rather than the importance of MVPA bouts alone.

2.2.2 Light Intensity Physical Activity

The evidence based decision making by Tremblay et al. (2016) in the formation of the 24-*Hour Guidelines* also points to the benefits of encouraging light physical activity (LPA) as a method for minimizing physical inactivity. This is any body movement that does not result in sweat production or shortness of breath, including incidental activities. Some examples in children and youth include slow walking/walking a dog, household chores, and light games such as hopscotch or croquet (CSEP, 2017). The new 24-Hour *Guidelines* recommend children and youth age 5-17 achieve several hours of a variety of structured and unstructured LPA each day (Tremblay et al., 2016). The benefit of encouraging additional minutes of LPA was found to be positively associated with cardiometabolic biomarkers in the review by Poitras et al. (2016). While there is limited evidence available on the overall health impact of accumulating LPA minutes throughout the day, Tremblay et al. (2016) reiterate the potential benefits of LPA and TPA. While these levels require more research into their positive effects, Tremblay et al. (2016) recommend they should continue to be considered in future work to elucidate the potential health benefits, rather than MVPA alone.

2.2.3 Moderate to Vigorous Physical Activity

The 24-Hour Movement Guidelines recommend children and youth age 5-17 should accumulate on average a minimum of 60 minutes of moderate-to-vigorous physical activity (MVPA) each day (Tremblay et al., 2016). This is the recommended level of intensity required to achieve the maximum health benefits of physical activity. MVPA is considered activity that causes the participant to sweat, elevate the heart rate and be out of breath (CSEP, 2017). Regular participation in MVPA has been associated with a variety of known health benefits in children and youth including improved body composition, cardiovascular and metabolic health, musculoskeletal health, mental health, and academic achievement (Carson, Tremblay, Chaput, & Chastin, 2016; Janssen & LeBlanc, 2010; Poitras et al., 2016; Strong et al., 2005). This recommendation has remained as an important indicator for disease prevention and health promotion (Tremblay et al., 2016).

2.2.4 Sedentary Behaviour

Sedentary behaviour (SED) has been included in the 24-Hour Guidelines in order to demonstrate recommendations among the movement continuum, that sedentary time is one component of the whole day (Tremblay et al., 2016). The *Guidelines* recommend limiting sitting for extended periods, no more than 2 hours of recreational screen time per

day and replacing sedentary behaviours with physical activity in order to achieve greater health benefits. There is substantial evidence related to the negative implications of SED on children's health. For example, in a systematic review conducted by Carson et al. (2016), they found a gradient can be observed across health behaviours in school aged children and youth. The results demonstrate that as SED time decreases, there are positive associations with health. Research has indicated links between SED and overweight/obesity status in children and increased risks of cardiovascular disease, cancer, and diabetes into adulthood (Carson et al., 2016; Lewis, Napolitano, Buman, Williams, & Nigg, 2017). These effects are distinct from those attributed to physical inactivity. This further supports the notion that movement and non-movement behaviours should be considered together when assessing children's healthy living behaviours.

2.2.5 Outdoor Time

Researchers are now leaning toward a recommendation of increasing outdoor time as a tool for improving physical activity guideline compliance (Larouche, Garriguet, Gunnell, Goldfield, & Tremblay, 2016; Larouche, Garriguet, & Tremblay, 2017; Tremblay et al., 2015). Evidence suggests increasing time outdoors significantly increases activity and steps achieved per day (Larouche et al., 2017). For example, sixty additional minutes of outdoor times increases a child's average daily TPA by 7 minutes (Larouche et al., 2016). This is the first time in Canada's use of the guidelines that this recommendation has been made, and is related to the recent *Position Statement on Active Outdoor Play* for improving children's health (Tremblay et al., 2015).

2.3 Socio-Ecological Perspective to Physical Activity Environments

A growing body of research on physical activity takes a socio-ecological approach to understanding how children's health is shaped by their surrounding environments. According to this framework, interpersonal and physical environments are interdependent, whereby an individual's experience of health and factors impacting physical activity are nested as the innermost level within a variety of influence structures (Bronfenbrenner, 1979). Therefore, when concerned about the current trend in children's physical activity levels, researchers must consider how determinants form an integrated effect on behaviour and health (Humpel, 2002). As explained by Mitchell, Clark, and Gilliland (2016) neighbourhoods and the community environment both limit and facilitate physical activity in children, based on the opportunities available and restrictions impeding activity. For example, Smith et al. (2015) highlighted how aspects of the built environment (streets, buildings, park availability) influence adolescent's ability to move freely through their communities as they cannot drive and lack financial resources. This is a result of the structures both deliberately created (i.e., the built environment) and unintended consequences on the intrapersonal level (characteristics/factors within individual that determine physical activity) of influence.

Giles-Corti and Donovan (2002) suggested a socio-ecological model of recreational physical activity that considered individual, social and physical environment determinants of physical activity behaviour. Giles-Corti and Donovan (2002) propose the importance of such a model is access to supportive physical environment alone is not sufficient for increasing physical activity behaviours. Rather, individual and social environment factors that form environmental perceptions of barriers to physical activity may be equally as important for predicting behaviour as the objectively measured physical environment (Hume et al., 2004). By considering the environmental influences on physical activity through a socio-ecological lens, we can begin to consider how children's physical activity levels are influenced by external, modifiable larger social and environmental contexts (Public Health Ontario, 2013). In order to better understand how these perceptions influence activity in different environments, it is first important to understand how to define varying levels of urbanicity.

2.4 Defining Environments

2.4.1 Understanding "Rural"

The literature review determined that there is a lack of clarity surrounding the definitions of terms used to describe different environments, such as rural and suburban. There are a variety of definitions that tend to emphasize many different indicators for these (and similar) land use terms, but often the characteristics emphasized are density, and distance

to density (Abraham, Sommerhalder, & Abel, 2010; Davison & Lawson, 2006; Lofshult, 2004; Markey, B., Lauzon, G., & Ryser, 2015; Sandercock, Angus, & Barton, 2010). Statistics Canada's definition indicated rural areas have less than 1000 residents, and may contain agricultural lands, and remote and wilderness areas (Statistics Canada, 2015). Any areas with over 1000 people are instead considered population centres, that vary in size between small, medium and large population centres (Statistics Canada, 2015). However, this definition is inadequate for determining the experience of individuals because it neglects communities that fall in between these categories but still consider their lifestyle to fit within a rural category, especially when compared to the largest census metropolitan areas. This can create a variety of implications for health research and programming, such as inappropriate blanketing policies applying to population centres, which neglect to consider the history, geography, socio-economic status, and development trajectories of small communities. These "one-size-fits-all" policies can cause insufficient availability health care, infrastructure, and social programs. Therefore, in order to better operationalize the concept of urbanicity to include a variety of experiences of rurality, a population density spectrum approach will be utilized in this thesis.

2.4.2 Defining Urbanicity and the Population Density Spectrum

Urbanicity was previously defined by Vlahov & Galea (2002) as the "impact of living in an urban area at a given time", more specifically "the conditions present in urban areas to a much greater extent than non-urban areas" (p.55). This definition highlights the contrast between cities and the surrounding areas, and these differences have been shown to have an association with health (Cyril, Oldroyd, & Renzaho, 2013; Jones-Smith & Popkin, 2010). Rather than using the dichotomous definition, as explained by Jones-Smith & Popkin (2010) a spectrum approach enables both an inter- and intra-urban and rural comparison.

A population density and built form spectrum to measure indicators of urbanicity allows for a definition that considers heterogeneity of different land uses, while providing an objective method to operationalize data (Babey, Tan, Wolstein, & Diamant, 2015). This spectrum splits communities into definitions of urbanicity based categorization of built form and population density, with five categories to provide greater insight into differences in physical activity versus a more simplistic rural versus urban, or trilateral (including suburban) division as recommended by Sandercock et al. (2010). Therefore, *urban large-cities* areas are characterized by grid-like road networks, high population density, and high land use mix within settlements greater than 100,000 people. *Suburban large-cities* are areas that are characterized by cul-de-sac road networks, lower population density, and low land use mix within settlements greater than 100,000 people. *Urban small-towns* in this definition accounts for settlement areas with a population between 10,000 and 100,000 people. *Rural small-towns* are settlements with a population between 1,000 and 10,000. Areas defined as simply *rural* are all other areas of our study area mostly characterized by agricultural land and natural areas.

When considering levels of urbanicity in communities, research has demonstrated mixed results on the influence on children's physical activity patterns. Sandercock et al. (2010) conducted a systematic review on the differences in physical activity levels living in different built environments, classified by land use. The majority of studies indicated either no differences, or that urban children were less likely to be active than rural children. This included two studies which compared urban and rural children in Canada, and in both groups no significant differences in MVPA presented (Plotnikoff, Bercovitz, & Loucaides, 2004; Tremblay et al., 2005). Contrary to the results of this systematic review, Moore, Brinkley, Crawford, Evenson, and Brownson (2013) found daily MVPA was significantly lower in rural youth versus urban youth. This group of authors reference their work to be consistent with Davis et al. (2008) who indicated rural children expend less energy in physical activity than urban children each week.

What Moore et al. (2013) neglect to highlight, however, is that Davis et al. (2008) also found that urban children had higher rates of sedentary behaviour than rural counterparts. Davis et al. (2008) attribute this to the possibility that rural children participated in more activities not covered by traditional self-report physical activity assessments, such as farm chores. This conclusion draws attention back to findings of the systematic review by Sandercock et al. (2010), that indicates that the majority of the studies analyzed used a simple urban versus rural division of groups. This is problematic because a dichotomous division between urban and rural neglects to highlight heterogeneity of land uses.

In five of the articles authors found by Sandercock et al. (2010) (Joens-Matre et al., 2008; Kristjansdottir & Vilhjalmsson, 2007; Nelson, Gordon-Larsen, Song, & Popkin, 2006; Springer, Hoelscher, Castrucci, Perez, & Kelder, 2009; Springer, Hoelscher, & Kelder, 2006), when using a spectrum approach (i.e., considering more than a dichotomous rural versus urban relationship) to explaining the built environment, that includes suburban or small city populations, rural *and* urban children were less active than children in suburban areas or small cities. The authors of this work hypothesize that suburban children are the most active because suburban areas have a mixture of urban (access) and rural (openness) characteristics of the environment, but also because of the typically higher levels of socio-economic status, and low populations of ethnic minorities in suburban areas (Sandercock et al., 2010). This is an important mediating factor to consider because both minority status low socio-economic status are significantly associated with decreased levels of physical activity (Felton et al., 2002; Nelson et al., 2006; Sandercock et al., 2010). This demonstrates the need to utilize an approach such as the socio-ecological framework, which accounts for social and geographic determinants of health.

Sandercock et al. (2010) go so far as to say that because the articles they examined oversimplified environments by splitting them into rural *or* urban, this may have led to a misinterpretation of the results. Therefore, a comprehensive understanding of urbanicity's effect on physical activity is still unknown, as current research has been limited by heterogeneous definitions of urbanicity and rurality, and methodologies for measuring environment types (Sandercock et al., 2010; Davidson & Lawson, 2006). However, researchers do agree that characteristics within a child's neighbourhood environment, specifically the presence of environmental supports (including the social, physical and political environment) such as social cohesion, local infrastructure, and accessibility to green areas for playing, influence their physical activity (Moore et al., 2013). Current work recommends researchers conduct work using larger studies, objective measures of physical activity, a less simplified comparison for classifying urbanicities, and attempt to understand how influences within these communities are similar and different

(Sandercock et al., 2010; Hume et al., 2004), a contribution which my research seeks to make.

A number of assumptions exist surrounding the similarities and differences in environmental barriers influencing physical activity. For example, research indicates children are impacted by overarching social characteristics of their communities including socioeconomic status, infrastructure (quality and quantity), presence of nature, and safety (Loebach & Gilliland, 2010; Walia & Liepert, 2012; Yousefian, Ziller, Swartz, & Hartley, 2009), as well as geographic factors such as accessibility to physical activity opportunities especially based on community size (i.e., population less than 10,000 versus 250,000 or more demonstrated a greater lack of accessibility to physical activity opportunities) (Canadian Fitness and Lifestyle Research Institute, 2013). Despite the threats to population health in Canada presented by physical inactivity, there is a paucity of research on the opportunities for physical activity and aspects of the built environment across the spectrum of urbanicity (Yousefian et al., 2009). The majority of literature regarding environment-activity links focuses on larger cities, neglecting areas outside of larger metropolitan areas, and little research has been conducted to examine opportunities for physical activity in built environment across rural and urban settings (Yousefian et al., 2009; Moore et al., 2013). This is a gap which my research seeks to fill.

Furthermore, research highlights the need to study variation between different rural communities (DesMeules et al., 2006; Markey, Lauzon & Ryser, 2015; Walia & Liepert, 2012). In Ontario alone, there are five general types of rural (and northern or remote) communities: urban fringe communities, agriculture communities, cottage country communities, Northern Ontario communities, and Indigenous communities (Markey et al., 2015). In Canada this can extend further to include prairie and maritime communities (Walia & Leipert, 2012). This demonstrates the importance of considering heterogeneity between rural communities (DesMeules et al., 2006), especially when attempting to form health-influencing policy. Walia and Leipert (2012), who conducted qualitative research on barriers to physical activity in Southwestern Ontario, suggest research in different types of rural communities would allow for insights on "issues common and unique to various rural contexts and rural youth populations" (p.12).

2.5 Factors Influencing Physical Activity across Environments

When considering issues or opportunities that hinder/encourage physical activity, a valuable way to consider the environmental influences on children's perceptions is through recognition of barriers and facilitators to physical activity in different urbanicities. Researchers recognize the value in understanding the personal and situational influences on physical activity engagement of children (Loebach & Gilliland, 2010). These perspectives demonstrate the importance of understanding influences and how future interventions will best enhance facilitators, while mitigating barriers to physical activity (Sherar et al., 2009). Sherar et al. (2009) offered useful categories for classifying barriers that have been adapted to define both barriers and facilitators in local outdoor spaces for participating in physical activity in rural communities. *Barriers* are factors that make physical activity difficult or completely inhibit it in outdoor spaces. *Facilitators* are factors that make physical activity possible and promote the behaviour in outdoor spaces.

2.5.1 Barriers and Facilitators to Physical Activity

Table 1 was developed based on 49 studies, to demonstrate similarities and differences between the urban (including suburban), and rural areas in children's perceptions of barriers and facilitators to physical activity. This table provides a summary of the number of articles located within each subheading that presented significant findings. Two additional tables are available in the appendices. Appendix A provides all titles included with a corresponding number, and Appendix B uses these numbers to demonstrate which of the articles fall within each facilitator/barrier heading in Table 1.
Table 2.1 Number of findings demonstrating barriers and facilitators discussed by urbanicity, available in the literature.

		Individual				Social				Physical			
Level of Urbanicity Identified	Facilitator (F) or Barrier (B) to PA	Perceived Safety/Unsafe	Pref. for Weather	Perceived Aesthetics/ Quality	Pref. for Activities	Social Factors	Accessibility (transportation)	Parent perceptions/ rules	Socio-economic circumstance	Planned/ structured activities	Facilities & Amenities	City Planning/ Design	Access (proximity/ availability)
Rural	F	1			1	1	1	1	1	2	1	1	1
	В	3	1		1	2	4	4	2	2	2	3	4
Urban	F	3		2	1	5	2			1	3	1	1
	В	4	1			1	1	2	1	1	2	3	3
Not	F	2		2		2	1				1	1	1
Stated	В	4	2	1	1	1	1				1		3

* Only significant findings represented

Access to physical activity facilities at the community level was both a barrier, and somewhat a facilitator in rural and urban populations studied; however, it is important to provide further context within this level to understand how researchers categorized access. In the urban populations, access was usually determined by ability to walk or bike to locations for physical activity (Loebach & Gilliland, 2010; Grow et al., 2008). Whereas in rural populations, lack of access was usually determined as a barrier if use depended on the need to drive to locations because of a lack of activities close to home (Grow et al., 2008; Loebach & Gilliland, 2010). This example can be expanded upon further in the city planning and design category. Although three barriers to physical activity and one facilitator to physical activity were found in both the urban and rural study populations, there is a difference in the context of what city planning and design means for the communities. For example, in urban communities the layout of streets (i.e., lack of connectivity, intersecting train tracks, and traffic volume) acted as a barrier to physical activity (Loebach & Gilliland, 2010). In rural populations, however, street planning presented as a barrier within the built environment due to underdeveloped centers, dispersed residential patterns, and lack of public open space (Yousefian et al., 2009).

Interpersonal barriers to physical activity are much higher for rural populations, whereas interpersonal facilitators are strong predictors of physical activity in urban areas. One

strong facilitator of physical activity in urban areas was neighbourhood social cohesion and social factors, including social relationships with neighbours and nearby community centres that promote social interaction (Aarts, Wendel-Vos, Van Oers, Van De Goor, & Schuit, 2010). Walia and Liepert (2012) indicated that as a result of distance in rural communities, and the reliance on parents for transportation to visit friends, it is difficult for youth in rural communities to engage in spontaneous group social activities. These researchers suggest rural communities must facilitate social cohesion by allowing for gathering spaces to promote physical activity through social opportunities. By identifying the impact of factors of rural residence that create perceived barriers to interpersonal forces influencing physical activity, this could help researchers to understand one way to adapt the environments of children in an effort to promote physical activity. Considering the positive impact of social factors on children's physical activity in urban areas, more research is needed to understand the influence of such factors in rural populations.

The number of findings for individual barriers and facilitators in rural and urban populations are similar. This may indicate that individual perceptions have a similar level of influence between rural and urban communities (Loebach & Gilliland, 2010). However, it is important to note differences in the types of examples given in the populations for these barriers. For example, perceived lack of safety in an urban area may be due to poor lighting, crime, or traffic volume and speed (Hume et al., 2004; Grow et al., 2008; Loebach & Gilliland, 2010); whereas in rural areas, lack of safety extended to include lack of adult supervision in the neighbourhood, a fear of hunters, and wild animals or loose dogs (Holt et al., 2016). Another example of similarities between the two urbanicities in this theme was perceived aesthetics. Researchers working with rural populations often encouraged greening/street beautification in rural areas to improve children's physical activity (Smith et al., 2015), something that was proven to be a facilitator in urban areas (Abraham et al., 2010, Loebach & Gilliland, 2010).

Suburban was often grouped in the definition of urban in all except one study (Babey et al., 2015), rather than standing alone as a level of urbanicity. This is important to note because children living in suburban areas have demonstrated higher levels of physical activity than rural and urban children (Sandercock, Angus, & Barton, 2010). This

highlights the importance of considering a spectrum of urbanicities, rather than a single dichotomous relationship.

The evidence above indicates the utility of using the same indicators for measuring physical activity barriers and facilitators across the urbanicity spectrum. This approach gives us an indication of what we need to look for in where these differences lie across these different environments. Due to the heterogeneity of the urbanicities, it is important to gather more specific context as to how children's physical activity is affected in the respective populations. For example, if safety was indicated as a concern among people in all urbanicities, what aspects of safety are the specific populations (i.e., rural, urban, suburban) most concerned about? This would allow for better suited policy related to mitigating barriers with facilitators to physical activity. Appendix A demonstrates all findings for barriers and facilitators that were discussed in articles found. The strength of my research is that we used the same survey in all study areas, which vary in urbanicity. This means the research will use identical measurement of barriers across the levels of urbanicity. By using the same indicators across the spectrum to allow for comparability, this will provide future practitioners better direction in where to focus their energy with interventions in order to most effectively impact children's physical activity levels.

2.6 Evidence found within the Socio-ecological Framework of Recreational Physical Activity

The following sections consider evidence regarding previous interpretation of the barriers within the Giles-Corti and Donovan (2002) version of the socio-ecological framework of recreational physical activity.

2.6.1 Individual Environment

The individual environment is shaped by the influence of intrapersonal factors such as psychological and demographic factors, personal preferences and choices (Townsend & Foster, 2013). These factors are widely accepted to have significant impact physical activity behaviour. Demographic factors accepted as health-related dimensions include gender, ethnicity, education, disability, age and sexual orientation (Insel, Roth, Irwin &

Burke, 2016). These dimensions influence what an individual perceives as a barrier in the social or physical environment.

2.6.2 Social Environment

The social environment includes social networks, norms, supports, and standards among groups, individuals, and organizations (Townsend & Foster, 2013). These forces which exist formally and informally throughout children's neighbourhoods. This level of the socio-ecological model has demonstrated a well-established association on the influence of physical activity, but is sometimes overlooked (Ball, 2006; Clark & Scott, 2013). Main themes discussed in the literature as mechanisms of influence within this category for children are parental influence (i.e. rules and perceptions), and relationships with peers positively (i.e. presence of friends) and negatively (i.e. bullying). These themes in a variety of contexts have demonstrated significant influence on the physical activity of children (Abraham et al., 2010; Beets & Foley, 2008; Hume, Salmon, & Ball, 2004; Jago et al., 2009; Lee et al., 2015; Sherar et al., 2009). Given the knowledge that an association exists, this thesis will seek to understand the amount of influence the social environment has on children's physical activity compared with, and in relation to other levels of the socio-ecological model.

2.6.3 Physical Environment

The physical environment is defined as a combination of built and natural environment that influences children's neighbourhoods. As explained by Clark and Scott (2013), this includes the density, diversity and design of the landscapes. These factors have both passive (i.e. urban design influencing accidental physical activity) and active (infrastructure built to promote physical activity) influence on the individual (Giles-Corti & Donovan, 2002). While there is a general consensus that the physical environment impacts physical activity levels (Clark & Scott, 2013), researchers report on a variety of mechanisms by which this relationship may occur. The main themes identified throughout the literature were: accessibility to physical activity resources, availability, neighbourhood aesthetics, and safety.

When considering a lack of accessibility as a barrier to physical activity opportunities, the literature discusses this issue due to proximity and distance of facilities and opportunities (including inability to walk), transportation related issues, and residential density and design. While there are many examples in the literature, some facilities and opportunities mentioned include parks, playgrounds, recreation centers, sports fields, beaches, and public pools. Researchers tend to agree, there is a significant positive association of accessibility to these opportunities and physical activity levels, however the strength of the relationship varies based on children's ages. Controlling for age, evidence has indicated children were more likely to be physically active if there was a physical activity facility within a census block of their home (Gordon-Larsen, Nelson, Page, & Popkin, 2006), as well as within a 500 meter walk of their home (Gilliland et al., 2012). Whereas when children perceived accessibility to physical activity opportunities as a barrier in their neighbourhood, they were less likely to be physically active (Timperio, Crawford, Telford, & Salmon, 2004). Researchers hypothesize that accessibility may especially be an issue for children in rural areas. Markey, Lauzon, and Ryser (2015) indicated parents in smaller communities (population less than 10,000) were more likely to indicate accessibility as a barrier to physical activity for their children than those in the largest communities. This concern was echoed by the voices of children in qualitative research (Jago et al., 2009; Sherar et al., 2009; Walia & Leipert, 2012). This evidence highlights the importance of considering the impact of accessibility on children in the many different types of neighbourhoods across Canada, as they may experience barriers at varying levels in their environments. It is also important however, to consider the types, amount, and availability of infrastructure accessible to the populations being served.

Availability of physical activity promoting infrastructure is described as the presence and amount of age appropriate equipment/activities/landscape design features etc. that provide the opportunities to participate in physical activity. This could include the presence of playgrounds, sports nets, sidewalks, and hiking/biking trails. Availability differs from accessibility because for example, although a park may be close in proximity it may not be perceived as an appropriate play structure by older children, and what is considered appropriate is unavailable. Furthermore, access works in tandem with availability; if a sidewalk is not present (availability) there is no possibility that children can access it.

Research dictates the importance of perceived availability of safe and useable physical activity infrastructure for the promotion of physical activity behaviour (Davison & Lawson, 2006; Evenson et al., 2006; Grow et al., 2008; Yousefian et al., 2009). While some researchers have found a lack of association between availability of bike lanes and walking trails with physical activity in specific demographic groups (such as by gender, age, and socio-economic status) (Davison & Lawson, 2006; Ewing, Schroeer, & Greene, 2004; Jago, Baranowski, Zakeri, & Harris, 2005), arguably this demonstrates the importance of a more detailed understanding as to who experiences a lack of availability of specific resources as a barrier, in order to cater more specifically to the population with the highest risk of negative health concerns.

Less discussed in the literature was the availability of neighbourhood aesthetics, and their role in promoting physical activity. Research has indicated that aesthetically pleasing public landscapes have been found to foster individual wellbeing (Abraham et al., 2010), but more research is needed to understand specifically the influence on children. Features such as trees along streets, interesting things to look at, and neighbourhood lighting, have been found to influence children's physical activity levels (Evenson et al., 2006; (Grow et al., 2008; Loebach & Gilliland, 2010; Mota, Almeida, Santos, & Ribeiro, 2005), however some researchers argue not the presence but the absence of aesthetically pleasing landscapes (causing neighbourhood disorder i.e. presence of garbage, lack of lighting and presence of graffiti) is more important to influence physical activity (Davison & Lawson, 2006; Jago et al., 2005, Mota et al., 2005). Important to note, Loebach & Gilliland (2010) suggest when considering neighbourhood aesthetics and disorder for influencing physical activity, there is a need to understand how children perceive these factors specific to their environments. By attempting to comprehend how children perceive such features, we can better understand how to mitigate these barriers to physical activity. For example, children often associate poor neighbourhood aesthetics with a lack of safety (Loebach & Gilliland, 2010). This demonstrates the interplay of physical environment factors within the socio-ecological model.

Throughout the literature, a lack of neighbourhood safety has been interpreted in a number of ways, such as presence or fear of strangers, loose animals, traffic dangers, poor

neighbourhood infrastructure (e.g., lack of lighting and sidewalks), and crime rates (including gang activity). The implications of poor neighbourhood safety have demonstrated mixed results among the research that has been conducted. In a literature review conducted by Davison and Lawson (2006), of nine studies identified which discussed a perceived lack of safety and children's physical activity, the majority reported no association (Adkins, Sherwood, Story, & Davis, 2004; Mota et al., 2005; Sallis, Alcaraz, Mckenzie, & Hovell, 1999; Trost et al., 2002; Zakarian, Hovell, Hofstetter, Sallis, & Keating, 1994; Burdette & Whitaker, 2005). However, many researchers are reporting the opposite. For example, Evenson et al. (2006), Carver et al. (2005), Gómez, Johnson, Selva, and Sallis (2004), and Molnar, Gortmaker, Bull, & Buka (2004) found perception of safety in neighbourhood was positively associated with physical activity levels. Furthermore, results of a national survey in the United States (2007-2009) indicated a significant association between safety concerns and time spent outside (Larson, Green, & Cordell, 2011), a known positive influencer of physical activity levels. Qualitative studies which sought the perception of respondents through interviews and focus groups found supportive evidence for this association (Smith & Barker, 2001; Loebach & Gilliland, 2010). While the relationship is unclear, Beets and Foley (2008) suggest it may not be the presence of actual environmental characteristics that directly affect safety influencing physical activity levels, but rather the perceptions of neighbourhood characteristics that promote safety perhaps are more influential in decisions to participate in physical activity. This highlights the importance of considering the intersectionality of the socio-ecological model between the individual and social environments and the physical environment.

2.7 Discussion and Conclusions

The purpose of this chapter was to provide an overview on the literature surrounding Canadian physical activity guidelines and evidence for practice, and children's perceptions of their activity environments. This is specifically pertaining to children's perceived barriers and the resultant influence on physical activity levels. While researchers tend to agree across the literature that the most appropriate model for understanding environmental influence on physical activity is the socio-ecological model, there is a paucity of research focusing on children in rural areas in Canada, and a tendency for research to focus efforts on metropolitan areas. This literature review further highlighted research gaps, as explained throughout this chapter.

This chapter highlighted several gaps in the literature. Firstly, there is a lack of clarity surrounding terminology and problematic division of definitions when it comes to thinking about rurality and urbanicity. A large majority of previous research uses a dichotomous, or trichotomous scale for measuring differences between rural and urban, and often defining rural as not urban. The way of thinking creates a problematic gap because it lumps children into inappropriate subgroups, which can lead to a skewed interpretation of the physical activity evidence. In order to mitigate this issue, this thesis will use a population density and built form spectrum to measure indicators of urbanicity. This will allow for a definition that considers heterogeneity of different land uses while maintaining objectivity in measurement. Furthermore, this research will add to the literature due to the current lack of research focusing on Canadian children outside of metropolitan areas.

The importance of this research lies in addressing the gap of rural children's perceptions on what barriers affect their physical activity in outdoor spaces. The literature review determined the need for researchers to include the voices of children in planning decisions (Lee et al., 2015). Despite the recognition of the influence physical activity in outdoor spaces has on children, there is still a lack of understanding of the relationship between the neighbourhood and children's perceptions and behaviours (Loebach & Gilliland, 2010).

- Aarts, M. J., Wendel-Vos, W., van Oers, H. A., van de Goor, I. A., & Schuit, A. J. (2010). Environmental determinants of outdoor play in children: a large-scale crosssectional study. *American journal of preventive medicine*, 39(3), 212-219.
- Abraham, A., Sommerhalder, K., & Abel, T. (2010). Landscape and well-being: a scoping study on the health-promoting impact of outdoor environments. *International Journal of Public Health*, 55(1), 59–69. https://doi.org/10.1007/s00038-009-0069-z
- Adkins, S., Sherwood, N. E., Story, M., & Davis, M. (2004). Physical Activity among African-American Girls: The Role of Parents and the Home Environment. *Obesity Research*, 12(S9), 38S–45S. https://doi.org/10.1038/oby.2004.267
- Babey, S. H., Tan, D., Wolstein, J., & Diamant, A. L. (2015). Neighborhood, family and individual characteristics related to adolescent park-based physical activity. *Preventive medicine*, 76, 31-36. https://doi.org/10.1016/j.ypmed.2015.04.001
- Ball, K. (2006). Integrating understanding of intrapersonal, social and environmental determinants of physical activity. *Journal of Science and Medicine in Sport*, 9, 367– 370. https://doi.org/10.1016/j.jsams.2006.06.010
- Beets, M. W., & Foley, J. T. (2008). Association of Father Involvement and Neighborhood Quality with Kindergartners' Physical Activity: A Multilevel Structural Equation Model. *American Journal of Health Promotion*, 22(3), 195–203. https://doi.org/10.4278/ajhp.22.3.195
- Bronfenbrenner, U. (1979). *The ecology of human development*. Cambridge, MA: Harvard University Press.
- Burdette, H. L., & Whitaker, R. C. (2005). Resurrecting free play in young children: looking beyond fitness and fatness to attention, affiliation, and affect. *Archives of pediatrics & adolescent medicine*, 159(1), 46-50.

- Canadian Fitness and Lifestyle Research Institute. (2013). 2010-2011 Physical Activity Monitor. In *Bulletin 14: Barriers to children's participation in physical activity*. . Ottawa, ON. Retrieved from http://www.cflri.ca/sites/default/files/node/1334/files/CFLR
- Canadian Society for Exercise Physiology [CSEP]. (2017). Glossary of Terms: Canadian 24-Hour Movement Guidelines. Retrieved from http://www.csep.ca/CMFiles/Guidelines/24hrGlines/24HourGuidelinesGlossary_201 7.pdf
- Carson, V., Tremblay, M. S., Chaput, J.-P., & Chastin, S. F. M. (2016). Associations between sleep duration, sedentary time, physical activity, and health indicators among Canadian children and youth using compositional analyses. *Applied Physiology, Nutrition, and Metabolism, 41*(6 (Suppl. 3)), S294–S302. https://doi.org/10.1139/apnm-2016-0026
- Carver, A., Salmon, J., Campbell, K., Baur, L., Garnett, S., & Crawford, D. (2005). How Do Perceptions of Local Neighborhood Relate to Adolescents' Walking and Cycling? *American Journal of Health Promotion*, 20(2), 139–147. https://doi.org/10.4278/0890-1171-20.2.139
- Clark, A. F., & Scott, D. M. (2013). Does the social environment influence active travel? An investigation of walking in Hamilton, Canada. *Journal of Transport Geography*, 31, 278–285. https://doi.org/10.1016/j.jtrangeo.2013.06.005
- Cyril, S., Oldroyd, J. C., & Renzaho, A. (2013). Urbanisation, urbanicity, and health: a systematic review of the reliability and validity of urbanicity scales. *BMC Public Health*, 13(1), 513. https://doi.org/10.1186/1471-2458-13-513
- Davis, A. M., Boles, R. E., James, R. L., Sullivan, D. K., Donnelly, J. E., Swirczynski, D. L., & Goetz, J. (2008). Health behaviors and weight status among urban and rural children. *Rural and Remote Health*, 8(2), 810.

Davison, K., & Lawson, C. T. (2006). Do attributes in the physical environment influence

children's physical activity? A review of the literature. *International Journal of Behavioral Nutrition and Physical Activity*, *3*(1), 19. https://doi.org/10.1186/1479-5868-3-19

- DesMeules, M., Pong, R., Lagacé, C., Heng, D., Manuel, D., Pitblado, R., ... & Koren, I. (2006). How healthy are rural Canadians? An assessment of their health status and health determinants. *Canadian Institute for Health Information*.
- Evenson, K. R., Birnbaum, A. S., Bedimo-Rung, A. L., Sallis, J. F., Voorhees, C. C., Ring, K., & Elder, J. P. (2006). Girls' perception of physical environmental factors and transportation: reliability and association with physical activity and active transport to school. *International Journal of Behavioral Nutrition and Physical Activity*, 3(1), 28. https://doi.org/10.1186/1479-5868-3-28
- Ewing, R., Schroeer, W., & Greene, W. (2004). School Location and Student Travel Analysis of Factors Affecting Mode Choice. *Transportation Research Record: Journal of the Transportation Research Board*, 1895, 55–63. https://doi.org/10.3141/1895-08
- Felton, G. M., Dowda, M., Ward, D. S., Dishman, R. K., Trost, S. G., Saunders, R., & Pate, R. R. (2002). Differences in Physical Activity Between Black and White Girls Living in Rural and Urban Areas. *Journal of School Health*, 72(6), 250–255. https://doi.org/10.1111/j.1746-1561.2002.tb07338.x
- Giles-Corti, B., & Donovan, R. J. (2002). The relative influence of individual, social and physical environment determinants of physical activity. *Social Science & Medicine*, 54(12), 1793–1812.
- Gilliland, J. A., Rangel, C. Y., Healy, M. A., Tucker, P., Loebach, J. E., Hess, P. M., ... & Wilk, P. (2012). Linking childhood obesity to the built environment: a multi-level analysis of home and school neighbourhood factors associated with body mass index. *Can J Public Health*, *103*(9), 15-21. https://doi.org/10.2307/canajpublheal.103.2012.0s15

- Gómez, J. E., Johnson, B. A., Selva, M., & Sallis, J. F. (2004). Violent crime and outdoor physical activity among inner-city youth. *Preventive Medicine*, *39*(5), 876–881. https://doi.org/10.1016/j.ypmed.2004.03.019
- Gordon-Larsen, P., Nelson, M. C., Page, P., & Popkin, B. M. (2006). Inequality in the built environment underlies key health disparities in physical activity and obesity. *Pediatrics*, 117(2), 417–424. https://doi.org/10.1542/peds.2005-0058
- Grow, H. M., Saelens, B. E., Kerr, J., Durant, N. H., Norman, G. J., & Sallis, J. F. (2008).
 Where Are Youth Active? Roles of Proximity, Active Transport, and Built Environment. *Med. Sci. Sports Exerc*, 40(12), 2071–2079. https://doi.org/10.1249/MSS.0b013e3181817baa
- Holt, N. L., Neely, K. C., Spence, J. C., Carson, V., Pynn, S. R., Boyd, K. A., ... & Robinson, Z. (2016). An intergenerational study of perceptions of changes in active free play among families from rural areas of Western Canada. *BMC public health*, *16*(1), 829.
- Hume, C., Salmon, J., & Ball, K. (2004). Children's perceptions of their home and neighborhood environments, and their association with objectively measured physical activity: a qualitative and quantitative study. *Health Education Research*, 20(1). https://doi.org/10.1093/her/cyg095
- Humpel, N. (2002). Environmental factors associated with adults' participation in physical activity A review. *American Journal of Preventive Medicine*, 22(3), 188– 199. https://doi.org/10.1016/S0749-3797(01)00426-3
- Insel, P. M., Roth, W. T., Irwin, J. D., Burke, S. M. (2016). *Core Concepts in Health 2nd Canadian Edition*. Whitby, ON: McGraw Hill Education.
- Jago, R., Baranowski, T., Zakeri, I., & Harris, M. (2005). Observed Environmental Features and the Physical Activity of Adolescent Males. *Am J Prev Med American Journal of Preventive Medicine*, 29(2), 98–104. https://doi.org/10.1016/j.amepre.2005.04.002

- Jago, R., Thompson, J. L., Page, A. S., Brockman, R., Cartwright, K., & Fox, K. R. (2009). Licence to be active: parental concerns and 10–11-year-old children's ability to be independently physically active. *Journal of Public Health*, *31*(4), 472–477. https://doi.org/10.1093/pubmed/fdp053
- Janssen, I., & LeBlanc, A. G. (2010). Systematic review of the health benefits of physical activity and fitness in school-aged children and youth. *International Journal of Behavioral Nutrition and Physical Activity*, 7(1), 40. https://doi.org/10.1186/1479-5868-7-40
- Joens-Matre, R. R., Welk, G. J., Calabro, M. A., Russell, D. W., Nicklay, E., & Hensley, L. D. (2008). Rural–Urban Differences in Physical Activity, Physical Fitness, and Overweight Prevalence of Children. *The Journal of Rural Health*, 24(1), 49–54. https://doi.org/10.1111/j.1748-0361.2008.00136.x
- Jones-Smith, J. C., & Popkin, B. M. (2010). Understanding community context and adult health changes in China: Development of an urbanicity scale. *Social Science & Medicine*, 71, 1436–1446. https://doi.org/10.1016/j.socscimed.2010.07.027
- Kristjansdottir, G., & Vilhjalmsson, R. (2007). Sociodemographic differences in patterns of sedentary and physically active behavior in older children and adolescents. *Acta Paediatrica*, 90(4), 429–435. https://doi.org/10.1111/j.1651-2227.2001.tb00445.x
- Krueger, H., Turner, D., Krueger, J., & Ready, A. E. (2014). The economic benefits of risk factor reduction in Canada: tobacco smoking, excess weight and physical inactivity. *Can J Public Health*, 105(1), e69-e78. https://doi.org/10.2307/canajpublheal.105.1.0e69
- Larouche, R., Garriguet, D., Gunnell, K. E., Goldfield, G. S., & Tremblay, M. S. (2016).
 Outdoor time, physical activity, sedentary time, and health indicators at ages 7 to 14: 2012/2013 Canadian Health Measures Survey. *Statistics Canada*, 27(9), 3–13.
- Larouche, R., Garriguet, D., & Tremblay, M. S. (2017). Outdoor time, physical activity and sedentary time among young children: The 2012–2013 Canadian Health

Measures Survey. *Can J Public Health*, *107*(6), 500. https://doi.org/10.17269/cjph.107.5700

- Larson, L. R., Green, G. T., & Cordell, H. K. (2011). Children's Time Outdoors: Results and Implications of the National Kids Survey. *Journal of Park and Recreation Administration*, 29(2), 1–20.
- Lee, H., Tamminen, K. A., Clark, A. M., Slater, L., Spence, J. C., & Holt, N. L. (2015). A meta-study of qualitative research examining determinants of children's independent active free play. *International Journal of Behavioral Nutrition and Physical Activity*, 12(1), 5. https://doi.org/10.1186/s12966-015-0165-9
- Lewis, B. A., Napolitano, M. A., Buman, M. P., Williams, D. M., & Nigg, C. R. (2017). Future directions in physical activity intervention research: expanding our focus to sedentary behaviors, technology, and dissemination. *Journal of Behavioral Medicine*, 40(1), 112–126. https://doi.org/10.1007/s10865-016-9797-8
- Loebach, J., & Gilliland, J. (2010). Child-led tours to uncover children's perceptions and use of neighborhood environments. *Children Youth and Environments*, 20(1), 52-90. https://doi.org/10.7721/chilyoutenvi.20.1.0052
- Lofshult, D. (2004). Getting kids active. IDEA Health & Fitness Source, 22(5), 16-17.
- Markey, B., Lauzon, G., & Ryser, M. (Eds. . (2015). Rural restructuring and its impact on community recreation opportunities. Annals of Leisure Research (Vol. 18). Brandon, MB. https://doi.org/10.1080/11745398.2014.980834
- Mitchell, C., Clark, A., & Gilliland, J. (2016). Built Environment Influences of Children's Physical Activity: Examining Differences by Neighbourhood Size and Sex. *International Journal of Environmental Research and Public Health*, 13(1), 130. https://doi.org/10.3390/ijerph13010130
- Molnar, B. E., Gortmaker, S. L., Bull, F. C., & Buka, S. L. (2004). Unsafe to Play? Neighborhood Disorder and Lack of Safety Predict Reduced Physical Activity among Urban Children and Adolescents. *American Journal of Health Promotion*,

18(5), 378–386. https://doi.org/10.4278/0890-1171-18.5.378

- Moore, J. B., Brinkley, J., Crawford, T. W., Evenson, K. R., & Brownson, R. C. (2013). Association of the built environment with physical activity and adiposity in rural and urban youth. *Preventive Medicine*, 56, 145–148. https://doi.org/10.1016/j.ypmed.2012.11.019
- Mota, J., Almeida, M., Santos, P., & Ribeiro, J. C. (2005). Perceived neighborhood environments and physical activity in adolescents. *Preventive medicine*, 41(5-6), 834-836. https://doi.org/10.1016/j.ypmed.2005.07.012
- Nelson, M. C., Gordon-Larsen, P., Song, Y., & Popkin, B. M. (2006). Built and Social Environments: Associations with Adolescent Overweight and Activity. *American Journal of Preventive Medicine*, *31*(2), 109–117. https://doi.org/10.1016/j.amepre.2006.03.026
- Petticrew, M., & Roberts, H. (2006). Systematic Reviews in the Social Sciences: A Practical Guide. Oxford: Blackwell Publishing. https://doi.org/10.1002/9780470754887
- Plotnikoff, R. C., Bercovitz, K., & Loucaides, C. A. (2004). Physical Activity, Smoking, and Obesity Among Canadian School Youth: Comparison Between Urban and Rural Schools. *Canadian Journal of Public Health*, 413-418. https://doi.org/10.2307/41994418
- Poitras, V. J., Gray, C. E., Borghese, M. M., Carson, V., Chaput, J. P., Janssen, I., ... & Sampson, M. (2016). Systematic review of the relationships between objectively measured physical activity and health indicators in school-aged children and youth. *Applied Physiology, Nutrition, and Metabolism, 41*(6), S197-S239. https://doi.org/10.1139/apnm-2015-0663
- Public Health Ontario. (2013). Addressing obesity obesity in children and youth: evidence to guide action for Ontario. Toronto, ON: Queen's Printer for Ontario.

Sallis, J. F., Alcaraz, J. E., McKenzie, T. L., & Hovell, M. F. (1999). Predictors of change

in children's physical activity over 20 months. *American journal of preventive medicine*, 16(3), 222-229.

- Sandercock, G., Angus, C., & Barton, J. (2010). Physical activity levels of children living in different built environments. *Preventive Medicine*, 50(4), 193–198. https://doi.org/10.1016/j.ypmed.2010.01.005
- Sherar, L. B., Gyurcsik, N. C., Humbert, M. L., Dyck, R. F., Fowler-Kerry, S., Baxter-Jones, A. D. G., ... Baxter-Jones, A. D. G. (2009). Activity and Barriers in Girls (8– 16 yr) Based on Grade and Maturity Status. *Med. Sci. Sports Exerc*, 41(1), 87–95. https://doi.org/10.1249/MSS.0b013e31818457e6
- Smith, F., & Barker, J. (2001). Commodifying the countryside: the impact of out-ofschool care on rural landscapes of children's play. *Area*, 33(2), 169–176. https://doi.org/10.1111/1475-4762.00020
- Smith, A. L., Troped, P. J., McDonough, M. H., & DeFreese, J. D. (2015). Youth perceptions of how neighborhood physical environment and peers affect physical activity: a focus group study. *International Journal of Behavioral Nutrition and Physical Activity*, 12(1), 80. https://doi.org/10.1186/s12966-015-0246-9
- Springer, A. E., Hoelscher, D. M., Castrucci, B., Perez, A., & Kelder, S. H. (2009).
 Prevalence of physical activity and sedentary behaviors by metropolitan status in 4th-, 8th-, and 11th-grade students in Texas, 2004-2005. *Preventing Chronic Disease*, 6(1), A21.
- Springer, A. E., Hoelscher, D. M., & Kelder, S. H. (2006). Prevalence of Physical Activity and Sedentary Behaviors in US High School Students by Metropolitan Status and Geographic Region. *Journal of Physical Activity and Health*, 3(4), 365– 380. https://doi.org/10.1123/jpah.3.4.365
- Statistics Canada (2015). Rural Area. Retrieved October 13, 2017, from https://www12.statcan.gc.ca/census-recensement/2011/ref/dict/98-301-X2011001eng.pdf

- Strong, W. B., Malina, R. M., Blimkie, C. J. R., Daniels, S. R., Dishman, R. K., Gutin, B.,
 ... Trudeau, F. (2005). Evidence based physical activity for school-age youth. *The Journal of Pediatrics*, 146(6), 732–737. https://doi.org/10.1016/j.jpeds.2005.01.055
- Timperio, A., Crawford, D., Telford, A., & Salmon, J. (2004). Perceptions about the local neighborhood and walking and cycling among children. *Preventive Medicine*, 38(1), 39–47. https://doi.org/10.1016/j.ypmed.2003.09.026
- Townsend, N., & Foster, C. (2013). Developing and applying a socio-ecological model to the promotion of healthy eating in the school. *Public Health Nutrition*, 16(6), 1101– 1108. https://doi.org/10.1017/S1368980011002655
- Tremblay, M., Gray, C., Babcock, S., Barnes, J., Bradstreet, C., Carr, D., ... Brussoni, M. (2015). Position Statement on Active Outdoor Play. *International Journal of Environmental Research and Public Health*, 12(6), 6475–6505. https://doi.org/10.3390/ijerph120606475
- Tremblay, M. S., Barnes, J. D., Copeland, J. L., Esliger, D. W., Tremblay, J. D., Barnes, J. L., & Copeland, D. W. E. (2005). Conquering Childhood Inactivity: Is the Answer in the Past? *Med. Sci. Sports Exerc*, *37*(7), 1187–1194. https://doi.org/10.1249/01.mss.0000170079.71522.a6
- Tremblay, M. S., Carson, V., Chaput, J.-P., Connor Gorber, S., Dinh, T., Duggan, M., ...
 Zehr Canadian, L. (2016). Canadian 24-Hour Movement Guidelines for Children and Youth: An Integration of Physical Activity, Sedentary Behaviour, and Sleep 1. *Appl. Physiol. Nutr. Metab. Appl. Physiol. Nutr. Metab*, 4118(18), 311–327. https://doi.org/10.1139/apnm-2016-0151
- Tremblay, M. S., Colley, R. C., Saunders, T. J., Healy, G. N., & Owen, N. (2010). Physiological and health implications of a sedentary lifestyle. *Applied Physiology*, *Nutrition, and Metabolism*, 35(6), 725–740. https://doi.org/10.1139/H10-079
- Tremblay, M. S., LeBlanc, A. G., Carson, V., Choquette, L., Connor Gorber, S., Dillman, C., ... Spence, J. C. (2012). Canadian Sedentary Behaviour Guidelines for the Early

Years (aged 0–4 years). *Applied Physiology, Nutrition, and Metabolism*, *37*(2), 370–380. https://doi.org/10.1139/h2012-019

- Tremblay, M. S., LeBlanc, A. G., Janssen, I., Kho, M. E., Hicks, A., Murumets, K., ... Duggan, M. (2011a). Canadian Sedentary Behaviour Guidelines for Children and Youth. *Applied Physiology, Nutrition, and Metabolism*, 36(1), 59–64. https://doi.org/10.1139/H11-012
- Tremblay, M. S., Shephard, R. J., Brawley, L. R., Cameron, C., Lynn Craig, C., Duggan, M., ... Young, T. (2007). Physical activity guidelines and guides for Canadians: facts and future. *Canadian Journal of Public Health Appl. Physiol. Nutr. Metab. J. Public Health Appl. Physiol. Nutr. Metab*, *32*(32), 218–224. https://doi.org/10.1139/H07-125
- Tremblay, M. S., Warburton, D. E. R., Janssen, I., Paterson, D. H., Latimer, A. E., Rhodes, R. E., ... Duggan, M. (2011b). New Canadian Physical Activity Guidelines. *Applied Physiology, Nutrition, and Metabolism*, 36(1), 36–46. https://doi.org/10.1139/H11-009
- Trost, S. G., Pate, R. R., Sallis, J. F., Freedson, P. S., Taylor, W. C., Dowda, M., & Sirard, J. (2002). Age and gender differences in objectively measured physical activity in youth. *Med. Sci. Sports Exerc*, 34(2), 350–355.
- Vlahov, D., & Galea, S. (2002). Urbanization, Urbanicity, and Health. Journal of Urban Health: Bulletin of the New York Academy of Medicine, 79(90001), 1S–12. https://doi.org/10.1093/jurban/79.suppl_1.S1
- Walia, S., & Liepert, B. (2012). Perceived facilitators and barriers to physical activity for rural youth: an exploratory study using photovoice. *Rural and Remote Health*, *12*(1842).
- Yousefian, A., Ziller, E., Swartz, J., & Hartley, D. (2009). Active Living for Rural Youth: Addressing Physical Inactivity in Rural Communities. *Journal of Public Health Management and Practice*, 15(3), 223–231.

https://doi.org/10.1097/PHH.0b013e3181a11822

Zakarian, J. M., Hovell, M. F., Hofstetter, C. R., Sallis, J. F., & Keating, K. J. (1994).
Correlates of vigorous exercise in a predominantly low SES and minority high school population. *Preventive medicine*, 23(3), 314-321.
https://doi.org/10.1006/pmed.1994.1044

Chapter 3

3 Data Collection; Tools and Measurement

The following chapter will provide more information on specific data collection procedures for the subsequent studies included in this thesis. By describing these details, this chapter aims to provide more clarity on the study area, participants and recruitment, tools to measure variables, and data cleaning procedures, to give the reader context into decisions related to the statistical analysis throughout the remainder of the thesis. This thesis follows an integrated article format and therefore some of the information discussed here to provide clarity will be repeated within the studies in following chapters. This chapter, however, will not provide additional detail on statistical methods for analyzing data, which will instead appear only in the relevant studies. The chapter is divided into six main sections. Section 3.1 provides an in-depth description of the research setting and participant recruitment. Section 3.2 discusses the data analysis procedures related to the dependent variables. Section 3.4 will justify the methods selected to analyze the variables examined throughout the remainder of this thesis. Section 3.5 will conclude this chapter and Section 3.6 will provide the references used within this chapter.

3.1 Research Setting

The studies will draw on a multi-year, population based study called the Spatial Temporal Environment and Activity Monitoring (STEAM) project conducted by the Human Environmental Analysis Laboratory (HEAL) at Western University. STEAM used a combination of methods including GPS monitoring, accelerometers, daily activity diaries, parent and child surveys and focus groups to investigate effects of the built environment on health behaviours in children. Data was collected over 14 days at two time points in each study cohort, allowing for an understanding of the influences of children's environments on their perceptions of health behaviour opportunities over time. This study was approved by the Non-Medical Research Ethics Board of the University of Western Ontario (NM-REB #: 17918S & 108029) prior to the onset of the studies (see Appendix C and D). All schoolboards participating in the STEAM project granted permission through their internal research ethics board to complete the protocol. Parents provided consent for their children to participate, and all children provided assent to participation (see Appendix E and F).

3.1.1 Sample

Between the years of 2010-2013, data was collected in six regions (with varying urbanicities) of Southwestern Ontario across four public schoolboards and one private (English and French). Research was conducted in both the fall and spring months in seven day intervals for each study period. Across the time span, data was collected from a total of 33 randomly selected schools (2 in year 1, 6 in year 2, 10 in year 3, and 14 in year 4) of 63 initially contacted. These schools represented a socially and spatially stratified sample of the total population. Across the four-year study period, there was 100% retention of schools. Recruitment presentations were made to 1394 students, of which 932 agreed to participate (66.9% participation rate). A total of 791 students (84.9%) in this group completed the data collection across both time points in the Southwestern Ontario cohorts.

In 2016, the study was replicated in three Northwestern Ontario communities located in the Thunder Bay District, in four schools (English only), in two public schoolboards. Research was conducted in the fall and winter in seven day intervals during each study period. Across the time span, data was collected from purposefully selected schools. Purposeful selection was the result of two major factors: 1) a pre-existing relationship of the research team with the school boards and community, and 2) these were the only four schools located in the areas. Across the study period, there was 100% retention of schools. Recruitment presentations were made to 194 students, of which 136 participated in data collection in the first round of the study (70.1% participation). A total of 125 students (91.2%) in this group completed the data collection across both time points in the Northwestern Ontario cohort.

This thesis will draw from a total data set of 892 students in study one (Chapter 4), and 546 students in study two (Chapter 5) between the ages of 8-14 years. The data used to inform this research will include all child surveys to understand perceptions (self-

reported), corresponding parent surveys to determine demographics, GPS derived home location (Study 2), and accelerometry to determine the physical activity levels within urbanicities (Study 2).



Figure 3.1 Map of STEAM Study Areas in Northwestern and Southwestern Ontario including levels of urbanicity.

3.2 Data Collection

The following section discusses the major tools used to collect data that will be analyzed in the present studies.

3.2.1 Survey Tools

The parent and youth surveys for the STEAM project were developed by the HEAL at Western University. These questions were initially developed for the pilot study but have evolved to include questions regarding the perceptions of home and school environments, active travel to school, healthy eating, physical activity habits, parents' rules, safety and crime perceptions, quality-of-life measurements, and individual-level social and demographic questions including postal code. The most recent version of the relevant survey tool questions can be found in Appendix G.

Eight questions incorporated to measure barriers in Study 1 relating to safety on the streets, presence of crime and local infrastructure were adapted from a commonly used and validated tool called *The Neighbourhood Environment Walkability Survey* (NEWS) (Brownson et al., 2004). NEWS is used to measure residents' perceptions of their neighbourhoods and the design features related to physical activity (Brownson et al., 2004; Cerin, Saelens, Sallis, & Frank, 2006). This tool has demonstrated modest correlations of neighbourhood design with physical activity measured with accelerometers (Atkinson, Sallis, Saelens, Cain, & Black, 2005) and self-reported (De Bourdeaudhuij, Sallis, & Saelens, 2003). The NEWS was developed based on transportation and urban planning literature, and with input from urban planners. Questions initially listed in the NEWS were recoded from facilitators into barriers in the present study to measure the influence of perceiving barriers to activity.

The additional nine questions used in Study 1 were developed based on background relationships identified in the literature, use in previous studies, or to measure necessary socio-demographic characteristics of the participants. Specifications related to data management and background literature are discussed in Chapter 4.

3.2.2 Accelerometers

To synthesize results in Study 2, physical activity levels will be measured objectively using accelerometer data. Accelerometry is the gold standard tool for assessing fieldbased physical activity in children (Borghese et al., 2017). The model utilized in this study was the Actical® Z accelerometers (Phillips - Respironics, Oregon, USA). This device uses 30-s epochs to measure energy expenditure (METS), providing an index of physical activity intensity throughout the course of wear time. This tool has demonstrated high levels of reliability especially compared to other devices (Esliger & Tremblay, 2006), with one major benefit being the ability to differentiate between time spent engaging in sedentary, light, moderate and vigorous physical activity behaviour (Puyau, Adolph, Vohra, Zakeri, & Butte, 2004).

Participants in the STEAM study were asked to wear portable Actical® Z accelerometers on their right hip attached with a nylon-elastic band. Participants wore the device for eight consecutive days (4-6 weekdays and 2-3 weekend days) for all waking hours, removing it only for sleeping, bathing and swimming. To determine activity levels, validated movement thresholds were applied (see Figure 2.1 for range indicators) (Puyau et al., 2004; Tremblay et al., 2016).

One limitation to the use of accelerometers, however, is the lack of an established standard wear time requirement in the literature when measuring children's physical activity. This includes a lack of clarity on the importance of weekdays versus weekend days when attempting to interpret children's physical activity behaviour trends despite known differences in physical activity levels (Trost, Pate, Freedson, Sallis, & Taylor, 2000). Trost, et al., (2000) however found that a 4-5 day monitoring period has a testretest reliability of 0.8 among children (grade 1 to 6), and 0.7 among adolescents (grade 7 to 12). Furthermore, an inclusion of at least one weekend day has been found to be more representative of physical activity patterns as a whole (Comte et al., 2013). As well, in a study by Rich et al. (2013) it was found that the reliability measure of activity using accelerometers increased by 6% when including both weekday and weekend measurements of children's physical activity behaviour. Based the literature review related to use of the device, allowing for comparability of studies, and using the established evidence, this study will use a minimum of four valid days (ten consecutive hours of wear time) including one weekday and one weekend. This criteria maximizes our sample potential while attempting to best represent average physical activity behaviours in a typical week.

3.2.3 GPS Devices

Home location for each child was identified through the passive tracking of participants with the use of a VisionTac VGPS-900 GPS logger. This device continuously records spatial locations in 1-second intervals. Participants were asked to wear the GPS devices

during all waking hours for 8 consecutive days unless they were sleeping, bathing, or swimming.

3.2.4 Median Household Income

When attempting to control for parental income, the research team had to adapt the measure from Study 1 to Study 2. Rather than using parents' self-report information median household income (MHHI) measured in CAD at the dissemination area level was controlled for in the model. This strategy was used because a large majority of parents elected not to report their income on the parent survey. We were unable to impute this information to account for missing cases because this information was not missing at random.

Dissemination areas (DA) are geographic units made up of one or more adjacent dissemination blocks. DA are the smallest standard geographic unit available for Canadian census data (Statistics Canada, 2012). MHHI was determined by overlaying child participants' home locations, based on the GPS monitoring, with DA level census data for the corresponding year of study (i.e. 2011 or 2016) in ArcGIS 10.1 (ESRI, 2017; Statistics Canada, 2016). MHHI as a control was applied in the path analysis. This was because previous work (Study 1) determined parent's income had a significant effect on the likelihood of children reporting perceived barriers to their physical activity. Furthermore, the 2011 and 2016 Canadian Census information demonstrate differences in the MHHI in the study areas thereby indicating a need to control for this information (see https://censusmapper.ca).

3.3 Justification for Analysis of Variables

3.3.1 Procedure for Analyzing Children's Perceptions

Each study handles the use of the child's report of perceiving barriers differently. This was due to the specific research questions of the studies, and the statistical analysis methods used in each. The original survey questions were asked with four point Likert-type questions. These questions forced children to choose between *strongly disagree*, *disagree*, *agree*, and *strongly agree*, rather than allowing them to take a neutral stance.

In Study 1, children's responses to the survey questions were treated as dependent variables. This study analyzes the relationship of children's environments and the likelihood that they would agree to perceiving specific barriers to physical activity. As a result, the 4-point data were recoded to binary variables (i.e., 0 for disagree, 1 for agree) to enhance validity of inference in this analysis (Harwell & Gatti, 2001). Furthermore the extent that children agree or disagree that the barrier was present was not necessary for measuring the objective of this study. Four questions asked about the presence of facilitators and were reverse recoded to maintain consistency of measuring barriers in this study (i.e., *do not* know people, *not* enough sidewalks, *not* enough bike lanes, *not* enough trees). In order to help the reader understand the findings in relation to one another, the results were presented by organizing them thematically into three groups. These groups were *safety*, *social*, and *neighbourhood* barriers, and were developed based on the literature review.

In Study 2, children's responses were included as independent variables within a structural equation model to understand their effect on objectively measured MVPA of participants. This study used the three thematically defined groups to assess children's responses as Likert scales. Each score has a minimum of four questions, that were combined into a single composite score for each participant, to provide a quantitative interval measurement scale (i.e. 1 for *strongly disagree*, 2 for *disagree*, 3 for *agree*, 4 for *strongly agree*) (Boone & Boone, 2012). This tool was used to consider the responses as continuous variables within the structural equation modeling.

3.3.2 Changes in Analysis of Activity

In the creation of the new 24-Hour Movement Guidelines for Children and Youth, Tremblay et al. (2016) adapted a new analysis method when assessing adherence to the behaviours included in the guidelines. Previously, the measure focused on physical activity adherence and analyzed if 60 minutes of MVPA was achieved on at least 6 of 7 days per week, in isolation of the other behaviour recommendations (Colley et al., 2011; Tremblay et al., 2016). The new methodology suggests instead that researchers should examine time spent engaging in behaviours, and average them across the entire week. Tremblay et al. (2016) suggest this paradigm shift recognizes the variability in movement behaviours on the continuum (Figure 3), and the total accumulation of volume of activity.

As a result of the new methodology, minutes of activity tend to be inflated when compared to the previous 6 of 7 days method (33% versus 7% respectively achieving MVPA recommendation) (Colley et al., 2017). The average approach, however, recognizes the day to day variations in activity emphasizing sufficient total weekly volume and ensures consistency in the approach for each movement behaviour (Janssen, Roberts, & Thompson, 2017; Roberts et al., 2017). Based on the trends in recent literature related to measuring adherence to the *Guidelines* and the likelihood for analyzing these behaviours using the average approach in future research, this thesis will use this method when analyzing accelerometer data. By using the average approach, this will allow for comparability between studies past and future using the same method.

3.3.3 Measuring Urbanicity and the Spectrum Tools

In Study 1, urbanicity in which the primary home of each child is located was classified into five groups: (1) Urban large-city, which includes areas that are characterized by grid-like road networks, high population density, and high land use mix within settlements greater than 100,000 people; (2) Suburban large-city, which includes areas that are characterized by irregular, looping and cul-de-sac road networks, lower population density, and low land use mix within settlements greater than 100,000 people; (3) Urban small-town, which includes settlement areas with a population between 10,000 and 100,000 people; (4) Rural small-towns, which include settlements with a population between 1,000 and 10,000; and (5) Rural areas, which are all other areas of our study area, with low population density and mostly characterized by agricultural land and natural areas. Home location was compared to this categorization of built form and population density spectrum to determine level of urbanicity, allowing for greater insight into location-based presence of environmental physical activity barriers.

After completion of Study 1 Taylor, Clark and Gilliland determined that the approach could be further improved upon to better explain the variation between the most rural and urban areas. Using the GPS home identified location, we were able to classify participants within the urbanicity spectrum based on intersection and population density. These measures have been used previously in the literature to measure urbanicity and the association of built environment features with MVPA in adolescents (Boone-Heinonen, Popkin, Song, & Gordon-Larsen, 2010). According to a nationally representative sample in American children in a study by Boone-Heinonen et al. (2010), differences in MVPA levels were found to exist in adolescents with a three-level urbanicity categorization. This categorization included a 1, 3, 5, and 8 km buffer, controlling for population density (increases as urbanicity increases) and measuring intersection density. Boone-Heinonen et al. (2010) found MVPA was positively and independently associated with increasing intersection density (3 or more-way intersections per square km). These measures were determined to provide a good starting point for our research as they allow for objectively measured data, that we could compare our GPS identified points with reliable indicators collected in the Canadian Census. In order to create our urbanicity spectrum, six main steps were carried out.

Using Canadian Census data (Statistics Canada, 2016), dissemination block population density (people/km²) for each child's home location was calculated using ArcGIS 10.1 (ESRI, 2017). A dissemination block is an area bounded on all sides by roads and/or boundaries of standard geographic areas, and is the smallest geographic area population and dwelling counts are disseminated in the census (Statistics Canada, 2012). The next step was to create a 500m Euclidean buffer around each child's home. According to Cavagna, Franzetti, and Fuchimoto (1983), at age 12, children can walk up to 5km/hour, meaning a 500m buffer is about 6 minutes walking distance for the average child (Mitchell, Clark, & Gilliland, 2016). Step three was to compute the number of 4-way intersections within the 500m buffer, and step four was to compute intersection density for each buffer (# of intersections $/ \text{km}^2$). This captured the built form measurement of the spectrum (Boone-Heinonen et al., 2010). Results up to this point were transferred from ArcGIS into the Statistical Package for Social Sciences software (SPSS) (IBM Corp, 2016). Step five was to compute Z-scores for each child, using the raw scores of population density and intersection density. The final step was to compute the index for urbanicity. The formula was: ($Z_{Population Density} \times Z_{Intersection Density}$). This formula was

moderately correlated with the original urbanicity score measure used in Study 1 ($r_s = -0.69$, p < 0.001), representing the geography of the regions.

3.4 Measuring the Data

The following table demonstrates all variables that went into the statistical models in the studies, details on how they were measured, and where they were derived from.

Variable	Independent, Dependent or Control	Study 1 or Study 2	How it was Measured	Data Source				
Demographic Variables								
Gender	Independent in Study 1 Control in Study 2	Both	Binary Variable (No children selected other).	Child Survey				
Age	Independent in Study 1 Control in Study 2	Both Continuous Variable		Child Survey				
Visible Minority	Independent	Study 1	Binary Variable	Child Survey				
# of Parents in Main Home	Independent	Study 1	Binary Variable	Child Survey				
Household Arrangement	Independent	Study 1	Binary Variable	Child Survey				
Parental Employment	Independent	Study 1	Binary Variable	Parent Survey				
Median Family Income	Independent	Study 1	Continuous Variable	Income at DA Level of residence				
Household Income at Dissemination Area Level	Control	Study 2	Continuous Variable	Income at DA Level of residence				
Urbanicity								
Urbanicity	Independent	Study 1	Ordinal, Categorical Variable	Postal Code, Child or Parent Survey				
Urbanicity	Exposure	Study 2	Continuous Variable	GPS, Census Data				
Perceptions								
Perceptions of Barriers	Dependent	Study 1	4-point Likert data was recoded to	Child Survey				

Table 3.1 Measuring Variables

			binary variables (disagree/agree) to enhance validity of inference with regression				
Perceptions of Barriers	Independent	Study 2	Created scored index based on themed questions using 4-point Likert data	Child Survey			
Physical Activity							
MVPA	Dependent	Study 2	Total Average Minutes	Accelerometers			

3.5 Conclusion

The purpose of this chapter was to discuss rationale of the data collection strategies, and analysis of the following studies. The aim was to give the reader a comprehensive understanding of the research tools and measurement, to provide insight into decision-making and thought processes. Due to the integrated article format of this thesis, justification details of methods are not included in length in the studies. After reading this chapter, one should have a better grasp of the background for all tools used in Chapters 4 and 5. Having this knowledge will provide a foundational understanding for the selected statistical methods for analyzing data, which appears in the relevant studies.

- Atkinson, J. L., Sallis, J. F., Saelens, B. E., Cain, K. L., & Black, J. B. (2005). The Association of Neighborhood Design and Recreational Environments With Physical Activity. *American Journal of Health Promotion*, 19(4), 304–309.
- Boone-Heinonen, J., Popkin, B. M., Song, Y., & Gordon-Larsen, P. (2010). What neighborhood area captures built environment features related to adolescent physical activity? *Health & Place*, 16(6), 1280–1286.
- Boone, H. N., & Boone, D. A. (2012). Analyzing likert data. *Journal of extension*, 50(2), 1-5.
- Borghese, M. M., Tremblay, M. S., Leblanc, A. G., Leduc, G., Boyer, C., & Chaput, J. P. (2017). Comparison of ActiGraph GT3X+ and Actical accelerometer data in 9–11year-old Canadian children. *Journal of sports sciences*, 35(6), 517-524. https://doi.org/10.1080/02640414.2016.1175653
- Brownson, R. C., Chang, J. J., Eyler, A. A., Ainsworth, B. E., Kirtland, K. A., Saelens, B. E., & Sallis, J. F. (2004). Measuring the environment for friendliness toward physical activity: a comparison of the reliability of 3 questionnaires. *American Journal of Public Health*, 94(3), 473–483. https://doi.org/10.2105/AJPH.94.3.473
- Cavagna, G. A., Franzetti, P., & Fuchimoto, T. (1983). The mechanics of walking in children. *The Journal of Physiology*, 343(1), 323–339. https://doi.org/10.1113/jphysiol.1983.sp014895
- Cerin, E., Saelens, B. E., Sallis, J. F., & Frank, L. D. (2006). Neighborhood Environment Walkability Scale: Validity and Development of a Short Form. *Med. Sci. Sports Exerc*, 38(9), 1682–1691. https://doi.org/10.1249/01.mss.0000227639.83607.4d
- Colley, R. C., Carson, V., Garriguet, D., Janssen, I., Roberts, K. C., & Tremblay, M. S.
 (2017). Physical activity of Canadian children and youth. *Health Reports*, 28(10), 8–16. https://doi.org/October 2017

- Colley, R. C., Garriguet, D., Janssen, I., Craig, C. L., Clarke, J., & Tremblay, M. S. (2011). Physical activity of Canadian children and youth: Accelerometer results from the 2007 to 2009 Canadian Health Measures Survey. *Statistics Canada*, 22(1).
- Comte, M., Hobin, E., Majumdar, S. R., Plotnikoff, R. C., Ball, G. D. C., McGavock, J., & Teams, M. and H. H. I. (2013). Patterns of weekday and weekend physical activity in youth in 2 Canadian provinces. *Applied Physiology, Nutrition, and Metabolism, 38*(2), 115–119. https://doi.org/10.1139/apnm-2012-0100
- De Bourdeaudhuij, I., Sallis, J. F., & Saelens, B. E. (2003). Environmental Correlates of Physical Activity in a Sample of Belgian Adults. *American Journal of Health Promotion*, 18(1), 83–92. https://doi.org/10.4278/0890-1171-18.1.83
- Esliger, D. W., & Tremblay, M. S. (2006). Technical reliability assessment of three accelerometer models in a mechanical setup. *Medicine and Science in Sports and Exercise*, 38(12), 2173–2181. https://doi.org/10.1249/01.mss.0000239394.55461.08
- ESRI. (2017). ArcGIS Desktop. Redlands, CA: Environmental Systems Research Institute.
- Harwell, M. R., & Gatti, G. G. (2001). Rescaling Ordinal Data to Interval Data in Educational Research. *Review of Educational Research*, 71(1), 105–131. https://doi.org/10.3102/00346543071001105
- IBM Corp. (2016). SPSS Software. New York: IBM Corp, United States.
- Janssen, I., Roberts, K. C., & Thompson, W. (2017). Is adherence to the Canadian 24-Hour Movement Behaviour Guidelines for Children and Youth associated with improved indicators of physical, mental, and social health? *Applied Physiology*, *Nutrition, and Metabolism*, 42(7), 725–731. https://doi.org/10.1139/apnm-2016-0681
- Mitchell, C., Clark, A., & Gilliland, J. (2016). Built Environment Influences of Children's Physical Activity: Examining Differences by Neighbourhood Size and Sex. *International Journal of Environmental Research and Public Health*, 13(1), 130.

https://doi.org/10.3390/ijerph13010130

- Puyau, M. R., Adolph, A. L., Vohra, F. A., Zakeri, I., & Butte, N. F. (2004). Prediction of Activity Energy Expenditure Using Accelerometers in Children. *Med. Sci. Sports Exerc*, 36(9), 1625–1631. https://doi.org/10.1249/01.MSS.0000139898.30804.60
- Rich, C., Geraci, M., Griffiths, L., Sera, F., Dezateux, C., & Cortina-Borja, M. (2013).
 Quality Control Methods in Accelerometer Data Processing: Defining Minimum
 Wear Time. *PLoS ONE*, 8(6), e67206. https://doi.org/10.1371/journal.pone.0067206
- Roberts, K. C., Yao, X., Carson, V., Chaput, J.-P., Janssen, I., & Tremblay, M. S. (2017). Meeting the Canadian 24-Hour Movement Guidelines for Children and Youth. *Statistics Canada*, 28(10), 3–7.
- Statistics Canada. (2016). CANSIM (database). Using CHASS (distributor). Retrieved June 11, 2018, from http://dc2.chass.utoronto.ca/cansim2
- Statistics Canada. (2012). Census Dictionary. Catalogue no. 98-301-X2011001. https://doi.org/92-566-X
- Tremblay, M. S., Carson, V., Chaput, J. P., Connor Gorber, S., Dinh, T., Duggan, M., ... & Janssen, I. (2016). Canadian 24-hour movement guidelines for children and youth: an integration of physical activity, sedentary behaviour, and sleep. *Applied Physiology, Nutrition, and Metabolism, 41*(6), S311-S327. https://doi.org/10.1139/apnm-2016-0151
- Trost, S. G., Pate, R. R., Freedson, P. S., Sallis, J. F., & Taylor, W. C. (2000). Using objective physical activity measures with youth: How many days of monitoring are needed? *Sci. Sports Exerc*, 32(2), 426–431.

Chapter 4

4 Context Matters: Measuring children's perceived barriers to physical activity across varying Canadian environments

4.1 Introduction

Recent research suggests less than 35% of Canadian children and youth are meeting the daily recommendation of 60 minutes of moderate to vigorous physical activity (MVPA) (Colley et al., 2017; ParticipACTION, 2018). It has been well established that low levels of physical activity (PA) among children is of serious concern because inactivity in childhood creates long-lasting health risks that track into adulthood (Bauman, 2003; Pate et al., 1999; Warburton, Nicol, & Bredin, 2006). It is widely accepted that an individual's PA behaviour is shaped by their interactions with their physical and social environments of daily life. This is the major tenet of the socio-ecological perspective; health is an outcome of the quality of the individual-environment fit (Grzywacz & Fuqua, 2000). It emphasizes that the intrapersonal and interpersonal factors, that form environmental perceptions of barriers to PA, may be as important for predicting behaviour as the objectively measured physical environment (Giles-Corti & Donovan, 2002; Hume, Salmon, & Ball, 2004). The primary objective of this study is to demonstrate how intrapersonal, interpersonal, and physical environment factors influence children's perceptions of barriers to PA. The secondary objective of this study is to utilize an expanded definition of urbanicity to determine the similarities and differences in children's perceptions in relation to the level of urbanicity of their home location.

4.1.1 Literature Review

Giles-Corti and Donovan (2002) have suggested a socio-ecological model of recreational PA which considers intrapersonal, interpersonal, and physical environment level determinants of PA behaviour. These different levels of environment operate through a reciprocal relationship whereby the individual's PA behaviour is affected by multiple levels of environmental influences, and PA behaviour shapes the surrounding social environment (Townsend & Foster, 2013). The intrapersonal environment is shaped by the influence of intrapersonal forces such as psychological and demographic factors, personal preferences and choices (Townsend & Foster, 2013). Factors within this level include gender, ethnicity, and age (Townsend & Foster, 2013). The interpersonal environment includes social networks, norms, supports, and standards, among groups, individuals, and organizations (Townsend & Foster, 2013). Main themes discussed in the literature as mechanisms of influence are socio-economic status, parental influence, and relationships with peers. These dimensions influence what an individual perceives as a barrier to PA in their environment, a key contributor to children's activity levels (Hume et al., 2004).

While research surrounding children's perceptions of PA barriers exists, there is little focusing on children in the diverse rural communities of Canada. This is important because evidence suggests there is an elevated risk for health concerns related to physical inactivity in rural versus urban Canadian communities due to differences in built environment and social factors (Moore, Brinkley, Crawford, Evenson, & Brownson, 2013). However, in a systematic review conducted by Sandercock, Angus, and Barton (2010), most studies indicated either no differences, or that urban children were less likely to be active than rural children. This is a problematic conclusion because a dichotomous division between urban and rural populations neglects to highlight heterogeneity of land uses. For example, when using a spectrum approach (i.e., considering more than a dichotomous rural versus urban relationship) to explaining the built environment, that includes suburban or small city populations, many researchers found rural *and* urban children were less active than children in suburban areas or small cities (Joens-Matre et al., 2008; Springer, Hoelscher, Castrucci, Perez, & Kelder, 2009, Springer, Hoelscher, & Kelder, 2006). This highlights the value of considering barriers to PA with a geographical classification system (Sandercock et al., 2010), such as a spectrum from rural to urban.

In order to understand how children engage in MVPA across Canada, there is a need to understand geographic variation in the determinants of activity (Orton et al., 2017). One approach that can be conceptualized is the use of children's daily contexts of living. Regarding health, context has been described as the circumstantial environment in which something takes place, and includes the interplay of the physical, social, cultural and structural environments coming together to shape the individual's experience (Orton et al., 2017; Williams, 2003). When considering the health and MVPA of children across varying environments, whether they are experienced at the intrapersonal, interpersonal or physical level, we must consider the interplay of the variety of factors that shape how behaviour is formed within interactions of daily life.

4.2 Methods

4.2.1 Data Source and Population

The study uses data from a larger population-based project investigating environmental influences on children's health and well-being, including PA and perceptions of barriers to PA. Study design has been described in detail elsewhere (Mitchell et al., 2016). This study involves participants from Southwestern and Northwestern Ontario and will be using data from surveys of youth and their parents, including responses to questions about socio-demographics, socio-economic status, and perceptions of the barriers for PA participation. This study was conducted in accordance with the Declaration of Helsinki and the Canadian Tri-Council Policy Statement; Ethical Conduct for Research Involving Humans and the protocol was approved by the University Non-Medical Research Ethics Board and the respective research officers and/or committees of the participating school boards. All children who participated in this study provided assent and were given parental consent.

Data was collected between 2010 and 2013 in 33 Southwestern Ontario schools, including 932 children in grades 5 to 8 (age 9-14) (66.9% participation rate). Schools were randomly selected and stratified by geographical context and neighbourhood socioeconomic status to ensure the participating sample was representative of the population in the region. In 2016, the study was replicated in four rural Northwestern Ontario schools (100% response rate), including 136 students in grades 4 to 8 (age 8-14) (70.1% participation rate).

A child participant's data was included in this study if it met three criteria: 1) completion of survey by the child participant; 2) completion of a corresponding survey by the child's
parent/guardian; and 3) identified postal code of their home location. Data for 892 (out of 1068) children met the inclusion criteria and were retained for analysis.

4.2.2 Dependent Variables

The dependent variables in this study are dimensions of children's perceptions of barriers to PA. The measures of children's perceptions of barriers are based on child survey questions assessing barriers to activity in the respondents' neighbourhood parks/playgrounds, trees in their neighbourhood and safety in their neighbourhood. A full list of the questions can be found in Table 1. Responses were provided on a 4-point Likert scale (strongly disagree, somewhat disagree, somewhat agree, strongly agree), but the 4-point data was recoded to binary variables (i.e., 0 for disagree, 1 for agree) to enhance validity of inference for this analysis (Harwell & Gatti, 2001). Four questions asked about the presence of facilitators and were reverse recoded to maintain consistency in this study (i.e., *do not* know people, *not* enough sidewalks, *not* enough bike lanes, *not* enough trees).

4.2.3 Independent Variables

The independent variables were identified in the PA literature and organized into levels of the social-ecological model: intrapersonal, interpersonal, and physical environment (Giles-Corti & Donovan, 2002). The intrapersonal variables include demographic data from the child survey. Gender is based on child self-identification and coded as a binary variable: girl (0) or boy (1). Age is a continuous variable measured in years. Visible minority is based on reported ethnicity and is coded as a binary variable: Caucasian (0) or non-Caucasian ethnicity (1).

Six variables are used to measure a child's interpersonal environment. Lone parent household is a binary variable defined as a child living with two parents (0) or one parent (1). Household arrangement is dichotomized into a child living in one home (0) or more than one home (1). Parental employment status is measured for both mother and father, with unemployed parents (including self-identified as unemployed, at home with children, students, or on disability/sick leave) as (0) and employed parents (including selfemployed, full-time employed, or part-time employed) as (1). Median Family Income (CAD) is the median family income from the 2011 National Household Survey measured at the census dissemination area in which the home is located and categorized as under \$60,000 (0); between \$60,000 - \$99,999 (1); and \$100,000 and more (2) (Statistics Canada, 2017a, 2017b).

The physical environment variable included in this study is the level of "urbanicity" in which the primary home of each child is located (Tillmann, Clark, & Gilliland, 2018). We categorize urbanicity into five classes: (1) Urban large-city, which includes areas that are characterized by grid-like road networks, high population density, and high land use mix within settlements greater than 100,000 people; (2) Suburban large-city, which includes areas that are characterized by irregular, looping and cul-de-sac road networks, lower population density, and low land use mix within settlements greater than 100,000 people; (3) Urban small-town, which includes settlement areas with a population between 10,000 and 100,000 people; (4) Rural small-town, which includes settlements with a population between 1,000 and 10,000; and (5) Rural areas, which are all other areas of our study area, with low population density and mostly characterized by agricultural land and natural areas. Home location was compared to this categorization of built form and population density spectrum to determine level of urbanicity, allowing for greater insight into location-based presence of environmental PA barriers.

4.2.4 Data Analysis

A series of logistic regression models with robust standard errors in STATA IC 15 (StataCorp., 2015) were used to compare what children consider to be barriers to their PA at varying levels of the socio-ecological model. Logistic regression was selected because it is more robust and has assumptions such as normal distribution or equal variance (Hosmer, Lemeshow, & Sturdivant, 2013). Odds ratios (OR) were calculated to examine associations between a variety of barriers and levels of the socio-ecological model. They were interpreted as the odds of agreeing with a barrier having influence on PA over disagreeing (Hilbe, 2011), and included robust standard error accounts for the observations biased due to clustering (such as within schools). Barriers children reported as influential were significant if $p \le 0.05$.

4.3 Results

Descriptive statistics are presented in Table 4.1 and the sample distribution of independent variables is presented in Table 4.2. To better categorize barriers, results were organized by themed barrier groups: barriers to safety (Table 4.3); social relationships (Table 4.4); and neighbourhood environment (Table 4.5). A total of 34 barriers were found to be significant, based on the associations of what children perceived as influential in their environments. While there was some variety in the patterning of results, all independent variables demonstrated a relationship with children's perceptions of barriers except paternal employment status. Full model results are presented in Tables 4.3-4.5.

Table 4.1 Survey questions measuring barriers, and sample distribution of

responses.

Question Measuring Barrier	Representing code in Table 4.3-4.5	Sample Size (N)	% Agreed
Perception of Safety			
There is so much traffic on streets near my home that it's difficult/unpleasant to bike or play on the street	Too much traffic	852	21.7
Most drivers go too fast while driving in our neighbourhood	Drive too Fast	848	37.4
I am worried about being or walking by myself in my neighbourhood and local streets because I am worried about being taken or hurt by a stranger	Worried about Strangers	851	18.9
There is a lot of crime in my neighbourhood (ex: strangers, gangs, drugs)	Crime	850	9.1
Perception of Social Factors			
There are no other kids to play with at parks/playgrounds in my neighbourhood	No one to play with	850	35.3
I get bullied or teased when I go to parks/playgrounds in my neighbourhood	Bullied at park	847	7.2
I have nobody to go with to parks/playgrounds in my neighbourhood	No one to go with	840	25.2
I [do not] know a lot of people in my neighbourhood	Do not know people	853	21.8
There are too many people/it feels too crowded at parks/playgrounds in my neighbourhood	Too crowded at park	849	15.1
Perception of Neighbourhood Environment			
Parks/playgrounds in my neighbourhood are too far from my house/takes too much time to get there	Too Far from Home	850	17.9
There is not enough room at parks/playgrounds in my neighbourhood for the activities I like	Not enough room	848	20.87
There is too much garbage/graffiti at parks/playgrounds in my neighbourhood	Garbage/Graffiti	850	13.9
There are [not] enough sidewalks on the street in my neighbourhood	Not enough sidewalks	847	37.5
There are [no] bicycle lanes or trails in or near my neighbourhood that are easy to get to	Not enough bike lanes	851	48.4

There are [not] a lot of trees along the streets in my neighbourhood	Not enough trees	852	23.9
There is no or not enough equipment or activities I like	Not enough equipment	848	32.2
Note: numbers may not add to full sample size due	e to missing values		

Table 4.2 Descriptive characteristics of independent variables

		% of
Independent Variable	Ν	Total
	(N= 892)
Intrapersonal Environment		
Gender		
Boy	396	44%
Age, mean years (Std. Err.)	11.1 (0.03)	
Visible minority	240	28%
Interpersonal Environment		
# of parents in main home		
Living with one parent	200	22%
Household Arrangement		
Live in more than one	144	16%
home	144	1070
Mother Employment Status		
Unemployed	138	16%
Father Employment Status		
Unemployed	54	6%
Median Family Income,		
CAD		
Middle Family Income, \$60,000 - \$99,9999	128	14%
High Family Income, \$100,000 and more	224	25%
Physical Environment		
Urbanicity		
Suburban Large City	399	45%
Urban Large City	83	9%
Urban Small-Town	80	9%
Rural Small-Town	147	16%
Rural	183	21%

66

4.3.1 Intrapersonal Factors

At the intrapersonal level, each independent variable demonstrated statistically significant influence on reporting perception of at least one barrier. Girls and visible minorities were more likely to report social barriers than their counterparts. Girls were 1.4 times more likely to report *No one to go with* (p = 0.044) than boys. Children who are visible minorities were 1.6 times more likely to report *Do not know people* (p = 0.016) as a barrier to PA than Caucasian children. Neighbourhood barriers were reported in two cases. With each increase in age by one year, children were more likely to report *Not enough room* as a barrier (p = 0.047). As well, children who are visible minorities were 1.5 times more likely to report *Not enough room for activities* (p = 0.006) compared to their counterparts. Two of three groups of children were more likely to report lack of safety as a barrier to their PA compared to their counterparts. With each increased year in age, older children report *Worried about strangers* (OR = 0.79, p = 0.014) and *Too much traffic* (OR = 0.84, p = 0.050). Girls are also 2.2 times more likely indicate *Worried about strangers* (p < 0.01), and 1.5 times more likely to report *Drive too fast* as a significant barrier (p = 0.013) compared to boys.

4.3.2 Interpersonal Factors

The interpersonal variables were found to have some significant relationships with reporting safety, social and neighbourhood barriers. The children who reported the most significant barriers were those whose mother is employed. These children reported experiencing the safety-related barrier of *Drive too fast* (OR = 0.61, p = 0.028), the social barrier of *Too crowded at park* (OR = 0.37, p = 0.019), and neighbourhood barriers of *Garbage/graffiti* (OR = 0.35, p = 0.005) and *Not enough equipment* (OR = 0.61, p = 0.047). A child's mother being unemployed and paternal employment status were not related to significant neighbourhood barriers to PA. Children in lone-parent households reported two significant neighbourhood barriers to PA, including *Not enough sidewalks* (OR = 1.46, p = 0.039) and *Not enough bike lanes* (OR = 1.48, p = 0.047) when compared to children in single-parent household. Children who lived in one home (versus multiple) also reported three significant barriers to PA, including *Too crowded at parks* (OR = 1.84, p = 0.042), *Not enough trees* (OR = 2.33, p = 0.004), and *Garbage/graffiti* (OR = 0.042).

2.33, p = 0.004). Three barriers were significantly related to parental income, where children in low income families were more likely to report perceiving barriers than those from middle income families. This included being 3.3 times more likely to report *Not enough room* (p = 0.004) and 2 times more likely to report *No one to play with* (p = 0.013) than children from middle income families. There were no significant differences in reporting barriers between children in low and high income families.

4.3.3 Physical Environment

Children in the rural areas experienced significant barriers related to their neighbourhood environment and local infrastructure for PA when compared to children in suburban areas. This was represented in three barriers, *Too far from home* (OR = 4.32, p < 0.000), Not enough sidewalks (OR = 3.91, p < 0.001) and Not enough bike lanes (OR = 2.06, p < 0.001) 0.001). On the other hand, suburban children were more likely to perceive barriers related to safety than the rural groups. This group of children was 3.5 times more likely to report *Crime* compared to rural (OR = 0.17, p < 0.001) and 5.9 times more likely than rural small-town (, p < 0.001) children. They were also 2.8 times more likely to report *Garbage/graffiti* compared to rural children (p = 0.008), 3.6 times more likely to report Too much traffic (p < 0.001) and 1.9 times more likely to report Do not know people (p =0.039) compared to rural small-town children. Comparing urban and suburban children, urban children were more likely to report *Too much traffic* (OR = 1.82, p = 0.017), Worried about strangers (OR = 2.01, p = 0.001), Do not know people (OR = 2.71, p =(0.002), Not enough room (OR = 1.97, p = 0.028), and Not enough equipment (OR = 1.68, p = 0.040). The only reported barrier to PA that remained significant for urban smalltown and rural small-town children was a *Not enough trees* (see Table 4.5).

 Table 4.3 Logistic regression models examining factors related to children perceiving safety barriers to physical activity.

	Too much ti	Too much traffic Drive too Fast		Fast	Worried about	Strangers	Crime	
	OR (Std Err)	р	OR (Std Err)	р	OR (Std Err)	р	OR (Std Err)	р
Intrapersonal Environment								
Boys	0.79 (0.14)	0.167	0.66 (0.11)	0.013*	0.45 (0.08)	0.000*	1.40 (0.38)	0.208
Age	0.84 (0.07)	0.050*	0.92 (0.06)	0.209	0.79 (0.07)	0.014*	0.96 (0.13)	0.735
Visible Minority	0.92 (0.16)	0.630	0.68 (0.14)	0.070	1.45 (0.31)	0.085	1.41 (0.37)	0.182
Interpersonal Environment								
Lone Parent Household	1.24 (0.36)	0.470	1.28 (0.28)	0.247	1.21 (0.29)	0.422	1.17 (0.24)	0.441
Live in one home	1.04 (0.25)	0.870	1.05 (0.24)	0.820	0.95 (0.26)	0.851	1.39 (0.45)	0.309
Mother unemployed	0.60 (0.16)	0.052	0.61 (0.14)	0.028*	1.08 (0.25)	0.734	0.62 (0.24)	0.215
Father unemployed	0.92 (0.19)	0.703	0.72 (0.19)	0.229	0.64 (0.21)	0.171	1.58 (1.00)	0.474
Family Income (ref: Low Family	Income)							
Middle Income	0.58 (0.21)	0.134	1.11 (0.28)	0.667	0.92 (0.34)	0.812	2.48 (1.31)	0.082
High Income	0.64 (0.18)	0.102	0.80 (0.17)	0.300	0.55 (0.20)	0.099	1.34 (0.77)	0.607
Physical Environment								
Urbanicity (ref: Suburban Large	City)							
Urban Large City	1.82 (0.46)	0.017*	1.67 (0.50)	0.086	2.01 (0.43)	0.001*	1.46 (0.40)	0.167
Urban Small-town	1.25 (0.28)	0.318	1.51 (0.35)	0.077	1.07 (0.32)	0.800	0.41 (0.40)	0.059
Rural Small-town	0.28 (0.75)	0.000*	1.21 (0.22)	0.301	0.82 (0.23)	0.464	0.29 (0.08)	0.000*
Rural	1.04 (0.28)	0.889	1.29 (0.30)	0.279	1.25 (0.43)	0.525	0.17 (0.07)	0.000*
Constant	2.41 (2.47)	0.4390	1.86 (1.36)	0.399	3.70 (4.13)	0.242	0.07 (0.12)	0.116
Pseudo R2	0.	0537	C	0.026	0	.0542	0.	.0856

	No one to play with		Bullied at	park	No one to go with		Do not kn people	low	Too crowded	Too crowded at park	
	OR (Std Err)	р	OR (Std Err)	р	OR (Std Err)	р	OR (Std Err)	р	OR (Std Err)	р	
Intrapersonal Environment											
Boys	0.77 (0.11)	0.080	1.17 (0.30)	0.533	0.71 (0.12)	0.044*	1.00 (0.20)	0.990	1.42 (0.26)	0.062	
Age	0.94 (0.08)	0.501	0.81 (0.11)	0.120	1.01 (0.10)	0.903	0.90 (0.07)	0.169	1.03 (0.08)	0.696	
Visible Minority	1.08 (0.16)	0.582	1.14 (0.39)	0.705	1.37 (0.23)	0.061	1.64 (0.33)	0.016*	1.44 (0.38)	0.160	
Interpersonal Environment											
Lone Parent Household	1.19 (0.21)	0.296	0.88 (0.53)	0.471	1.11 (0.26)	0.640	1.03 (0.22)	0.875	1.31 (0.37)	0.338	
Live in one home	1.17 (0.21)	0.715	0.31 (0.35)	0.743	1.19 (0.26)	0.415	0.89 (0.21)	0.606	1.84 (0.55)	0.042*	
Mother unemployed	0.62 (0.16)	0.070	0.68 (0.20)	0.063	0.78 (0.19)	0.313	1.10 (0.37)	0.772	0.37 (0.16)	0.019*	
Father unemployed	0.71 (0.24)	0.302	0.67 (0.30)	0.373	1.20 (0.38)	0.564	1.37 (0.54)	0.419	0.75 (0.40)	0.580	
Family Income (ref: Low	Family Income)										
Middle Income	0.50 (0.14)	0.013*	0.59 (0.41)	0.447	0.93 (0.30)	0.822	1.13 (0.37)	0.708	0.84 (0.36)	0.691	
High Income	0.90 (0.22)	0.435	0.62 (0.36)	0.419	1.50 (0.55)	0.272	1.04 (0.35)	0.904	0.78 (0.30)	0.514	
Physical Environment											
Urbanicity (ref: Suburban	Large City)										
Urban Large City	1.32 (0.33)	0.272	0.55 (0.26)	0.213	1.92 (0.65)	0.052	2.71 (0.85)	0.002*	0.98 (0.54)	0.974	
Urban Small-town	0.71 (0.27)	0.366	1.15 (0.64)	0.802	0.83 (0.22)	0.479	1.53 (0.38)	0.086	0.82 (0.30)	0.599	
Rural Small-town	0.83 (0.29)	0.584	1.47 (0.43)	0.186	1.32 (0.32)	0.255	0.53 (0.16)	0.039*	0.98 (0.23)	0.926	
Rural	1.22 (0.28)	0.372	0.72 (0.26)	0.363	1.54 (0.49)	0.175	0.95 (0.29)	0.876	1.14 (0.34)	0.699	
Constant	1.50 (1.52)	0.687	1.03 (1.74)	0.987	0.17 (0.18)	0.104	0.60 (0.50)	0.541	0.06 (0.06)	0.007	
Pseudo R2	0.020			0.055		0.030	0.04	17	0.035		

Table 4.4 Logistic regression models examining factors related to children perceiving social barriers to physical activity.

Table 4.5 Logistic regression models examining factors related to children perceiving neighbourhood environment	nt
barriers to physical activity.	

	Too Far from Home		Not enoug room	gh	Garbage/Gr	affiti	Not enou sidewalk	gh ts	Not enough lanes	bike	Not enoug trees	gh	Not enough eq	uipment
	OR(Std Er	r) p	OR(Std Err	r) p	OR(Std Err)	р	OR(Std Err) p	OR(Std Err)	р	OR(Std Err)	р	OR(Std Err)	р
Intrapersonal Envir	ronment													
Boys	0.70(0.13)	0.057	1.07(0.19)	0.695	0.99(0.24)	0.975	0.99(0.12)	0.926	0.87(0.12)	0.297	1.11(0.19)	0.565	1.01(0.14)	0.967
Age	0.83(0.09)	0.070	0.90(0.05)	0.047^{*}	0.89(0.06)	0.094	1.04(0.06)	0.501	0.93(0.09)	0.438	0.99(0.07)	0.927	1.08(0.07)	0.204
Visible Minority	0.97(0.21)	0.882	1.51(0.23)	0.006^*	1.06(0.22)	0.795	1.22(0.24)	0.308	1.08(0.14)	0.585	1.22(0.26)	0.349	1.31(0.23)	0.127
Interp	ersonal Envir	ronment	ţ											
Lone Parent Household	0.67(0.16)	0.090	0.95(0.27)	0.857	1.37(0.41)	0.293	1.46(0.26)	0.039*	1.48(0.29)	0.047*	* 1.51(0.35)	0.078	0.87(0.19)	0.521
Live in one home	0.69(0.20)	0.212	1.07(0.28)	0.793	2.33(0.68)	0.004^{*}	1.25(0.24)	0.237	1.24(0.26)	0.305	1.80(0.43)	0.014^{*}	1.00(0.21)	0.984
Mother unemployed	0.91(0.27)	0.757	0.62(0.16)	0.070	0.35(0.13)	0.005^{*}	0.92(0.23)	0.724	0.79(0.17)	0.293	0.76(0.17)	0.215	0.61(0.15)	0.047*
Father	1.12(0.49)	0.804	0.98(0.31)	0.938	1.88(0.90)	0.188	0.86(0.30)	0.674	1.07(0.31)	0.816	1.45(0.53)	0.308	0.79(0.23)	0.408
Family Income (ref: Income)	Low Family													
Middle Income	1.64(0.75)	0.280	0.30(0.13)	0.004*	0.82(0.29)	0.583	1.27(0.44)	0.497	0.80(0.18)	0.32	8 1.52(0.52)	0.222	2 0.73(0.23)	0.315
High Income	1.62(0.68)	0.249	0.57(0.20)	0.115	0.68(0.27)	0.340	0.93(0.20)	0.751	1.04(0.20)	0.85	8 0.99(0.26)	0.968	8 0.75(0.22)	0.333
Physi	ical Environn	nent												
Urbanicity (ref: Sub	urban Large	City)												
Urban Large City	1.18(0.39)	0.606	1.97(0.61)	0.028*	1.39(0.44)	0.305	1.06(0.20) 0.751	0.85(0.22)	0.52	3 0.86(0.3	1) 0.66	52 1.68(0.43)	0.040^{*}
Urban Small- town	1.73(0.63)	0.131	1.61(0.50)	0.123	1.26(0.47)	0.530	1.37(0.38) 0.256	5 1.64(0.65)	0.21	0 2.47(0.92) 0.01	6* 0.92(0.29)	0.795
Rural Small- town	1.49(0.73)	0.413	0.82(0.23)	0.472	0.63(0.16)	0.067	2.05(0.76) 0.053	3 1.18(0.18)	0.26	1.49(0.22	.) 0.00	7* 0.78(0.20)	0.321
Rural	4.32(1.67)	0.000^{*}	1.50(0.35)	0.077	0.36(0.13)	0.008^{*}	3.91(1.30) 0.000)* 2.06(0.34)	0.00	0* 1.46(0.47	0.23	9 1.09(0.29)	0.743
Constant	1.12(1.40)	0.926	0.83(0.62)	0.803	0.39(0.34)	0.286	0.16(0.12) 0.016	5 1.62(1.87)	0.67	7 0.11(0.09) 0.00	8 0.23(0.17)	0.048
Pseudo R2	0.08	5	0.04	1	0.05	57	0.05	52	0.02	4	0.0	30	0.0	15

4.4 Discussion

This study employed a series of logistic regression models to examine associations between children's perceptions of barriers to PA and different intrapersonal, interpersonal, and physical environmental factors. This study contributes to the literature on children's physical activity by examining and interpreting how children from a wide range of environmental contexts perceive barriers to PA. To our knowledge, this is the first Canadian study that considers PA barriers for a full range of geographic contexts (i.e., five levels of "urbanicity"). By demonstrating the differing impacts of the integrated social and environmental contexts in relation to the varying intrapersonal environments for diverse groups of children, we can better prioritize areas for mitigating barriers to PA, by targeting variation in the experience of place to most effectively address the issue.

There are several key findings in this study. First, girls perceived more barriers than boys and children who are visible minorities perceived more barriers than Caucasian children. Second, maternal employment status had a significant effect on children's perceptions of barriers while paternal employment status did not, and children in low-income families were more likely to report barriers than those in middle- or high-income families. Finally, children from every level of urbanicity reported different significant barriers. These results demonstrate variations in how barriers to activity are experienced based on a child's context of place. While there were some commonalities across environments, there are striking differences in the way that context influences what children perceive as barriers to their PA.

Previous research has reported mixed results on the importance of children's perceived safety in their community as a barrier to PA (Davison & Lawson, 2006). Although the present study did not link perceptions to actual PA, the findings support other studies that report perceived lack of safety has a negative association with PA levels (e.g., Gómez, Johnson, Selva, & Sallis, 2004). By considering different sub-themes of safety (i.e., traffic, crime, and strangers), the present study offers a deeper understanding into how perceptions of safety may act as a barrier to children's PA.

The present study found various aspects of traffic safety in the neighborhood were significant barriers to PA for girls, older children, and children whose mother was employed. This aligns with trends highlighted within a systematic review by Lee et al., (2015) who reported traffic safety was considered a barrier that decreases activity levels in studies they analyzed. A finding unique to this study however, is that children from suburban areas were more likely to report traffic related issues as a barrier to PA than rural small-town children. Based on this finding, policy initiatives in these areas should focus on strategies for traffic calming methods to reduce the threat of traffic speed in neighbourhoods (Larsen, Gilliland, & Hess, 2012).

Perception of crime was considered a significant barrier to PA for children in suburban large cities compared to rural small towns and rural areas. Additionally, children in urban areas were more likely to report being worried about being taken or hurt by a stranger in their neighbourhood than their suburban counterparts. Beets and Foley (2008) suggested that it may not be the actual measure of crime, but rather the perceived measure of safety in the environment that is important to children. This example shows how children's experience of place may be misaligned with the reality of their situation. As a result, research must focus on the ways children feel crime is present in their communities in addition to objective crime rates, to understand strategies for mitigating this as a barrier to PA.

Children whose mothers were employed reported five out of sixteen examined barriers had significant influences on their PA. On the other hand, none of the examined barriers were statistically significant for children whose mothers were unemployed; likewise, paternal employment status did not have a significant influence on children's responses to reporting barriers. These findings raise several new questions. How and why do the gendered patterns of parental employment status influence children's perception of barriers to PA? Furthermore, how is a child's experience of place influenced by their parent's employment? Future research may use qualitative methods to investigate more deeply why maternal employment status seems to matter more for children's PA perceptions than paternal employment. In response to objective two, this study used an expanded definition of urbanicity to determine the similarities and differences of children's perceptions in different geographical contexts. The investigation of context was related to categorization of physical environments by level 'urbanicity' taking into consideration dimensions of built form and population density. It was determined that children in the largest areas (urban large-city and suburban large-city) and the smallest areas (rural) reported the most barriers, however these differed relative to urbanicity. While children in urban and suburban areas reported issues related to safety and social barriers, children from the less populated areas consistently reported experiencing physical environment barriers to being physically active. Children in the rural areas reported absence of infrastructure and distance as the major barriers, while children in rural small towns and urban small towns reported these resources were present, but other forces such as neighbourhood aesthetics are perceived to influence their PA behaviour. This makes an important distinction of the differing needs of Canadian children in different contexts, highlighting the importance of considering rural children's variation in the experience of place, rather than one homogenous "not urban" population.

This study demonstrated the need for a place-specific approach to understanding the barriers children perceive as influential to their PA. By separating a dichotomous ruralurban definition into five levels in a spectrum, a distinction in the way children experience their environment demonstrates the importance of considering context specific definitions of both urban and rural spaces. In areas with characteristics similar to the urban and suburban areas of study, policy should focus on improving perceptions of safety and interventions should focus on drawing on social connections in the community to alleviate barriers of crime, garbage/graffiti and strangers.

The results of this study suggest policymakers in rural and urban small-towns should seek to engage children as stakeholders for improvements to the current infrastructure and improve the variety of available activities. Policy interventions for rural areas need to focus on opportunities for mitigating perceived distance through transportation opportunities, including active travel (i.e. sidewalks, bike trails) to improve independent mobility, promoting efficacy for children to access resources that currently exist (i.e. school playgrounds out of school time).

4.4.1 Strengths and Limitations

As with any research, this study presents several limitations that must be considered. First, there may have been other confounders at all levels of the socio-ecological model that were not accounted for in the models. This could include indicators for which we did not have specific measures or adequate data, such as household-level socio-economic status, or additional variables that could have been considered at the built environment level, such as access to recreation facilities and parks. This research focused on a more general environment level to build on the gaps of previous research; however, future work will look for significance with more specific built environment factors and identify their impact on PA levels. The second limitation was there was no consideration of facilitators to activity. While these are important to understand for promoting physical activity in children, we chose to focus our modelling on the presence of barriers. This is a potential area for future research. Additionally, the present study did not link the perception of barriers with a measurement of PA. Nevertheless, the aim of this study was to thoroughly examine children's perceptions of barriers, to provide a basis for future research that may examine how these factors represent actual barriers to children's PA. Future research should focus on examining how these perceived barriers relate to actual PA behaviour.

Despite the limitations mentioned above, it is important to note that the current study assessed how the perceptions of barriers to PA for a large sample of children varied in relation to multiple intrapersonal, interpersonal and physical environment factors across a wide variety of geographical contexts within Ontario. To the best of our knowledge, this is the first study assessing perceived barriers to PA, in such a large sample of children. Additionally, this is the first study to assess differences in perceptions of barriers to PA by place, at a five level urbanicity spectrum, demonstrating the value of this approach for future research especially in the Canadian context. This research highlights not only the heterogeneity of children's physical environments, but also the variety in how children experience barriers based on their intrapersonal and interpersonal environment. While results may be limited in generalizability due to the particular geographic context of

Southwestern and Northwestern Ontario, they provide an important new perspective on understanding children's physical activity behaviour in practice. A criticism of Canadian health policy and practice is the one-size-fits-all approach to targeting health issues, especially in rural areas (Markey, Lauzon, & Ryser, 2015). The present research provides evidence to suggest the variation in experiences of children in different places, highlighting the need for context-specific investigation and interventions. This work emphasizes the importance of gaining children's perspective as stakeholders in practice and policy, in order to effectively impact their environment for promoting health.

4.5 Conclusion

Future work must consider targeting barriers for specific groups in the formation of policy and practice to improve effectiveness of programs. Policy must focus on two main areas for alleviating barriers to outdoor PA; improve centralization of recreation in communities, and ensure infrastructure is relevant to user needs, accessible, aesthetically pleasing, and safe. For health professionals in practice, interventions should target girls and younger children focusing on discussions around plans for safety and the presence of crime in children's neighbourhoods. Finally, researchers should continue to recognize the heterogeneity of neighbourhoods, viewing more than an urban versus rural dichotomous definition and consider the implications of external factors on children's PA levels.

- Bauman, A. E. (2004). Updating the evidence that physical activity is good for health: an epidemiological review 2000–2003. *Journal of science and medicine in sport*, 7(1), 6-19.
- Beets, M. W., & Foley, J. T. (2008). Association of father involvement and neighborhood quality with kindergartners' physical activity: a multilevel structural equation model. *American Journal of Health Promotion*, 22(3), 195-203. https://doi.org/10.4278/ajhp.22.3.195
- Colley, R. C., Garriguet, D., Janssen, I., Craig, C. L., Clarke, J., & Tremblay, M. S. (2011). Physical activity of Canadian children and youth: accelerometer results from the 2007 to 2009 Canadian Health Measures Survey. *Health reports*, 22(1), 15.
- Davison, K. K., & Lawson, C. T. (2006). Do attributes in the physical environment influence children's physical activity? A review of the literature. *International journal of behavioral nutrition and physical activity*, 3(1), 19. https://doi.org/10.1186/1479-5868-3-19
- Giles-Corti, B., & Donovan, R. J. (2002). The relative influence of individual, social and physical environment determinants of physical activity. *Social science & medicine*, 54(12), 1793-1812.
- Gomez, J. E., Johnson, B. A., Selva, M., & Sallis, J. F. (2004). Violent crime and outdoor physical activity among inner-city youth. *Preventive medicine*, 39(5), 876-881. https://doi.org/10.1016/j.ypmed.2004.03.019
- Grzywacz, J. G., & Fuqua, J. (2000). The social ecology of health: Leverage points and linkages. *Behavioral medicine*, 26(3), 101-115. https://doi.org/10.1080/08964280009595758
- Harwell, M. R., & Gatti, G. G. (2001). Rescaling ordinal data to interval data in educational research. *Review of Educational Research*, 71(1), 105-131. https://doi.org/10.3102/00346543071001105

- Hilbe, J.M., (2011). Logistic Regression, in: International Encyclopedia of Statistical Science. Springer Berlin Heidelberg, Berlin, Germany, pp. 755–758. https://doi.org/10.1007/978-3-642-04898-2_344
- Hosmer Jr, D. W., Lemeshow, S., & Sturdivant, R. X. (2013). Applied logistic regression (Vol. 398). John Wiley & Sons.
- Hume, C., Salmon, J., & Ball, K. (2004). Children's perceptions of their home and neighborhood environments, and their association with objectively measured physical activity: a qualitative and quantitative study. *Health education research*, 20(1), 1-13.https://doi.org/10.1093/her/cyg095
- Joens-Matre, R. R., Welk, G. J., Calabro, M. A., Russell, D. W., Nicklay, E., & Hensley, L. D. (2008). Rural–urban differences in physical activity, physical fitness, and overweight prevalence of children. *The Journal of rural health*, 24(1), 49-54.https://doi.org/10.1111/j.1748-0361.2008.00136.x
- Larsen, K., Gilliland, J., & Hess, P. M. (2012). Route-based analysis to capture the environmental influences on a child's mode of travel between home and school. *Annals of the Association of American Geographers*, *102*(6), 1348-1365. https://doi.org/10.1080/00045608.2011.627059
- Lee, H., Tamminen, K. A., Clark, A. M., Slater, L., Spence, J. C., & Holt, N. L. (2015). A meta-study of qualitative research examining determinants of children's independent active free play. *International Journal of Behavioral Nutrition and Physical Activity*, 12(1), 5. https://doi.org/10.1186/s12966-015-0165-9
- Markey, B., Lauzon, G., & Ryser, M. (Eds.). (2015). State of Rural Canada Report. *Canadian Rural Revitalization Foundation*. Retrieved November, 27, 2017.
- Mitchell, C. A., Clark, A. F., & Gilliland, J. A. (2016). Built environment influences of children's physical activity: examining differences by neighbourhood size and sex. *International journal of environmental research and public health*, 13(1), 130.

Moore, J. B., Brinkley, J., Crawford, T. W., Evenson, K. R., & Brownson, R. C. (2013).

Association of the built environment with physical activity and adiposity in rural and urban youth. *Preventive medicine*, *56*(2), 145-148.https://doi.org/10.1016/j.ypmed.2012.11.019

- Orton, L., Halliday, E., Collins, M., Egan, M., Lewis, S., Ponsford, R., ... & Popay, J. (2017). Putting context centre stage: evidence from a systems evaluation of an area based empowerment initiative in England. *Critical Public Health*, 27(4), 477-489. https://doi.org/10.1080/09581596.2016.1250868
- ParticipACTION. (2018). The Brain + Body Equation: Canadian kids need active bodies to build their best brains. The 2018 ParticipACTION Report Card on Physical Activity for Children and Youth. Toronto, ON. Retrieved from https://www.participaction.com/en-ca/thought-leadership/report-card/2018
- Pate, R. R., Trost, S. G., Dowda, M., Ott, A. E., Ward, D. S., Saunders, R., & Felton, G. (1999). Tracking of physical activity, physical inactivity, and health-related physical fitness in rural youth. *Pediatric Exercise Science*, 11(4), 364-376.
- Sandercock, G., Angus, C., & Barton, J. (2010). Physical activity levels of children living in different built environments. *Preventive Medicine*, 50(4), 193-198. https://doi.org/10.1016/j.ypmed.2010.01.005
- Springer, A.E., Hoelscher, D.M., Castrucci, B., Perez, A., Kelder, S.H., 2009. Prevalence of physical activity and sedentary behaviors by metropolitan status in 4th-, 8th-, and 11th-grade students in Texas, 2004-2005. *Prev. Chronic Dis. 6*, A21.
- Springer, A. E., Hoelscher, D. M., & Kelder, S. H. (2006). Prevalence of physical activity and sedentary behaviors in US high school students by metropolitan status and geographic region. *Journal of physical activity and health*, 3(4), 365-380.https://doi.org/10.1123/jpah.3.4.365

StataCorp., 2015. Stata Statistical Software. College Station, TX: StataCorp LP.

Statistics Canada, 2017a. *Thunder Bay [Population centre], Ontario and Ontario [Province]* (table). Census Profile. 2016 Census. [WWW Document].

https://doi.org/Catalogue no. 98-316-X2016001

- Statistics Canada, 2017b. London [Population centre], Ontario and Ontario [Province] (table). Census Profile. [WWW Document]. https://doi.org/Catalogue no. 98-316-X2016001.
- Tillmann, S., Clark, A. F., & Gilliland, J. A. (2018). Children and Nature: Linking Accessibility of Natural Environments and Children's Health-Related Quality of Life. *International Journal of Environmental Research and Public Health*, 15(6), 1072. https://doi.org/10.3390/ijerph15061072
- Townsend, N., & Foster, C. (2013). Developing and applying a socio-ecological model to the promotion of healthy eating in the school. *Public health nutrition*, 16(6), 1101-1108. https://doi.org/10.1017/S1368980011002655
- Warburton, D. E., Nicol, C. W., & Bredin, S. S. (2006). Health benefits of physical activity: the evidence. *Canadian medical association journal*, 174(6), 801-809. https://doi.org/10.1503/cmaj.051351
- Williams, G. H. (2003). The determinants of health: structure, context and agency. Sociology of Health & Illness, 25(3), 131-154. https://doi.org/10.1111/1467-9566.00344

Chapter 5

5 Exploring the effect of perceptions on children's physical activity in varying geographic contexts: a structural equation modelling approach.

5.1 Background

The ongoing trend of low levels of physical activity in Canadian children is a concern for population health (Colley et al., 2017; ParticipACTION, 2018), as physical activity participation is associated with many physical, mental and social health benefits (Janssen & LeBlanc, 2010; Poitras et al., 2016; Strong et al., 2005). Over the past two decades, research has consistently demonstrated strong evidence of positive linear relationships between type, duration and intensity of physical activity and a variety of health outcomes, prompting recommendations for increasing regular physical activity as a health promotion and disease prevention strategy in children (Poitras et al., 2016; Strong et al., 2005; Warburton, Nicol, & Bredin, 2006; Biddle, Gorely, & Stensel, 2004; Fletcher et al., 1996). Physical activity, especially moderate-to-vigorous physical activity (MVPA) has been associated with benefits related to adiposity, cardiovascular health, brain development, musculoskeletal health and fitness, pro-social behaviour, academic achievement, and quality of life improvements, for children and youth (Jiménez-Pavón, Kelly, & Reilly, 2010; Poitras et al., 2016; ParticipACTION, 2018; Strong et al., 2005). The current Canadian guidelines for physical activity recommend children achieve 60 minutes of MVPA per day to achieve these optimal health benefits (Tremblay et al., 2016).

Despite the wide variety of potential benefits to children's health, only 33% of Canadian children achieve the recommended weekly average of at least 60 minutes of MVPA each day (Colley et al., 2017). A number of individual factors have been associated with the achievement of physical activity guidelines. These include ethnicity, adiposity, education or literacy, preference and choice (Sallis, Prochaska, & Taylor, 2000;Van Der Horst, Paw, Twisk, & Van Mechelen, 2007). While the influence of these individual factors demonstrate mixed results and clarity of these relationships is needed, three factors have

consistently and independently been associated with children's activity levels: age, gender and socioeconomic status. The literature has established that as children get older, they are less likely to be physically active, and achieve physical activity standards (Colley et al., 2017; Dumith, Gigante, Domingues, & Kohl, 2011). It has been well documented that girls are less likely to be physically active than boys (Colley et al., 2017; Dumith et al., 2011; Sallis et al., 1992). Finally, with increasing socioeconomic status, activity levels increase and sedentary time decreases (Epstein et al., 2006; Gebremariam et al., 2015; Sallis et al., 1992; Van Der Horst et al., 2007). In addition to these individual factors, physical environment factors play an important role in formation of activity behaviour.

Associations between children's ability to be physically active and their surrounding physical environments have been widely discussed. Recent systematic reviews examining the relationship between the physical environment and children's physical activity indicate relationships with physical environment factors, including a combination of the built and natural environments (Ding et al., 2011; Martins et al., 2017; Oliveira et al., 2014). Physical activity levels are not just influenced by their objective physical environment; as explained by Orstad, McDonough, Stapleton, Altincekic, and Troped (2017), physical activity behaviour also depends on the individual's perception (cognitive representation) of their physical environment. These perceptions are formed through their experiential context. Context is the environment in which children live, including objective measures that can be measured and evaluated such as population density and built form; but context also includes the interplay of the physical, social, cultural and structural forces to which they are exposed (Orton et al., 2017; Williams, 2003). Orstad et al. (2017) explain that children's perceptions of their surrounding environment develops through a cyclical process that is interactive with social, cognitive and affective experiences. Research has indicated that one's perceptions of their environment may be more important than the physical environment alone for predicting physical activity behaviour (Carroll-Scott et al., 2013; Hume et al., 2004; Orstad et al., 2017). The purpose of this study is to assess the mediating effect of children's perceptions of barriers to activity on the relationship between their environments and MVPA.

When considering issues or opportunities that hinder/encourage physical activity, a valuable way to consider the environmental influences on children's perceptions is through recognition of barriers to physical activity. Based on an individual's experience of context, children living in the same physical environment may experience differences in perception of barriers relative to their interactions with their environment. Three groups of barriers consistently demonstrate an influence on physical activity: neighbourhood, social, and safety barriers. Neighbourhood barriers are usually related to availability of and/or accessibility to physical activity resources in a child's environment and have consistently demonstrated an effect on activity levels. This could include issues due to distance to facilities, transportation options, and residential density or design factors, and the presence/amount of age appropriate equipment/activities/landscape design features for activity in a child's community (Davison & Lawson, 2006; Ding et al., 2011; Oliveira, Moreira, Abreu, Mota, & Santos, 2014, Martins et al., 2017). Social barriers are forces that exist formally and informally throughout children's neighbourhoods and have demonstrated significant influence on the physical activity of children throughout the literature. Mechanisms of influence include parental influence, and relationships with peers positively (i.e. presence of friends) and negatively (i.e. bullying) (Abraham et al., 2010; Beets & Foley, 2008; Hume, Salmon, & Ball, 2004; Jago et al., 2009; Lee et al., 2015; Sherar et al., 2009).

Throughout the physical activity literature, safety barriers have been interpreted in various ways and have demonstrated mixed results. Barriers include presence or fear of strangers, loose animals, traffic dangers, poor neighbourhood infrastructure, and crime (Davison & Lawson, 2006; Loebach & Gilliland, 2010; Smith & Barker, 2001). Beets and Foley (2008) suggest it may not be the presence of actual environmental characteristics that directly affect safety influencing physical activity levels, but rather the perceptions of neighbourhood characteristics that promote safety perhaps are more influential in decisions to participate in physical activity. Previous work by the authors (Chapter 4) sought to understand how these barriers were associated with children's environments. It was determined that 34 barriers related to safety, social relationships, and the neighbourhood were perceived to have influenced physical activity, and these perceptions differed related to children's environmental contexts.

It has been well established the environment and perceptions of barriers impact children's MVPA (Chapter 2), and that children perceive barriers in their environments differently (Chapter 4); while there is very little known about how the perceptions of barriers to physical activity alter the relationship between the physical environment and MVPA. To fill this gap, this paper will examine if children's perceptions of barriers to MVPA mediate the relationship between children's contexts and their MVPA behaviour. This research will provide valuable information to take a direct approach to targeting the MVPA of Canadian children and youth (Barnes & Tremblay, 2017). Furthermore, while research exists assessing of the relationship between subjective environmental barriers to physical activity with objective physical activity, results in the literature primarily focus on populations in large urban or mid-sized cities, especially outside of Canada (Davison & Lawson, 2006; Ding et al., 2011; Martins et al., 2017). Based on the heterogeneous nature of the Canadian context, it is important to acknowledge the lack of generalizability of previous work to children's health in rural areas of this country. The present study will aim to address the paucity of research discussing children outside of large urban centers, by incorporating a spectrum measurement tool to assess the physical environment at multiple levels of urbanicity. This is one of the first studies in physical activity literature on Canadian children to take such an approach.

On the basis of the literature and evidence reviewed, the major hypothesis of this study was that children's perceived barriers to physical activity mediate the relationship between physical environment of their home neighbourhood with physical activity levels. The secondary hypothesis was that all three themed barrier scales (i.e., social, neighbourhood, and safety barriers) would have a significant effect on MVPA, based on existing literature and the findings of Chapter 4; however, these relationships would vary in intensity. We hypothesized perceived social barriers would have the strongest mediating effect between the physical environment and MVPA. This is because social factors such as neighbourhood social cohesion, relationships with neighbors, and availability of spontaneous group social activities have consistently demonstrated a positive association with children's physical activity levels in urban and rural subsamples of children (Aarts, Wendel-Vos, Van Oers, Van De Goor, & Schuit, 2010; Walia & Liepert, 2012). Perceived neighbourhood and safety barriers were hypothesized to have

the mediating effects to a lesser extent. While evidence of a relationship between perceiving greater barriers within these themes and experiencing lower physical activity levels does exist (Grow et al., 2008; Holt et al., 2016; Hume et al., 2004; Loebach & Gilliland, 2010; Yousefian et al., 2009), these forces are context specific and can change based on personal factors such as perceived self-efficacy for engaging (Ryan & Dzewaltowski, 2002), or external forces such as parental rules and local policies (Holt et al., 2016; Lee et al., 2015; Loebach & Gilliland, 2010; Ou et al., 2016; Yousefian et al., 2009).

5.2 Methods

This study draws from the Spatial Temporal Environment and Activity Monitoring (STEAM) project, a multi-year mixed methods research study (2010-2016) that investigates the environmental influences on the health and well-being of children ages 8 to 14 years. The data collection took place in two study locations, in Southwestern Ontario (2010-2013) and in Northwestern Ontario (2016). Schools in Southwestern Ontario were randomly selected based on socio-economic status and urbanicity of the school environment and all of the schools in the Northwestern Ontario community were selected to participate. All selected schools were invited to participate and enrolled through the principal. Children in grades 5 through 8, plus grade 4 children in Northwestern Ontario, were invited to participate once they received informed parental consent and provided their own informed assent. The STEAM protocol was approved by the University Non-Medical Research Ethics Board and the respective research officers of the participating school boards. Details of the project recruitment process can be found elsewhere (Loebach & Gilliland, 2016; Mitchell, Clark, & Gilliland, 2016).

For each cohort of students, data was collected over two seasons to allow for an examination of the impact of seasonality on children's mobility and health-related behaviours. This study focused on one season from each cohort to ensure the general seasonality is comparable between groups of children: spring (2010-2013) in the South and fall (2016) in the North. This study uses data provided by passive-GPS tracking, accelerometers, and the youth survey. The GPS monitor was worn by the participants

during all waking hours for up to 8 days and used in this study to identify spatiallyaccurate home locations for each child. Participants were also asked to wear an accelerometer, to objectively measure their activity levels, for eight consecutive days (4-6 weekdays and 2-3 weekend days) for all waking hours, removing it only for sleeping, bathing, and swimming. Finally, participants were asked to complete a detailed survey that asked children about demographics, and perceptions about their barriers to physical activity.

The initial dataset used for this study included 1,068 children from 33 schools across Ontario. Before conducting any analyses, a series of inclusion criteria were developed to ensure the quality and completeness of the observations used. The first criteria was that participants must have at least four days of accelerometer data with a minimum of 10 hours of valid wear time (Trost et al., 2000), and at least 1 valid weekday, and 1 valid weekend day (n=565) (Comte et al., 2013; Rich et al., 2013). Non-wear time was classified as 60 or more minutes of motionless bouts, and was excluded from analysis (Puyau et al., 2004). The second criterion was that participants must have completed questions on the youth survey about age, gender, and perceptions of barriers to PA (n=892). The final criterion was that a valid home location must be determined by the GPS data. Applying all the inclusion criteria to the dataset left a final dataset has 546 children (62% of all children in the sample) with complete data.

5.2.1 Measurement Instruments

5.2.1.1 Outcome Variable

The outcome variable used in this study was an objective measure of MVPA, defined as the average number of minutes children spend in MVPA across all valid days (Tremblay et al., 2016). The outcome variable was measured using a portable Actical® Z accelerometer that participants wore on their right hip (so as to not impede activity) attached with a nylon-elastic band. This device was calibrated to measure energy expenditure at 60-second epochs, providing an index of physical activity intensity throughout the course of wear time (Heil, 2013). MVPA movement thresholds were defined as 1,500 or more activity counts per minute (Puyau et al., 2004).

While there is no consistent gold standard for minimum thresholds for measuring accurate PA, the inclusion/exclusion criteria of 4 valid days with at least one weekday and one weekend day is found to be an acceptable threshold in the literature (Comte et al., 2013). A 4-5 day monitoring period has a test-retest reliability of 0.8 among children (grade 1 to 6), and 0.7 among adolescents (grade 7 to 12) (Trost et al., 2000). One valid weekday and one valid weekend day is required to ensure the differences in physical activity behaviour between weekdays and weekend days are accounted for when measuring average MVPA. Requiring both types of days created a better representation of physical activity levels for each participant across an entire week (Comte et al., 2013; Rich et al., 2013).

5.2.1.2 Exposure Variable

Previous work by Taylor, Clark and Gilliland (Chapter 4) demonstrated the need for considering more than a dichotomous urban-rural definition when analyzing the influence of children's environments on perceptions of barriers to physical activity. This study used objective measures of population density and intersection density to develop an urbanicity index, which is a spectrum approach considering the heterogeneity of built form and land uses, while providing an objective tool for classifying data (Babey et al., 2015). The urbanicity index was created based on the sum of z-scores of both population density and intersection density around the home location for each child. Population density was measured by identifying the number of people per square kilometer within each home location's census dissemination block. Intersection density was measured by the number of 4-way intersections per square kilometer within 500-metres of each home location.

5.2.1.3 Mediator

The mediating variables in the model were children's reported perceptions of barriers to their physical activity. These barriers were measured by the child survey, with a full list of questions found in Table 1. Survey questions were adapted from the validated *Neighbourhood Environment Walkability Survey* (Brownson et al., 2004). Additional questions were developed based on background relationships identified in the literature, use in previous studies, or to measure necessary socio-demographic characteristics of the participants. Four questions asked about the presence of facilitators and were reverse

recoded to maintain consistency in this study (i.e., *do not* know people, *not* enough sidewalks, *not* enough bike lanes, *not* enough trees). The survey was conducted with 4-point Likert-type questions (strongly disagree, somewhat disagree, somewhat agree, strongly agree), but the Likert-type data was recoded to three thematically defined groups to assess children's responses as Likert scales (see Table 5.1). Each score has a minimum of four questions, which were combined into a single composite score for each participant to provide a quantitative interval measurement scale (i.e. 1 for *strongly disagree*, 2 for *disagree*, 3 for *agree*, 4 for *strongly agree*) (Boone & Boone, 2012). This tool was used to consider the responses as continuous variables within the structural equation modeling.

 Table 5.1 Survey questions measuring barriers, and the corresponding themed groups.

Perceptions	Corresponding Survey Questions
(Themed Groups)	corresponding survey Questions
Neighbourhood Barriers (Neighbourhood)	 Parks/playgrounds in my neighbourhood are too far from my house/takes too much time to get there There is not enough room at parks/playgrounds in my neighbourhood for the activities I like There is too much garbage/graffiti at parks/playgrounds in my neighbourhood There are [not] enough sidewalks on the street in my neighbourhood There are [no] bicycle lanes or trails in or near my neighbourhood that are easy to get to There are [not] a lot of trees along the streets in my neighbourhood
	heighbourhood
<i>Social Barriers</i> (Social)	 There are no other kids to play with at parks/playgrounds in my neighbourhood I get bullied or teased when I go to parks/playgrounds in my neighbourhood I have nobody to go with to parks/playgrounds in my neighbourhood I [do not] know a lot of people in my neighbourhood There are too many people/it feels too crowded at parks/playgrounds in my neighbourhood
Safety Barriers (Safety)	 There is so much traffic on streets near my home that it's difficult/unpleasant to bike or play on the street Most drivers go too fast while driving in our neighbourhood

3.	I am worried about being or walking by myself in my
	neighbourhood and local streets because I am worried
	about being taken or hurt by a stranger
4.	There is a lot of crime in my neighbourhood (ex:
	strangers, gangs, drugs)

5.2.1.4 Effect Modifier and Co-Variates

The model used gender as an effect modifier, as it was hypothesized that the magnitude of effect of the exposure urbanicity mediated by perceptions of barriers on MVPA would vary according to a child's gender. The hypothesis was made because girls are more likely to perceive significantly more barriers to physical activity than boys (Chapter 4) and boys achieve significantly more minutes of MVPA than girls (Colley et al., 2017). Median household income (MHHI in Figure 5.1) and age (age in Figure 5.1) are included in the model as control variables due to their strong explanatory power with both barriers to physical activity and MVPA. MHHI (in Canadian Dollars) is measured at the Census dissemination area that a child's home is located within. Data from the 2011 National Household survey was used for Southwestern Ontario and 2016 Census on Canada was used for Northwestern Ontario. Age, measured as a continuous variable in years, was assessed in the child survey as a demographic question.



NOTE: In SEM, observed variables are demonstrated with squares, and relationship pathways are demonstrated with arrows. As explained by Gunzler, Chen, Wu, and Zhang (2013), straight arrows with a single head indicate a causal relation from the base to the head. A curved arrow with two heads indicates a potential association between variables.

Figure 5.1 Conceptual model.

This model will be run twice with gender as the effect modifier. The red path indicates indirect measurement of the main research question: what is the effect of children's perceptions on the relationship between varying geographic contexts of the physical environment and MVPA behaviour?

5.2.2 Statistical Analysis

To assess the fit of the hypothesized model (Figure 5.1) to the data collected from participants we employed structural equation modeling (SEM). SEM allows researchers to test multiple regression equations simultaneously but instead of assuming a perfect relationship between all independent variables (as in regression), measurement error is accounted for in the model (Hoyle, 1995). SEM makes the assumption that all variables

are additive in a linear relationship, assessing the direct and indirect effects of the variables within the model (Hoyle, 1995). Data cleaning and preliminary analyses to test the data quality were conducted in SPSS (IBM Corp. 2016). Missing data were handled with full-information techniques. Statistical significance was determined at p < 0.05. Model testing was conducted in Mplus 7.4 (Muthén & Muthén, 2015). Model fit was not tested because it was a saturated model, therefore all possible pathways were included.

5.3 Results

The relationship between all of the measured variables within the model are presented in Table 5.2. The specific mediating effect measuring the main research question is presented in Table 5.3. Results are descriptively presented separately for girls and boys below.

			Model 1: Girls Model 2: Bo			l 2: Boys					
			b	se	z	<i>p</i> - Value		b	se	Z	<i>p</i> -Value
Neighbourhood	on	MVPA	0.52	1.79	0.29	0.77	-4	5.63	2.66	-2.11	0.04*
Social			-4.30	1.66	-2.59	0.01*	-(0.38	2.54	-0.15	0.88
Safety			0.01	0.02	0.80	0.43	C	0.02	0.02	0.96	0.33
Urbanicity			-0.85	0.84	-1.01	0.31	-3	3.10	1.31	-2.37	0.02*
Age			-2.36	1.17	-2.01	0.04*	_4	4.96	2.09	-2.38	0.02*
MHHI			-0.53	0.50	-1.06	0.29	-(0.15	0.70	-0.22	0.83
Perceptions											
Urbanicity	on	Neighbourhood	-0.06	0.03	-2.02	0.04*	-(0.07	0.04	-2.03	0.04*
Age			0.01	0.04	0.34	0.74	-(0.04	0.06	-0.74	0.46
MHHI			-0.03	0.02	-1.86	0.06	-(0.05	0.02	-2.39	0.02*
Urbanicity	on	Social	0.02	0.03	0.50	0.62	C	0.00	0.04	-0.00	0.00*
Age			-0.02	0.05	-0.40	0.69	-(0.00	0.06	-0.06	0.95
MHHI			-0.01	0.02	-0.35	0.73	-(0.03	0.02	-1.27	0.21
Urbanicity	on	Safety	-0.22	3.10	-0.07	0.94	4	4.01	4.30	0.94	0.35
Age			6.87	4.37	1.57	0.12	C).49	6.95	0.07	0.94
MHHI			2.09	1.86	1.13	0.26	2	2.56	2.29	1.12	0.26
Social	with	Neighbourhood	0.29	0.04	7.59	0.00*	C).27	0.05	6.12	0.00*
Safety			-1.02	3.32	-0.31	0.76	-]	1.85	0.65	-0.40	0.69
Safety	with	Social	4.08	3.59	1.14	0.26	-1	0.11	4.97	-2.04	0.04*

 Table 5.2 Results of the relationship between all variables within the model.

* Indicates significant results $p \le 0.05$ NOTE: "on" signifies a one way relationship, "with" signifies association between variables

				Moo	del 1: Girl	S	Model 2: Boys				
			b	se	Z	<i>p</i> -Value	b	se	Z	<i>p</i> -Value	
Effects from Urban to	MVP	A									
Total			-0.95	0.84	-1.13	0.26	-2.62	1.31	-2.00	0.05*	
Total Indirect			-0.10	0.18	-0.56	0.57	0.49	0.31	1.57	0.01*	
Direct											
Urbanicity	on	MVPA	-0.85	0.84	-1.01	0.31	-3.10	1.31	-2.37	0.02*	
Indirect											
Urbanicity to Neigh.	to	MVPA	-0.03	0.11	-0.29	0.77	0.41	0.28	1.47	0.14	
Urbanicity to Social	to	MVPA	-0.07	0.14	-0.49	0.63	0.00	0.014	0.00	1.00	
Urbanicity to Safety	to	MVPA	-0.00	0.04	-0.07	0.94	0.08	0.12	0.67	0.50	

Table 5.3 Results of the mediating effect and direct relationships of urbanicity and perceptions on MVPA.

* Indicates significant results $p \le 0.05$ Note: "on" signifies a one way relationship, "to" signifies mediating relationship

5.3.1 Girls

Across both Ontario study areas, a total of 316 participants identified as a girl. When analyzed within the model, four relationships remained significant. These relationships are seen in Table 5.2, Model 1. MVPA was significantly negatively related to two factors. With each year increase in age, MVPA decreased by about 2.4 minutes (p = 0.04). As well, with each increase in the likelihood of reporting perception of social barriers, MVPA decreased by 4.3 minutes (p = 0.01). With increasing urbanicity, girls were significantly less likely to report perceiving neighbourhood barriers (p = 0.04). Finally, girls who reported neighbourhood barriers were significantly more likely to report social barriers. These results can be visually described in Figure 5.2, where the model was broken down to represent the significant relationships only, and the positive or negative nature of the association.



Figure 5.2 Significant positive/negative association results for model, grouped by girls.

While there are relationships between factors within the model, perceptions of neighbourhood, social and safety barriers did not significantly mediate the relationship between urbanicity and MVPA in the final model. This is demonstrated in Table 5.3, Model 1. Additionally, when controlling for all relationships (i.e. *Total* in Table 5.3 Model 1), and independent of all other factors, there was no significant relationship between urbanicity and MVPA in girls.

5.3.2 Boys

Across both study areas, the total sample included 230 participants who identified as a boy. When analyzed within the model, seven relationships remained significant. These relationships are seen in Table 5.2, Model 2. MVPA had a significant negative relation with three factors. Each year increase in age caused MVPA to decline approximately 5 minutes (p = 0.02). With each increase in urbanicity on the spectrum, MVPA decreased by 3.1 minutes (p = 0.02). As well, with increasing perceptions of neighbourhood barriers, boys' MVPA declined by 5.6 minutes (p = 0.04). Boys were significantly less likely to report perceptions of neighbourhood barriers as the urbanicity of their home neighbourhood increased (p = 0.04), and as their MHHI increased (p = 0.02). Reporting perceptions of social barriers (p < 0.00), as was perceiving social and safety barriers (p = 0.04). These results can be visually described in Figure 5.3, where the model was broken down to represent the significant relationships only, and the positive or negative nature of the association.



Figure 5.3 Significant positive/negative association results for model, grouped by boys.

In the final model, as demonstrated in Table 5.3, Model 2, there is a significant overall effect in the relationship between urbanicity and MVPA while accounting for the other relationships in the model, where urbanicity increases by one unit, MVPA decreases by 2.6 minutes (p = 0.05). There is also a significant direct relationship between urbanicity and MVPA, such that as urbanicity increases, MVPA decreases by 3.1 minutes (p = 0.02). In the final model for boys, perceptions of neighbourhood, social, and safety barriers did not significantly mediate the relationship between urbanicity and MVPA.

5.4 Discussion

This study employed SEM to examine influences of children's perception and urbanicity on MVPA. Previous research shows (including in Chapter 4) that children's perceptions significantly differ based on their varying environments (Davison & Lawson, 2006; Moore et al., 2013). The present study suggests that these perceptions and environments affect objectively measured MVPA, but perceptions do not mediate the relationship between the physical environment and activity levels. This study contributes to the literature by furthering the understanding how urbanicity, barriers, and the interaction of the two impact children's physical activity behaviours. Furthermore, to our knowledge, this was one of the first studies in the Canadian physical activity literature to utilize the spectrum approach to assessing the urbanicity of children's home locations. With further applications, this method could prove to be a beneficial tool for assessing the heterogeneous Canadian geographic contexts.

The primary hypothesis of this research was: children's perceived neighbourhood, social, and safety barriers to physical activity would mediate the relationship of physical environment of their neighbourhood with physical activity levels. Based on the results of our models, this hypothesis was not supported. The results indicated that barriers and the physical environment have a significant interaction, and independently of each other have significant influences on MVPA, but perceptions of barriers do not mediate the relationship between the environment and physical activity in the study population. The results also indicate that barriers for MVPA differ for girls and boys.

A secondary hypothesis was that social barriers would have the strongest mediating effect between the physical environment and MVPA. While social barriers did not present a mediating relationship, they did independently have a significant effect on the MVPA levels of girls. These results demonstrate that regardless of urbanicity, girls report this barrier to physical activity. Social factors and barriers influencing physical activity, especially in girls, have been well studied in the literature (Bocarro et al., 2015; Dwyer et al., 2006; Pawlowski, Tjørnhøj-Thomsen, Schipperijn, & Troelsen, 2014; Sallis, Prochaska, Taylor, Hill, & Geraci, 1999; Spencer, Rehman, & Kirk, 2015). Qualitative research by Pawlowski et al. (2014) provides depth to this relationship, indicating girls reported having no one to play with, conflict, and peer influence were issues *why* social barriers influenced their physical activity. Given the ongoing disparities in levels of activity relative to gender, health promotion efforts must focus on alleviating social barriers in structured activity opportunities, to decrease barriers for increasing MVPA in
girls (Telford, Telford, Olive, Cochrane, & Davey, 2016). Based on the evidence of Chapter 4, these activities must be context specific.

While the hypothesis that perceived neighbourhood barriers would have a mediating effect in the relationship between urbanicity and MVPA was not supported, neighbourhood barriers were significantly associated with urbanicity and MVPA for boys. The relationship between neighbourhood barriers, urbanicity, and MVPA is complex, as each are negatively associated with each other. Higher levels of MVPA are associated with lower urbanicity and perceiving fewer neighbourhood barriers. Although perceiving issues of distance, availability, and accessibility of neighbourhood resources in rural areas may not be surprising, researchers should seek to understand why boys in these areas can overcome these barriers to achieve higher levels of PA than their urbandwelling counterparts. Furthermore, practitioners and researchers should consider the ways in which rural boys achieve more MVPA minutes than their counterparts. This may include activities not discussed in the traditional self-report physical activity assessments (i.e. farm chores) (Davis et al., 2008). These findings could be a beneficial starting point to determining the disparities in activity minutes based on home location of boys, despite the increased number of opportunities in increasingly urban settings.

Findings related to the effect of the co-variate age on MVPA were supportive of the findings Chapter 4. In both girls and boys, as age in years increased, minutes of MVPA decreased. This echoes recent research by Colley et al. (2017) that found the same pattern in a national sample of Canadian children. While previous work determined that older children were significantly more likely than younger children to report specific barriers to physical activity, the present study did not find age significantly related to barriers when using composite scores to assess perceptions. Future work could consider examining the interplay of age in the formation of perceptions, and how this may change over time with longitudinal monitoring. In practice, it is important to develop interventions to increase MVPA in older children. As the amount of free play and physical activity during the school day decreases, health practitioners must look at strategies for engaging older children in continuation of activity habits, and provide new opportunities to continue building an appreciation for physical activity (Gilliland et al, 2015).

5.4.1 Strengths and Limitations

This study has limitations that warrant attention. Firstly, the present study only modelled MVPA behaviour. While this is the level of behaviour recommended to achieve the maximum health benefits of activity (Colley et al., 2017), recent Canadian physical activity research has suggested considering the importance of different levels of activity and sedentary behaviour (Poitras et al., 2016; Tremblay et al., 2016). Future Canadian research should consider assessing the mediating effect of children's perceptions of barriers with these multiple levels of activity to assess influences on activity achievement across the whole day. A further limitation was using dissemination area-level MHHI as the indicator for income, rather than parents' self-report information. This could have led to a potential misrepresentation of income of the study population, and a lack of significant results. This strategy was used because a large majority of parents elected not to report their income on the parent survey, and we were unable to impute this information. One final limitation is that there may have been measures within our model that were unaccounted for. Based on the complexity of relationships in the formation of MVPA behaviour, this will be an issue with any physical activity study. Our model was based on substantive evidence in the three areas and focused on a more basic hypothesis of the mitigating relationship of perceptions, to build on the gaps of previous research.

Despite the limitations, this research laid the groundwork for future research to continue to consider the complex interaction of children's perceptions, how they are formed within the environment, and their effect on physical activity. This study also has several strengths worth mentioning. For example, although we cannot make casual inferences for the mediating effect of perceptions on children's physical activity in every case, it does provide a foundation for elucidating the relationship. This is of critical importance due to the conflicting results regarding the relevance of perceptions for influencing physical activity in children (Haughton-McNeill, Wyrwich, Brownson, Clark, & Kreuter, 2006). This was a novel and rigorous approach to assessing this relationship, using a large and diverse sample of Canadian children from two geographically distinct areas of Ontario. This study also demonstrated the possibility for using an urbanicity spectrum and the value when assessing issues related to children's physical activity, accounting for the

limitations identified in the literature related to using a dichotomous rural versus urban definition of location (Babey et al., 2015; Jones-Smith & Popkin, 2010; Sandercock et al., 2010). Finally, while a mediating relationship was not statistically significant, this study filled a gap of Chapter 4, demonstrating that perceptions do have significant influence on objectively measured MVPA. This sets the stage for future research to consider these and additional barriers in diverse environments and populations, and the implications that perceptions may have on meeting the *Canadian 24-Hour Movement Guidelines* (Tremblay et al., 2016).

5.5 Conclusions

The purpose of this study was to improve efforts for quantifying the experience of children's daily activity contexts, by assessing the mediating effect of perceptions of barriers on the relationship between their environments and MVPA. These results offer insight into potential processes by which perceptions are shaped and impact MVPA and provide initial information to investigate these relationships further in future research. These findings suggest health promotion efforts will be most effective if they consider multipronged approaches directed toward place-specific experiences of barriers, especially targeting social barriers with girls and neighbourhood barriers with boys. The present study supports previous arguments that assessments of the objective environment are not enough to change children's physical activity behaviour (Orstad et al., 2017), and that researchers must improve efforts for quantifying the experience of children's daily activity contexts. This work highlighted the necessity for children's physical activity researchers in Canada to consider new ways for assessing similarities and differences in rural and urban populations. To our knowledge, this was one of the first studies in Canadian physical activity literature to utilize the spectrum approach to assessing the relationships between urbanicity and the experience of barriers. With further applications and improvements, this method could prove as a beneficial tool for objectively assessing the heterogeneous Canadian geography, its impact on children's experience of barriers, and their physical activity levels.

5.6 References

- Aarts, M. J., Wendel-Vos, W., Van Oers, H. A. M., Van De Goor, I. A. M., & Schuit, A. J. (2010). Environmental Determinants of Outdoor Play in Children A Large-Scale Cross-Sectional Study. *Am J Prev Med*, *39*(3), 212–219. https://doi.org/10.1016/j.amepre.2010.05.008
- Abraham, A., Sommerhalder, K., & Abel, T. (2010). Landscape and well-being: a scoping study on the health-promoting impact of outdoor environments. *International Journal of Public Health*, 55(1), 59–69.
 https://doi.org/10.1007/s00038-009-0069-z
- Babey, S. H., Tan, D., Wolstein, J., & Diamant, A. L. (2015). Neighborhood, family and individual characteristics related to adolescent park-based physical activity. *Preventive medicine*, 76, 31-36. https://doi.org/10.1016/j.ypmed.2015.04.001
- Beets, M. W., & Foley, J. T. (2008). Association of Father Involvement and Neighborhood Quality with Kindergartners' Physical Activity: A Multilevel Structural Equation Model. *American Journal of Health Promotion*, 22(3), 195–203. https://doi.org/10.4278/ajhp.22.3.195
- Biddle, S. J., Gorely, T., & Stensel, D. J. (2004). Health-enhancing physical activity and sedentary behaviour in children and adolescents. *Journal of Sports Sciences*, 22(8), 679–701. https://doi.org/10.1080/02640410410001712412
- Bocarro, J. N., Floyd, M. F., Smith, W. R., Edwards, M. B., Schultz, C. L., Baran, P., ... Suau, L. J. (2015). Social and environmental factors related to boys' and girls' parkbased physical activity. *Preventing Chronic Disease*, *12*, E97. https://doi.org/10.5888/pcd12.140532
- Boone, H. N., & Boone, D. A. (2012). Analyzing likert data. *Journal of extension*, 50(2), 1-5.
- Carroll-Scott, A., Gilstad-Hayden, K., Rosenthal, L., Peters, S. M., McCaslin, C., Joyce,R., & Ickovics, J. R. (2013). Disentangling neighborhood contextual associations

with child body mass index, diet, and physical activity: The role of built, socioeconomic, and social environments. *Social Science & Medicine*, *95*, 106–114. https://doi.org/10.1016/J.SOCSCIMED.2013.04.003

- Colley, R. C., Carson, V., Garriguet, D., Janssen, I., Roberts, K. C., & Tremblay, M. S. (2017). Physical activity of Canadian children and youth. *Health Reports*, 28(10), 8–16.
- Comte, M., Hobin, E., Majumdar, S. R., Plotnikoff, R. C., Ball, G. D. C., McGavock, J., & Teams, M. and H. H. I. (2013). Patterns of weekday and weekend physical activity in youth in 2 Canadian provinces. *Applied Physiology, Nutrition, and Metabolism, 38*(2), 115–119. https://doi.org/10.1139/apnm-2012-0100
- Davis, A. M., Boles, R. E., James, R. L., Sullivan, D. K., Donnelly, J. E., Swirczynski, D. L., & Goetz, J. (2008). Health behaviors and weight status among urban and rural children. *Rural and Remote Health*, 8(2), 810. Retrieved from http://www.ncbi.nlm.nih.gov/pubmed/18426334
- Davison, K., & Lawson, C. T. (2006). Do attributes in the physical environment influence children's physical activity? A review of the literature. *International Journal of Behavioral Nutrition and Physical Activity*, 3(1), 19. https://doi.org/10.1186/1479-5868-3-19
- Ding, D., Sallis, J. F., Kerr, J., Lee, S., & Rosenberg, D. E. (2011). Neighborhood Environment and Physical Activity Among Youth. *American Journal of Preventive Medicine*, 41(4), 442–455. https://doi.org/10.1016/j.amepre.2011.06.036
- Dumith, S. C., Gigante, D. P., Domingues, M. R., & Kohl, H. W. (2011). Physical activity change during adolescence: a systematic review and a pooled analysis. *International Journal of Epidemiology*, 40(3), 685–698. https://doi.org/10.1093/ije/dyq272
- Dwyer, J. J. M., Allison, K. R., Goldenberg, E. R., Fein, A. J., Yoshida, K. K., & Boutilier, M. A. (2006). Adolescent girls' perceived barriers to participation in

physical activity. Adolescence, 41(161), 75–89.

- Epstein, L. H., Raja, S., Gold, S. S., Paluch, R. A., Pak, Y., & Roemmich, J. N. (2006). Reducing Sedentary Behavior. *Psychological Science*, 17(8), 654–659. https://doi.org/10.1111/j.1467-9280.2006.01761.x
- Fletcher, G. F., Blair, S. N., Blumenthal, J., Caspersen, C., Chaitman, B., Epstein, S., ... Pollock, M. L. (1996). Statement on exercise. Benefits and recommendations for physical activity programs for all Americans. *Circulation*, 86(1), 340–344. https://doi.org/10.1161/01.cir.94.4.857
- Gebremariam, M. K., Altenburg, T. M., Lakerveld, J., Andersen, L. F., Stronks, K., Chinapaw, M. J., & Lien, N. (2015). Associations between socioeconomic position and correlates of sedentary behaviour among youth: a systematic review. *Obesity Reviews*, 16(11), 988–1000. https://doi.org/10.1111/obr.12314
- Grow, H. M., Saelens, B. E., Kerr, J., Durant, N. H., Norman, G. J., & Sallis, J. F. (2008).
 Where Are Youth Active? Roles of Proximity, Active Transport, and Built Environment. *Med. Sci. Sports Exerc*, 40(12), 2071–2079. https://doi.org/10.1249/MSS.0b013e3181817baa
- Gunzler, D., Chen, T., Wu, P., & Zhang, H. (2013). Introduction to mediation analysis with structural equation modeling. *Shanghai Archives of Psychiatry*, 25(6), 390–394. https://doi.org/10.3969/j.issn.1002-0829.2013.06.009
- Haughton-McNeill, L., Wyrwich, K. W., Brownson, R. C., Clark, E. M., & Kreuter, M.
 W. (2006). Individual, social environmental, and physical environmental influences on physical activity among black and white Adults: A Structural Equation Analysis. *Annals of Behavioral Medicine*, *31*(1), 36–44. https://doi.org/10.1207/s15324796abm3101_7
- Heil, D. P. (2013). Predicting Activity Energy Expenditure Using the Actical® Activity Monitor. *Research Quarterly for Exercise and Sport*, 77(1), 64–80. https://doi.org/10.1080/02701367.2006.10599333

- Holt, N. L., Neely, K. C., Spence, J. C., Carson, V., Pynn, S. R., Boyd, K. A., ...
 Robinson, Z. (2016). An intergenerational study of perceptions of changes in active free play among families from rural areas of Western Canada. *BMC Public Health*, *16*(1). https://doi.org/10.1186/s12889-016-3490-2
- Hoyle, R. H. (1995). Structural Equation Modeling: Concepts, Issues, and Applications.Thousand Oaks, CA: Sage. R.
- Hume, C., Salmon, J., & Ball, K. (2004). Children's perceptions of their home and neighborhood environments, and their association with objectively measured physical activity: a qualitative and quantitative study. *Health education research*, 20(1), 1-13. https://doi.org/10.1093/her/cyg095

IBM Corp. (2016). SPSS Software. New York: IBM Corp, United States.

- Jago, R., Thompson, J. L., Page, A. S., Brockman, R., Cartwright, K., & Fox, K. R. (2009). Licence to be active: parental concerns and 10–11-year-old children's ability to be independently physically active. *Journal of Public Health*, *31*(4), 472–477. https://doi.org/10.1093/pubmed/fdp053
- Janssen, I., & LeBlanc, A. G. (2010). Systematic review of the health benefits of physical activity and fitness in school-aged children and youth. *International Journal of Behavioral Nutrition and Physical Activity*, 7(1), 40. https://doi.org/10.1186/1479-5868-7-40
- Jiménez-Pavón, D., Kelly, J., & Reilly, J. J. (2010). Associations between objectively measured habitual physical activity and adiposity in children and adolescents: Systematic review. *International Journal of Pediatric Obesity*, 5(1), 3–18. https://doi.org/10.3109/17477160903067601
- Poitras, V. J., Gray, C. E., Borghese, M. M., Carson, V., Chaput, J. P., Janssen, I., ... & Sampson, M. (2016). Systematic review of the relationships between objectively measured physical activity and health indicators in school-aged children and youth. *Applied Physiology, Nutrition, and Metabolism, 41*(6), S197-S239.

- Jones-Smith, J. C., & Popkin, B. M. (2010). Understanding community context and adult health changes in China: Development of an urbanicity scale. *Social Science & Medicine*, 71, 1436–1446. https://doi.org/10.1016/j.socscimed.2010.07.027
- Lee, H., Tamminen, K. A., Clark, A. M., Slater, L., Spence, J. C., & Holt, N. L. (2015). A meta-study of qualitative research examining determinants of children's independent active free play. *International Journal of Behavioral Nutrition and Physical Activity*, 12(1), 5. https://doi.org/10.1186/s12966-015-0165-9
- Loebach, J. E., & Gilliland, J. A. (2016). Free Range Kids? Using GPS-Derived Activity Spaces to Examine Children's Neighborhood Activity and Mobility. *Environment and Behavior*, 48(3), 421–453. https://doi.org/10.1177/0013916514543177
- Loebach, J., & Gilliland, J. (2010). Child-led tours to uncover children's perceptions and use of neighborhood environments. *Children Youth and Environments*, 20(1), 52-90. https://doi.org/10.7721/chilyoutenvi.20.1.0052
- Martins, J., Marques, A., Peralta, M., Palmeira, A. L., Palmeira, A., & Carreiro Da Costa, F. (2017). Correlates of physical activity in young people: A narrative review of reviews. Implications for physical education based on a socio-ecological approach. *Retos, Nuevas Tendencias En Educación Física, Deporte Y Recreación, 31*, 292–299.
- Mitchell, C., Clark, A., & Gilliland, J. (2016). Built Environment Influences of Children's Physical Activity: Examining Differences by Neighbourhood Size and Sex. *International Journal of Environmental Research and Public Health*, 13(1), 130. https://doi.org/10.3390/ijerph13010130
- Moore, J. B., Brinkley, J., Crawford, T. W., Evenson, K. R., & Brownson, R. C. (2013). Association of the built environment with physical activity and adiposity in rural and urban youth. *Preventive Medicine*, 56, 145–148. https://doi.org/10.1016/j.ypmed.2012.11.019

Muthén, B. O., & Muthén, L. K. (2015). Mplus. In The comprehensive modelling

program for applied researchers: user's guide (5th ed.). Retrieved from http://www.statmodel.com/virg_nov_course.shtml

- Oliveira, A. F., Moreira, C., Abreu, S., Mota, J., & Santos, R. (2014). Environmental determinants of physical activity in children: A systematic review. Archives of Exercise in Health and Disease, 4(2), 254–261. https://doi.org/10.5628/aehd.v4i2.158
- Orstad, S. L., McDonough, M. H., Stapleton, S., Altincekic, C., & Troped, P. J. (2017). A Systematic Review of Agreement Between Perceived and Objective Neighborhood Environment Measures and Associations With Physical Activity Outcomes. *Environment and Behavior*, 49(8), 904–932. https://doi.org/10.1177/0013916516670982
- Orton, L., Halliday, E., Collins, M., Egan, M., Lewis, S., Ponsford, R., ... Popay, J. (2017). Putting context centre stage: evidence from a systems evaluation of an area based empowerment initiative in England. *Critical Public Health*, 27(4), 477–489. https://doi.org/10.1080/09581596.2016.1250868
- Ou, J., Levy, J., Peters, J., Bongiovanni, R., Garcia-Soto, J., Medina, R., & Scammell, M. (2016). A Walk in the Park: The Influence of Urban Parks and Community Violence on Physical Activity in Chelsea, MA. *International Journal of Environmental Research and Public Health*, 13(1), 97. https://doi.org/10.3390/ijerph13010097
- ParticipACTION. (2018). The Brain + Body Equation: Canadian kids need active bodies to build their best brains. The 2018 ParticipACTION Report Card on Physical Activity for Children and Youth. Toronto, ON. Retrieved from https://www.participaction.com
- Pawlowski, C. S., Tjørnhøj-Thomsen, T., Schipperijn, J., & Troelsen, J. (2014). Barriers for recess physical activity: a gender specific qualitative focus group exploration. *BMC Public Health*, 14(1), 639. https://doi.org/10.1186/1471-2458-14-639

Puyau, M. R., Adolph, A. L., Vohra, F. A., Zakeri, I., & Butte, N. F. (2004). Prediction of

Activity Energy Expenditure Using Accelerometers in Children. *Med. Sci. Sports Exerc*, *36*(9), 1625–1631. https://doi.org/10.1249/01.MSS.0000139898.30804.60

- Rich, C., Geraci, M., Griffiths, L., Sera, F., Dezateux, C., & Cortina-Borja, M. (2013).
 Quality Control Methods in Accelerometer Data Processing: Defining Minimum
 Wear Time. *PLoS ONE*, 8(6), e67206. https://doi.org/10.1371/journal.pone.0067206
- Ryan, G. J., & Dzewaltowski, D. A. (2002). Comparing the Relationships between
 Different Types of Self-Efficacy and Physical Activity in Youth. *Health Education*& *Behavior*, 29(4), 491–504. https://doi.org/10.1177/109019810202900408
- Sallis, J. F., Prochaska, J. J., & Taylor, W. C. (2000). A review of correlates of physical activity of children and adolescents. *Med. Sci. Sports Exerc.*, 32(5), 963–975.
- Sallis, J. F., Prochaska, J. J., Taylor, W. C., Hill, J. O., & Geraci, J. C. (1999). Correlates of physical activity in a national sample of girls and boys in Grades 4 through 12. *Health Psychology*, 18(4), 410–415. https://doi.org/10.1037/0278-6133.18.4.410
- Sallis, J. F., Simons-Morton, B. G., Epstein, L. H., Faucette, N., Iannotti, R. J., Killen, J. D., ... Taylor, W. C. (1992). Determinants of physical activity and intervention in youth. *Medicine and Science in Sports and Excercise*, 24(6), S248–S258.
- Sandercock, G., Angus, C., & Barton, J. (2010). Physical activity levels of children living in different built environments. *Preventive Medicine*, 50(4), 193–198. https://doi.org/10.1016/j.ypmed.2010.01.005
- Sherar, L. B., Gyurcsik, N. C., Humbert, M. L., Dyck, R. F., Fowler-Kerry, S., Baxter-Jones, A. D. G., ... Baxter-Jones, A. D. G. (2009). Activity and Barriers in Girls (8– 16 yr) Based on Grade and Maturity Status. *Med. Sci. Sports Exerc*, 41(1), 87–95. https://doi.org/10.1249/MSS.0b013e31818457e6
- Smith, F., & Barker, J. (2001). Commodifying the countryside: the impact of out-ofschool care on rural landscapes of children's play. *Area*, 33(2), 169–176. https://doi.org/10.1111/1475-4762.00020

- Spencer, R. A., Rehman, L., & Kirk, S. (2015). Understanding gender norms, nutrition, and physical activity in adolescent girls: a scoping review. *International Journal of Behavioral Nutrition and Physical Activity*, 12(1), 6. https://doi.org/10.1186/s12966-015-0166-8
- Strong, W. B., Malina, R. M., Blimkie, C. J. R., Daniels, S. R., Dishman, R. K., Gutin, B., ... Trudeau, F. (2005). Evidence based physical activity for school-age youth. *The Journal of Pediatrics*, 146(6), 732–737. https://doi.org/10.1016/j.jpeds.2005.01.055
- Telford, R. M., Telford, R. D., Olive, L. S., Cochrane, T., & Davey, R. (2016). Why Are girls less physically active than boys? findings from the LOOK longitudinal study. *PloS One*, 11(3), e0150041. https://doi.org/10.1371/journal.pone.0150041
- Tremblay, M. S., Carson, V., Chaput, J.-P., Connor Gorber, S., Dinh, T., Duggan, M., ...
 Zehr Canadian, L. (2016). Canadian 24-Hour Movement Guidelines for Children and Youth: An Integration of Physical Activity, Sedentary Behaviour, and Sleep 1. *Appl. Physiol. Nutr. Metab. Appl. Physiol. Nutr. Metab*, 4118(18), 311–327. https://doi.org/10.1139/apnm-2016-0151
- Trost, S. G., Pate, R. R., Freedson, P. S., Sallis, J. F., & Taylor, W. C. (2000). Using objective physical activity measures with youth: How many days of monitoring are needed? *Sci. Sports Exerc*, 32(2), 426–431.
- Van Der Horst, K., Paw, M. J. C. A., Twisk, J. W. R., & Van Mechelen, W. (2007). A brief review on correlates of physical activity and sedentariness in youth. *Medicine* and Science in Sports and Exercise, 39(8), 1241–1250. https://doi.org/10.1249/mss.0b013e318059bf35
- Walia, S., & Liepert, B. (2012). Perceived facilitators and barriers to physical activity for rural youth: an exploratory study using photovoice. *Rural and Remote Health*, 12(1842).
- Warburton, D. E. R., Nicol, C. W., & Bredin, S. S. D. (2006). Health benefits of physical activity: the evidence. *Canadian Medical Association Journal*, 174(6), 801–809.

https://doi.org/10.1503/cmaj.051351

- Williams, G. H. (2003). The determinants of health: structure, context and agency. Sociology of Health & Illness, 25(3), 131–154. https://doi.org/10.1111/1467-9566.00344
- Yousefian, A., Ziller, E., Swartz, J., & Hartley, D. (2009). Active Living for Rural Youth: Addressing Physical Inactivity in Rural Communities. *Journal of Public Health Management and Practice*, 15(3), 223–231. https://doi.org/10.1097/PHH.0b013e3181a11822

Chapter 6

6 Synthesis

The purpose of this final chapter is to summarize and discuss the research presented in this thesis, which investigated barriers to children's physical activity in their daily environments. This thesis examined how children's perceptions and environments affected their physical activity levels and suggested strategies for alleviating these barriers to promote physical activity. The following research questions were explored:

- 1) How do factors of children's intrapersonal, interpersonal, and physical environments influence their perceptions of barriers to PA?
- 2) What are the similarities and differences in children's perceptions of barriers to PA in relation to the level of urbanicity and geographic variation of their home location?
- 3) Do children's perceptions of barriers to PA mediate the relationship between their physical environment (urbanicity) and their MVPA levels?

The chapter consists of six sections: Section 6.1 summarizes the two independent studies described in Chapters 4 and 5, Section 6.2 discusses the research and methodologic contributions of this thesis to the literature on children's PA, Section 6.3 outlines the limitations of this thesis, and Section 6.4 provides an overview of possible policy and practice implications. Section 6.5 provides suggestions for future research, and Section 6.6 will conclude the chapter.

6.1 Summary of Studies

This thesis considers the geographic variation in how children's physical activity behaviour is influenced by barriers experienced in their everyday environments. Study 1 (Chapter 4) examined children's experience of barriers to physical activity based on what their reports of what they perceived to be influential. In order to capture the context by which perceptions are shaped, the socio-ecological framework of recreational physical activity was applied to consider factors of intrapersonal, interpersonal and physical environment when reporting barriers. The intrapersonal environment included gender, age and visible minority status. The interpersonal environment included the number of parents at home, household arrangement, maternal and paternal employment status, and family income. The physical environment included five categories of urbanicity based on the built form and population density of the participants' neighbourhoods. Binary logistic regression models and odds ratios were used to measure the relationship between children's reports of perceiving barriers with the various factors of the socio-ecological framework.

The study found at the intrapersonal level, girls were more likely to report safety and social barriers than boys. As children increased in age, they were more likely to report barriers related to safety and the neighbourhood environment. Visible minority children were more likely to report barriers in the social and neighbourhood environment. At the interpersonal level, children with employed mothers reported the most barriers. Children in low income families were more likely to report barriers than those in middle- or highincome families. At the physical environment level, children in urban neighbourhoods within large-cities, suburban neighbourhoods within large-cities, and rural areas were more likely to report barriers, however these differed relative to population size. Children in the smallest rural areas reported barriers of the neighbourhood environment such as a lack of infrastructure. Children in the highest populated areas were most concerned with social and safety barriers, such as not knowing people or worrying about strangers and crime. The results supported previous work by demonstrating that children's perceptions are formed within their environments (Babey et al., 2015; Davison & Lawson, 2006; Hume et al., 2004; Lee et al., 2015; Smith et al., 2015), and expanded on this literature by suggesting one's socio-ecologic context has an impact on the extent to which these barriers are considered influential. By demonstrating the differing impacts of the integrated social and environmental contexts in relation to the varying intrapersonal environments for different groups of children, this study advocated for tailoring approaches to best align alleviating barriers within children's contexts. This will allow for prioritizing target areas for mitigating barriers to PA in order to most effectively address the issue.

Elaborating on the insights and methods of Study 1, Study 2 (Chapter 5) sought to examine the mediating effect of perceiving these barriers in the relationship between varying geographic contexts and MVPA behaviour. This was completed using a structural equation model. Participants from Study 1 were carried over to Study 2, the only additional criteria applied for inclusion was 4 valid days of accelerometer data collection (including one weekday and one weekend day). In this study, the urbanicity scale was the exposure variable, MVPA (measured with accelerometers) was the outcome variable, three themed perception scores were the mediators, gender was the effect modifier and two variables were controlled for: age and median household income.

The results of Study 2 suggested environments and perceptions have a significant effect on the MVPA levels of boys and girls, however, the hypothesis was not supported because perceptions did not significantly mediate the relationship between urbanicity and MVPA. The results also indicate that the influence of factors affecting MVPA differ for boys and girls. Amongst boys, the primary finding of perceived neighbourhood barriers was associated with lower levels of activity, although it should be noted that this perception was more common with boys in rural areas. These boys from rural areas, however, had higher physical activity levels than their urban counterparts. Regardless of urbanicity, MVPA significantly decreased for girls when they perceived that there were social barriers to their activity. As age increased, minutes of MVPA decreased for both boys and girls. While associations between children's ability to be physically active and their surrounding physical environments have been widely examined in the literature (Ding et al., 2011; Martins et al., 2017; Oliveira, Moreira, Abreu, Mota, & Santos, 2014), there is little research regarding how the perceptions of barriers to physical activity alter the relationship between the physical environment and MVPA. This study contributes to the literature by furthering the understanding how urbanicity, barriers, and the interaction of the two impact children's physical activity behaviours. Chapter 5 highlighted the necessity for children's physical activity researchers in Canada to improve efforts for quantifying the experience of children's daily activity contexts.

6.2 Research Contributions

The present studies demonstrate the importance of considering all the factors that shape a child's context when considering the influences on their physical activity levels. Study 1 demonstrated that although there may be commonalities across environments, there are striking differences in the way that context influences what children consider barriers to their PA. This is of substantial concern because Study 2 demonstrates that perceiving barriers has a significant impact on the number of minutes of MVPA children are achieving.

Both studies emphasize the importance of factors at various socio-ecological levels for impacting children's physical activity. For example, age was consistently associated with perceiving barriers to activity. Study 2 found with a one-year increase in age, there was a 2.4 minute decrease in average daily MVPA in girls, and about a 5 minute decrease in boys. This is consistent with recent findings from Colley et al. (2017) that found that Canadian 6- to 11-year-olds achieved more weekly average MVPA than 12- to 17-year-olds did, based on the 2015 Canada Health Measures Survey. These findings highlight important implications for policy and programming related to children's health. There are multiple impacts of the integrated social and environmental contexts in relation to the varying intrapersonal environments for different groups of children. In practice, there is a need for multi-pronged approaches rather than current one-size-fits-all policies to best target how children perceive their experience of place in order to most effectively address the issue.

This thesis was one of the first studying Canadian children to consider a five-level spectrum approach to measure urbanicity rather than using an urban-rural dichotomy. This study therefore fills a gap in the literature highlighted by Sandercock, Angus, and Barton (2010), who drew attention to the need for a less simplistic comparison for classifying urbanicities, rather than a dichotomous definition of rural versus urban. This is especially important in the Canadian context where there is a wide heterogeneity of land uses and experiences related to both urbanicity and rurality. As previously stated in Chapter 2, in Ontario alone, there are five general types of rural (and northern or remote) communities: urban fringe communities, agriculture communities, cottage country

communities, Northern Ontario communities, and Indigenous communities (Markey, Lauzon, & Ryser, 2015). In Canada, this can further extend to include prairie and maritime communities (Walia & Liepert, 2012). Current policy definitions lump differing environments together under definitions that are based on population numbers and proximity to urban centers (Statistics Canada, 2015). This neglects to consider the history, geography, socio-economic status, and development trajectories of small communities.

The studies in this thesis utilized a spectrum approach to determine the impact of categorization of built form and population density at the physical environment level on children's perceptions of barriers to activity. The findings emphasized this approach as valuable for future work related to children's overall health when home location was linked as a determinate of behaviour. The present studies demonstrated that although children in rural areas were more likely to report neighbourhood environment barriers, as urbanicity increases, MVPA decreases, which is contrary to the literature (Joens-Matre et al., 2008; Springer et al., 2009, 2006). This work makes an important distinction between the needs of Canadian children in different contexts, highlighting the importance of considering variation in the experience of place; an important policy implication for health practitioners and physical activity researchers. Health practitioners must differentiate between perceived effects versus objective effects of barriers. This will help to determine whether there is a need to change perceptions of the environment or the need to make physical changes to the environment to impact physical activity behaviour, especially in rural areas (Ryan & Dzewaltowski, 2002). Given the results of the present studies, it is recommended that researchers continue to use a spectrum approach to define urbanicity. Doing so will help to better understand the differing experiences of residents in heterogeneous Canadian communities, based on the variation demonstrated in this thesis.

These findings further highlight the importance for considering the perspective of children when assessing the socio-ecological effects of the environment on physical activity. The literature review determined a need for researchers to include the voices of children in planning decisions (Lee et al., 2015). This is because, despite recognition of the influence environments have on children's physical activity, there is still a lack of

understanding regarding the relationship between the neighbourhood and children's perceptions and behaviours (Loebach & Gilliland, 2010). Furthermore, much of the research that exists uses parental reports on their children's behaviours (Ding et al., 2011), neglecting the potential differences between parents' and children's points of view. This work emphasizes the importance of gaining children's perspective as stakeholders in practice and policy, in order to effectively impact their environment for promoting health.

6.3 Limitations

The main limitations of the analysis methods in both Study 1 and 2 were the variables we could not able to account for. There are possible barriers and environmental factors we did not consider that could be mediating the relationships with perceptions or MVPA. This includes indicators for which we do not have specific measurements or data, such as socio-economic status, or objectively measured physical environment factors, including the built and natural environments. It is important to acknowledge that given the complexity of the influences on physical activity, there is no known direct cause and effect equation to increase physical activity behaviour. The research aimed to focus on more general factors to build on the gaps of previous research, as well as demonstrating the value on the use of an urbanicity spectrum. The limitation of Study 1 was addressed in Study 2 by considering the use of objective measure of MVPA specifically.

An additional limitation of Study 2 is that the use of structural equation modelling is not designed to uncover causal relationships, and measured values can change with the addition or deletion of a variable from the model. Despite these issues, the significant results of both studies provide starting points for future research to consider these issues more in depth, and the potential of additional confounding variables. These results demonstrate the value of considering children's environments from a variety of contexts, including a perspective of urbanicity, when attempting to influence their physical activity.

Accelerometers are considered a gold standard tool for their reliability in measuring children's physical activity patterns in this field of study (Borghese et al., 2017). The devices, however, are not without limitations. When using accelerometer measurements,

there are a number of considerations that may influence comparability of results. This includes the chosen data reduction procedures and cut-points, in addition to the model type (omnidirectional or uniaxial for example) (Borghese et al., 2017; Puyau et al., 2004). None the less, we used the validated techniques of Trost and colleagues (2000) to minimize such limitations, and included four days of monitoring (reliability of 0.80) (including weekdays and weekend days) as best practice to ensure rigorous results. Secondly, accelerometers worn on the wrist are more likely to record movements that may not be intended as physical activity, because they record any body movement of the attached location (Heil, 2013). To minimize this error, participants wore accelerometers attached to band around their waist. Additionally, there may have been a slight underestimation of the children's activity levels based on two limitations. Participants were required to remove the devices during water-based activities (such as swimming), to avoid damaging the equipment. Furthermore, accelerometers more accurately measure whole weight-bearing activities (such as running) and have difficulty measuring nonweight bearing activities (such as riding bikes), activities on inclines, or when lifting heavy objects (Heil, 2013). While the inability to understand these specific contexts of activity is a limitation, the validity of the Actical accelerometer outweighs the limitations when accurately measuring children's physical activity (Heil, 2013; Puyau et al., 2004).

One potential mechanism of behaviour not assessed in these studies that may have played an influential role in the relationship between perceptions and MVPA is self-efficacy. Self-efficacy is the belief an individual holds regarding his/her ability to engage in behaviours that lead to expected outcomes, such as physical activity (Ryan & Dzewaltowski, 2002). These beliefs will then influence if physical activity behaviour is adopted and maintained (Ryan & Dzewaltowski, 2002). Previous literature demonstrated this potential relationship. In a study of rural girls by Trost et al. (1997), self-efficacy for overcoming perceived barriers to physical activity was a significant predictor of vigorous physical activity. In a sample of older girls this finding was echoed by Motl et al. (2005) who found indirect effects of perceived barriers to self-reported physical activity were accounted for by factors of self-efficacy. Self-efficacy was not measured on the survey tool used in the STEAM project. Upcoming work produced by the HEAL will incorporate new survey questions to measure the effect of self-efficacy in work related to children's active travel, which if effective, can be applied in future work related to children's physical activity perceptions. Methods similar to those used in Study 2 could be beneficial to assess the relationship between the environment, perceiving barriers, physical activity levels, and the role children's beliefs in overcoming barriers. This could provide beneficial knowledge for program creation and policy change related to improving children's health.

A final limitation is the cross-sectional study design in both of the thesis studies. While the studies sites were stratified to be representative of the population in both Southwestern and Northwestern Ontario, the results will not necessarily represent patterns and behaviours of populations not included in the study. This points to an important finding and policy implication of the present thesis: there is a lack of generalizability related to the majority of evidence regarding relationships of children's physical activity and the environment, due to the cross-sectional nature of the field (Davison & Lawson, 2006; Ding et al., 2011). Utilizing blanket policies and taking a one-size-fits-all approach targeting health issues especially in rural areas (Canadian Rural Revitalization Foundation, 2015), does not accurately consider the variation in experiences of children in different places. This draws attention to the need for context-specific investigations and interventions.

6.4 Implications for Policy and Practice

This thesis aimed to explore how children's geographic contexts influence their perceptions of barriers to physical activity, and the influence of these factors on MVPA. To date there is a plethora of research surrounding the influence of the environment on children's physical activity (Davison & Lawson, 2006; Ding et al., 2011; Oliveira et al., 2014, Martins et al., 2017). Research has also indicated, that how children perceive their environment may be as important as the actual environment itself (Carroll-Scott et al., 2013; Hume et al., 2004). In order to understand how children's perceptions of their environments and the environments themselves influence physical activity, this thesis argues that a context-specific investigation is necessary. By focusing on population specific needs related to children's physical activity levels, there may be an opportunity to improve the efficiency of policy and practice for changing activity behaviour.

The main theme present across both studies is *context matters*. In order to understand how children engage in MVPA across Canada, future policy work needs to consider geographic variation in the determinants of behaviour (Orton et al., 2017). Health is formed in an individual's context. This is the circumstantial environment in which something takes place and including both the independent and the interplay of physical, social, cultural and structural environments. Policy makers must take into account the variety of factors that shape the formation of behaviours in daily interactions when considering the health and MVPA of children across varying environments. The studies demonstrate how the socio-ecological environments of each child affect how they perceive, and engage with their environments in pursuit of physical activity.

The *Report on the State of Public Health in Canada 2017* emphasized the need for research driven evaluations of the impact of community design on health, especially at population levels. This research has demonstrated that solutions to mitigating the low levels of MVPA in Canadian children requires community-level evaluations and approaches. For example, this thesis demonstrated the differences in experience between children in different urbanicities and how they experience MVPA as well as how they perceive barriers. This emphasized the need for attention outside of the urban context, and further attempted to fill a gap in the knowledge (Government of Canada, 2017).

Future policy work at the federal and provincial level, specifically targeting children such as the *Canadian Physical Activity Guidelines*, and elementary school curriculum must consider: 1) that context of physical activity environments varies across Canada; and 2) the importance of determining differences between subjective and objective experiences of the environment, moving away from one-size-fits-all approaches. One strategy could be through the incorporation of the spectrum approach, recognizing heterogeneous urbanicity and differing needs of individuals. This is crucial in the formation of policy and recommendations for practice, for stakeholders at downstream municipal, local health unit, and school levels. These downstream levels of policy can focus on creating best practices for assessing healthy public policy within their areas. For example, through health promotion initiatives, and evaluation tools (such as infrastructure audits), local officials can ensure environments are catering to the needs of children within the capacity of federal and provincial recommendations.

This research demonstrates that there are many factors influencing children's physical activity, to varying degrees. Individuals working directly with children to increase their physical activity levels such as parents, and practitioners (pediatricians, teachers, and recreation programmers), should attempt to understand children's perceptions barriers to activity in their environment. By evaluating the impact of these barriers on children's activity, these groups will have better insight into how to improve activity. One way by which this could be possible is through open conversations with children, acknowledging their position as stakeholders in decision-making. This could include asking children what factors in their current situation limit their physical activity, or preferences to be active, and looking for creative strategies and choices to overcome these barriers with the children.

6.4.1 Neighbourhood Environment Barriers to Physical Activity

Both studies demonstrated that perceptions of neighbourhood barriers impact physical activity. In Study 2, perceiving neighbourhood barriers lead to a significant decrease on MVPA in boys. Study 1 found all seven barrier questions were reported as significant by children in at least one socio-ecological variable group, however, they were often significantly related to urbanicity of the child's home location. One example of a barrier in this group was "There is no or not enough equipment or activities I like," which was reported as significant by 84% of participants. While the influence of accessible parks and playgrounds on physical activity behaviour has been well established in the literature (Davison & Lawson, 2006; Ding et al., 2011; McGrath et al., 2016; Oliveira et al., 2014), the present findings suggest safe, adequate, and age appropriate accessible activities or equipment are other aspects of availability that must be considered (Tucker et al., 2009). Although children may have playgrounds close to their home, they may not be interested in using these structures if they do not have access to what they consider age-appropriate and more challenging activities (Moore et al., 2013; Sallis et al., 1992). Boys for example have a tendency to navigate towards sports and open space activities (Farley, Meriwether, Baker, Rice, & Webber, 2008). This could be a beneficial starting point for health

promoters in areas which have cross over between rural and urban locations they are responsible for (such as Middlesex-London) to investigate the ways rural boys achieve more MVPA minutes than their counterparts despite encountering barriers. These findings could aid in determining disparities influencing activity minutes based on home location of boys, despite the increased number of opportunities in increasingly urban settings, and offer an opportunity for increaseing programming.

The only neighbourhood barrier reported as significantly influential by children in rural and urban small-towns was "there are not a lot of trees along the streets in my neighbourhood." The children in this group represent 25% of participants, when measured by urbanicity of their home location. The literature review in Chapter 2 discussed evidence regarding the positive associations of neighbourhood aesthetics and promoting physical activity. This includes features such as trees along the streets. Some researchers suggest it is not the presence but the absence of aesthetically pleasing landscapes as influential to physical activity (Davison & Lawson, 2006; Jago et al., 2005; Mota, Almeida, Santos, & Ribeiro, 2005) a statement supported by the findings in both studies of this thesis. We can speculate, this may be related to children's association of poor neighbourhood aesthetics with a lack of safety, or lack of enjoyment (Loebach & Gilliland, 2010). Nonetheless, city planners and researchers should work together to consider the importance of children's perceptions in streetscapes, and determine best practices for ensuring environments are supportive to their physical activity behaviour.

While neighbourhood features have been associated with physical activity in the past, it is important to acknowledge the mixed results and measurement tools. In the systematic review by Ding et al., (2011) they found 878 comparisons of relationships between neighbourhood environment features and physical activity (including our barriers), of which the number of significant associations was higher when physical activity was measured by report (two thirds of studies), versus objectively measured. Ding et al. (2011) suggest that when a combination of objective physical activity and perception measured environment features were used, there is an inconsistency in the relationships with physical activity across papers. This includes walking/biking facilities and neighbourhood incivilities in children, and access to parks or recreation facilities in

adolescents. The conclusions by Ding et al. (2011) are supported by the differences found between results related to gender in Study 2 of this thesis. Despite the lack of significance in girls, the results demonstrate an important finding for practice. Children's perceptions of their environments are still an important factor in influencing their own reports of physical activity behaviour. While perceptions may not be linked with objective physical activity directly, there may be a role for self-efficacy and the belief children have about engaging in physical activity (Ryan & Dzewaltowski, 2002). Research related to maximizing local recreation programming and infrastructure must seek out children's perspectives on how they view their environments because assessing perceptions is an important tactic for quantifying how children experience the context of daily living (Davison & Lawson, 2006).

6.4.2 Social Barriers to Physical Activity

Both studies demonstrated that perceptions of social barriers have an impact on physical activity. In Study 2, perceiving social barriers had a significant decrease on MVPA in girls. Study 1 found four social barriers that children significantly reported as influential to their MVPA. One barrier in Study 1 that was reported by girls as significantly influential was "I have nobody to go with to parks/playgrounds in my neighbourhood." As discussed in Study 2, regardless of urbanicity, girls reported this barrier to activity. Social factors and barriers influencing physical activity, especially in girls, are well studied in the literature (Bocarro et al., 2015; Dwyer et al., 2006; Pawlowski et al., 2014; Sallis et al., 1999; Spencer et al., 2015), and the present result has been supported by qualitative research asking girls about the barriers they face (Pawlowski et al., 2014). Given the ongoing disparities in levels of activity relative to gender, health promotion efforts must focus on alleviating social barriers in structured activity opportunities in and out of school time, to increase MVPA in girls (Telford et al., 2016).

A 2012 review (Salvy, de la Haye, Bowker, & Hermans) found research surrounding the social relationships of children has been mostly based on parent report, and tends to focus on related aspects of the social environment such as safety and neighbourhood social cohesion. Three additional barriers related to neighbourhood social cohesion were significantly reported in Study 1. This included "There are no other kids to play with at

parks/playgrounds in my neighbourhood" (reported by children in low income families compared to middle income), "I [do not] know a lot of people in my neighbourhood" (reported by visible minority children), and "There are too many people/it feels too crowded at parks/playgrounds in my neighbourhood" (reported by children who live in one home, 84% of children). Neighbourhood social cohesion is the extent of the connectedness and solidarity in a community, and is characterized by two main features: 1) absence of latent social conflict, and 2) present of strong social bonds (Kawachi & Berkman, 2000). As explained by Utter, Denny, Robinson, Ameratunga, and Milfont (2011), when children perceive their communities as socially cohesive, there has been evidence of positive associations with their physical activity in the literature. This is influenced by a community's shared goals, collective trust and norms. Social cohesion is an area from which stakeholders in local contexts could draw on, to strengthen community ties, or to utilize already existing strong relationships to increase children's MVPA directly or indirectly. This includes local officials, parents, and health promoters. Additional research should seek out the specific mechanisms by which companionship plays a role in children's activity patterns specifically in relation to their geographic contexts.

6.4.3 Safety Barriers to Physical Activity

The literature review in Chapter 2 provided a background regarding the mixed results of the implications of neighbourhood safety on physical activity. Researchers have reported both negative associations and non-significant relationships between both subjective and objective measures of a lack of safety and children's physical activity. This thesis also found mixed results on the influence of safety. In Study 2, there were no significant relationships between barriers of safety and children's MVPA. However, in Study 1, differentiated by the independent variables, groups of children perceived all variables as important. The majority of children (based on maternal employment) and girls reported "Most drivers go too fast while driving in our neighbourhood." Older children, children in low-income families, and children in urban and suburban large-cities reported "There is so much traffic on streets near my home that it's difficult/unpleasant to bike or play on the street." Girls, older children and those in urban large-cities reported "I am worried"

about being or walking by myself in my neighbourhood and local streets because I am worried about being taken or hurt by a stranger" as a significant barrier. Children in suburban large cities reported "There is a lot of crime in my neighbourhood (ex: strangers, gangs, drugs)" compared to rural small town and rural children. Despite their differing results, these papers lead to an important implication for health practitioners and policy makers.

As suggested by Beets and Foley (2008), it may not be the presence of actual environmental characteristics that directly affect safety influencing decisions to participate in physical activity, but rather the perceptions of neighbourhood characteristics that promote safety perhaps are more influential. By identifying the variety of perceptions children feel are important to their physical activity, whether or not they actually influenced MVPA objectively, this could provide insight into how children view opportunities or barriers in their communities. This may be an opportunity for local officials and health promoters to draw on strengths of their community to encourage activity in their neighbourhood. Drawing on strengths of the community such as social cohesion and working towards a common goal may be an opportunity to change their perceptions and further promote physical activity (Loebach & Gilliland, 2010).

6.5 Recommendations for Future Research

Findings from Study 1 emphasize the need for research on children's health to consider environmental context as more than just the physical surroundings. When taking a context specific approach to understanding the barriers children perceive as influential to their physical activity, researchers must consider the interplay of the intrapersonal, interpersonal, and physical environments. Based on the findings of this research, this could include investigation into the impact of parental employment on children's physical activity and the formation of environmental perceptions. Another example is understanding how children perceive crime in their communities relative to the objective crime rates, as a mechanism by which to mitigate this perceived barrier. This research in the geography discipline must extend to include indicators we did not have specific measures for such as socio-economic status and park access, or factors that could be considered at further levels of the socio-ecological model such as public policy's effect on perceptions and behaviours. Researchers in the health sciences and psychology field should focus on understanding how perceptions of physical activity are influential in behaviour, the role of self-efficacy, and how changing perceptions may be beneficial to improving behaviour.

Findings from Study 2 support the role of both children's perceptions and varying environments play in meeting Canada's daily physical activity recommendations. Future work should continue to consider the contexts that children live as independent from previous findings, as results across settings cannot be generalized to create one-size-fitsall MVPA policy solutions. More research is needed to understand the mechanisms by which barriers and the environment impact MVPA. Specifically based on the results of the present studies, more work is needed to 1) explore how boys and girls are affected by their perceptions of barriers differently, in an attempt to close the gap between MVPA achievements; 2) continue to consider the relationships between perception formation, the environmental context, and the resulting effects on physical activity. Future work should consider using similar methods to study additional levels of activity including light and total physical activity, and sedentary behaviour. Furthermore, additional work should consider longitudinal and qualitative research approaches, to investigate intermediating mechanisms of the present findings.

Findings from both studies emphasize the importance and need for incorporating children as stakeholders in research, practice and policy that seeks to understand or will have an impact on their health. There has been little-to-no research explicitly examining children's perceptions of barriers of physical activity in outdoor spaces in the various rural communities of Canada (Lee et al., 2015). Researchers often exclude the voice of children in their studies, assuming children are unable to discuss their own needs (Loebach & Gilliland, 2010). In order to ensure conditions support physical activity, children must be valued as equal stakeholders in the research process, knowledge translation, and implementation of public policy (Faulkner, White, Riazi, Latimer-Cheung, & Tremblay, 2016).

One strategy by which this could be achieved and a final recommendation is through a focus on incorporating qualitative research through simultaneous methodological

triangulation (Morse, 1991). While surveys are often used to collect large scale data allowing for a degree of standardization and comparison of data across the study sites, qualitative methods allow for an understanding of *why* children perceive these barriers as influential. By incorporating qualitative methods such as focus groups into an integrated research design of a quantitative survey component, the quality of the resulting analysis can be more confidently accepted (Wolff, Knodel, & Sittitrai, 1993). This is because the strengths and limitations inherent to each method complement one another. Survey results will be representative and comparable between the different communities studied. Focus groups will help to provide depth and explanatory themes and categories that can explain results of the survey, specifically within the communities studied, which is important for policy development.

6.6 Conclusions and Implications

The purpose of this research was to investigate geographic variation in how children's physical activity behaviour is influenced by the experience of barriers in their everyday environments. By examining the effects of perceptions and environments on children's physical activity levels, we can begin to suggest strategies for future research, policy, and practice to help alleviate what children consider barriers to their activity. This research highlights that determinants describing each child within the socio-ecological model will impact how they perceive environmental barriers to their physical activity. These perceptions demonstrate an influence on the likelihood of being active. These papers highlight the need for context specific approaches to increasing physical activity when attempting to address disparities in children's behaviour. Both studies in this thesis highlight that children's geographies have an impact on their physical activity behaviour. This includes differences related to categorization of built form and population density, previously a gap in the literature.

Less than 35% of Canadian children are achieving the standard recommendation of physical activity each week. This number has remained relatively consistent since 2007 (Colley et al., 2017). Health practitioners, city officials, and researchers must look at tailoring approaches to address this issue given the heterogeneity of the Canadian context. The findings from this thesis will support development of policy, programs, and practices

that incorporate, encourage, and facilitate the use of children's perspectives to improve physical activity levels in their everyday environments.

6.7 References

- Babey, S. H., Tan, D., Wolstein, J., & Diamant, A. L. (2015). Neighborhood, family and individual characteristics related to adolescent park-based physical activity. *Preventive medicine*, 76, 31-36. https://doi.org/10.1016/j.ypmed.2015.04.001
- Beets, M. W., & Foley, J. T. (2008). Association of Father Involvement and Neighborhood Quality with Kindergartners' Physical Activity: A Multilevel Structural Equation Model. *American Journal of Health Promotion*, 22(3), 195–203. https://doi.org/10.4278/ajhp.22.3.195
- Bocarro, J. N., Floyd, M. F., Smith, W. R., Edwards, M. B., Schultz, C. L., Baran, P., ... Suau, L. J. (2015). Social and environmental factors related to boys' and girls' parkbased physical activity. *Preventing Chronic Disease*, *12*, E97. https://doi.org/10.5888/pcd12.140532
- Borghese, M. M., Tremblay, M. S., LeBlanc, A. G., Leduc, G., Boyer, C., & Chaput, J. P. (2017). Comparison of ActiGraph GT3X+ and Actical accelerometer data in 9–11year-old Canadian children. *Journal of Sports Sciences*, 1–8. https://doi.org/10.1080/02640414.2016.1175653
- Canadian Rural Revitalization Foundation. (2015). *State of Rural Canada Report*. Retrieved from sorc.crrf.ca/wp-content/uploads/2015/09/SORC2015.pdf
- Carroll-Scott, A., Gilstad-Hayden, K., Rosenthal, L., Peters, S. M., McCaslin, C., Joyce, R., & Ickovics, J. R. (2013). Disentangling neighborhood contextual associations with child body mass index, diet, and physical activity: The role of built, socioeconomic, and social environments. *Social Science & Medicine*, 95, 106–114. https://doi.org/10.1016/J.SOCSCIMED.2013.04.003
- Colley, R. C., Carson, V., Garriguet, D., Janssen, I., Roberts, K. C., & Tremblay, M. S.
 (2017). Physical activity of Canadian children and youth. *Health Reports*, 28(10), 8–16. https://doi.org/October 2017

- Davison, K., & Lawson, C. T. (2006). Do attributes in the physical environment influence children's physical activity? A review of the literature. *International Journal of Behavioral Nutrition and Physical Activity*, 3(1), 19. https://doi.org/10.1186/1479-5868-3-19
- Ding, D., Sallis, J. F., Kerr, J., Lee, S., & Rosenberg, D. E. (2011). Neighborhood Environment and Physical Activity Among Youth. *American Journal of Preventive Medicine*, 41(4), 442–455. https://doi.org/10.1016/j.amepre.2011.06.036
- Dwyer, J. J. M., Allison, K. R., Goldenberg, E. R., Fein, A. J., Yoshida, K. K., & Boutilier, M. A. (2006). Adolescent girls' perceived barriers to participation in physical activity. *Adolescence*, 41(161), 75–89.
- Farley, T. A., Meriwether, R. A., Baker, E. T., Rice, J. C., & Webber, L. S. (2008).
 Where Do the Children Play? The Influence of Playground Equipment on Physical Activity of Children in Free Play. *Journal of Physical Activity and Health*, 5(2), 319–331. https://doi.org/10.1123/jpah.5.2.319
- Faulkner, G., White, L., Riazi, N., Latimer-Cheung, A. E., & Tremblay, M. S. (2016). Canadian 24-Hour Movement Guidelines for Children and Youth: Exploring the perceptions of stakeholders regarding their acceptability, barriers to uptake, and dissemination. *Applied Physiology, Nutrition, and Metabolism, 41*(6 (Suppl. 3)), S303–S310. https://doi.org/10.1139/apnm-2016-0100
- Government of Canada (2017). *The Chief Public Health Officer's Report on the State of Public Health in Canada 2017 – Designing Healthy Living*. Ottawa. Retrieved from https://www.canada.ca/content/dam/phacaspc/documents/services/publications/chief-public-health-officer-reports-statepublic-health-canada/2017-designing-healthy-living/2017-designing-healthy-livingeng.pdf
- Heil, D. P. (2013). Predicting Activity Energy Expenditure Using the Actical® Activity Monitor. *Research Quarterly for Exercise and Sport*, 77(1), 64–80. https://doi.org/10.1080/02701367.2006.10599333

- Hume, C., Salmon, J., & Ball, K. (2004). Children's perceptions of their home and neighborhood environments, and their association with objectively measured physical activity: a qualitative and quantitative study. *Health education research*, 20(1), 1-13. https://doi.org/10.1093/her/cyg095
- Jago, R., Baranowski, T., Zakeri, I., & Harris, M. (2005). Observed Environmental Features and the Physical Activity of Adolescent Males. *Am J Prev Med American Journal of Preventive Medicine*, 29(2), 98–104. https://doi.org/10.1016/j.amepre.2005.04.002
- Joens-Matre, R. R., Welk, G. J., Calabro, M. A., Russell, D. W., Nicklay, E., & Hensley, L. D. (2008). Rural–Urban Differences in Physical Activity, Physical Fitness, and Overweight Prevalence of Children. *The Journal of Rural Health*, 24(1), 49–54. https://doi.org/10.1111/j.1748-0361.2008.00136.x
- Kawachi, I., & Berkman, L. (2000). *Social Epidemiology*. New York: Oxford University Press.
- Lee, H., Tamminen, K. A., Clark, A. M., Slater, L., Spence, J. C., & Holt, N. L. (2015). A meta-study of qualitative research examining determinants of children's independent active free play. *International Journal of Behavioral Nutrition and Physical Activity*, 12(1), 5. https://doi.org/10.1186/s12966-015-0165-9
- Loebach, J., & Gilliland, J. (2010). Child-Led Tours to Uncover Children's Perceptions and Use of Neighborhood Environments. Source: Children, Youth and Environments Collected Papers University of Cincinnati Children, Youth and Environments, 20(201), 52–90. https://doi.org/10.7721/chilyoutenvi.20.1.0052
- Markey, B., Lauzon, G., & Ryser, M. (Eds.). (2015). Rural restructuring and its impact on community recreation opportunities. Annals of Leisure Research (Vol. 18). Brandon, MB. https://doi.org/10.1080/11745398.2014.980834
- Martins, J., Marques, A., Peralta, M., Palmeira, A., & Carreiro da Costa, F. (2017). Correlates of physical activity in young people: A narrative review of reviews.

Implications for physical education based on a socio-ecological approach. *Retos, Nuevas tendencias en Educación Física, Deporte y Recreación, 31,* 292-299.

- McGrath, L. J., Hinckson, E. A., Hopkins, W. G., Mavoa, S., Witten, K., & Schofield, G. (2016). Associations Between the Neighborhood Environment and Moderate-to-Vigorous Walking in New Zealand Children: Findings from the URBAN Study. *Sports Medicine*, 46(7), 1003–1017. https://doi.org/10.1007/s40279-016-0533-x
- Moore, J. B., Brinkley, J., Crawford, T. W., Evenson, K. R., & Brownson, R. C. (2013). Association of the built environment with physical activity and adiposity in rural and urban youth. *Preventive Medicine*, 56, 145–148. https://doi.org/10.1016/j.ypmed.2012.11.019
- Morse, J. (1991). Approaches to Qualitative-Quantitative Methodological Triangulation. *Nursing Research*, 40(2), 120–123.
- Mota, J., Almeida, M., Santos, P., & Ribeiro, C. (2005). Perceived Neighborhood
 Environments and physical activity in adolescents. *Preventive medicine*, 41(5-6), 834-836. https://doi.org/10.1016/j.ypmed.2005.07.012
- Motl, R. W., Dishman, R. K., Ward, D. S., Saunders, R. P., Dowda, M., Felton, G., & Pate, R. R. (2005). Perceived physical environment and physical activity across one year among adolescent girls: self-efficacy as a possible mediator? *The Journal of Adolescent Health : Official Publication of the Society for Adolescent Medicine*, 37(5), 403–408. https://doi.org/10.1016/j.jadohealth.2004.10.004
- Oliveira, A. F., Moreira, C., Abreu, S., Mota, J., & Santos, R. (2014). Environmental determinants of physical activity in children: A systematic review. Archives of Exercise in Health and Disease, 4(2), 254–261. https://doi.org/10.5628/aehd.v4i2.158
- Orton, L., Halliday, E., Collins, M., Egan, M., Lewis, S., Ponsford, R., ... Popay, J. (2017). Putting context centre stage: evidence from a systems evaluation of an area based empowerment initiative in England. *Critical Public Health*, 27(4), 477–489.

- Pawlowski, C. S., Tjørnhøj-Thomsen, T., Schipperijn, J., & Troelsen, J. (2014). Barriers for recess physical activity: a gender specific qualitative focus group exploration. *BMC Public Health*, 14(1), 639. https://doi.org/10.1186/1471-2458-14-639
- Puyau, M. R., Adolph, A. L., Vohra, F. A., Zakeri, I., & Butte, N. F. (2004). Prediction of Activity Energy Expenditure Using Accelerometers in Children. *Med. Sci. Sports Exerc*, 36(9), 1625–1631. https://doi.org/10.1249/01.MSS.0000139898.30804.60
- Statistics Canada (2015). Rural Area. Retrieved October 13, 2017, from https://www12.statcan.gc.ca/census-recensement/2011/ref/dict/98-301-X2011001eng.pdf
- Ryan, G. J., & Dzewaltowski, D. A. (2002). Comparing the Relationships between
 Different Types of Self-Efficacy and Physical Activity in Youth. *Health Education*& *Behavior*, 29(4), 491–504. https://doi.org/10.1177/109019810202900408
- Sallis, J. F., Prochaska, J. J., Taylor, W. C., Hill, J. O., & Geraci, J. C. (1999). Correlates of physical activity in a national sample of girls and boys in Grades 4 through 12. *Health Psychology*, 18(4), 410–415. https://doi.org/10.1037/0278-6133.18.4.410
- Sallis, J. F., Simons-Morton, B. G., Epstein, L. H., Faucette, N., Iannotti, R. J., Killen, J. D., ... Taylor, W. C. (1992). Determinants of physical activity and intervention in youth. *Medicine and Science in Sports and Excercise*, 24(6), S248–S258.
- Salvy, S.J., de la Haye, K., Bowker, J. C., & Hermans, R. C. J. (2012). Influence of peers and friends on children's and adolescents' eating and activity behaviors. *Physiology & Behavior*, *106*(3), 369–378. https://doi.org/10.1016/J.PHYSBEH.2012.03.022
- Sandercock, G., Angus, C., & Barton, J. (2010). Physical activity levels of children living in different built environments. *Preventive Medicine*, 50(4), 193-198. https://doi.org/10.1016/j.ypmed.2010.01.005

Smith, A. L., Troped, P. J., McDonough, M. H., & DeFreese, J. D. (2015). Youth

perceptions of how neighborhood physical environment and peers affect physical activity: a focus group study. *International Journal of Behavioral Nutrition and Physical Activity*, *12*(1), 80. https://doi.org/10.1186/s12966-015-0246-9

- Spencer, R. A., Rehman, L., & Kirk, S. (2015). Understanding gender norms, nutrition, and physical activity in adolescent girls: a scoping review. *International Journal of Behavioral Nutrition and Physical Activity*, 12(1), 6. https://doi.org/10.1186/s12966-015-0166-8
- Springer, A. E., Hoelscher, D. M., Castrucci, B., Perez, A., & Kelder, S. H. (2009).
 Prevalence of physical activity and sedentary behaviors by metropolitan status in 4th-, 8th-, and 11th-grade students in Texas, 2004-2005. *Preventing Chronic Disease*, 6(1), A21.
- Springer, A. E., Hoelscher, D. M., & Kelder, S. H. (2006). Prevalence of Physical Activity and Sedentary Behaviors in US High School Students by Metropolitan Status and Geographic Region. *Journal of Physical Activity and Health*, 3(4), 365– 380. https://doi.org/10.1123/jpah.3.4.365
- Statistics Canada (2015). Rural Area. Retrieved October 13, 2017, from https://www12.statcan.gc.ca/census-recensement/2011/ref/dict/98-301-X2011001eng.pdf
- Telford, R. M., Telford, R. D., Olive, L. S., Cochrane, T., & Davey, R. (2016). Why Are girls less physically active than boys? findings from the LOOK longitudinal study. *PloS One*, 11(3), e0150041. https://doi.org/10.1371/journal.pone.0150041
- Trost, S. G., Pate, R. R., Freedson, P. S., Sallis, J. F., & Taylor, W. C. (2000). Using objective physical activity measures with youth: How many days of monitoring are needed? *Sci. Sports Exerc*, 32(2), 426–431.
- Trost, S. G., Pate, R. R., Saunders, R., Ward, D. S., Dowda, M., & Felton, G. (1997). A Prospective Study of the Determinants of Physical Activity in Rural Fifth-Grade Children. *Preventive Medicine*, 26(2), 257–263.

https://doi.org/10.1006/PMED.1996.0137

- Tucker, P., Irwin, J. D., Gilliland, J., He, M., Larsen, K., & Hess, P. (2009). Environmental influences on physical activity levels in youth. *Health & Place*, 15(1), 357–363. https://doi.org/10.1016/J.HEALTHPLACE.2008.07.001
- Utter, J., Denny, S., Robinson, E., Ameratunga, S., & Milfont, T. L. (2011). Social and physical contexts of schools and neighborhoods: associations with physical activity among young people in New Zealand. *American Journal of Public Health*, 101(9), 1690–1695. https://doi.org/10.2105/AJPH.2011.300171
- Walia, S., & Liepert, B. (2012). Perceived facilitators and barriers to physical activity for rural youth: an exploratory study using photovoice. *Rural and Remote Health*, 12(1842).
- Wolff, B., Knodel, J., & Sittitrai, W. (1993). Focus Groups and Surveys as Complementary Research Methods. In D. Morgan (Ed.), *Focus groups and surveys as complementary research methods: a case example.* (pp. 118–136). Thousand Oaks: SAGE. https://doi.org/10.4135/9781483349008
Appendices

Appendix A: Literature Review Documents with Corresponding Number Assigned (4 pages)

#	Year	Author	Title
1	2010	Sandercock, Angus, Barton	Physical activity levels of children living
2	2006	Davison, Lawson	Do attributes in the physical environment influence children's physical activity? A review of the literature
3	2010	Abraham, Sommerhalder, Abel	Landscape and well-being: a scoping study on the health-promoting impact of outdoor environments
4	2004	Lofshult	Getting Kids Active
5	2008	Mota, Ribeiro, Santos	Obese girls differences in neighbourhood perceptions, screen time and socioeconomic status according to level of physical activity
6	2004	Hume, Salmon, Ball	Children's perceptions of the home and neighborhood environments, and their associate with objectively measured physical activity: a qualitative and quantitative study
7	2016	Heidelberger, Smith	Low-Income, Urban Children's Perspectives on Physical Activity: A Photovoice Project
8	2015	Lee, Tamminen, Clark, Slater, Spence, Holt	A meta-study of qualitative research examining determinants of children's independent active free play
9	2016	Holt, Neely, Spence, Carson, Pynn, BoydRobinson	An intergenerational study of perceptions of changes in active free play among families from rural areas of Western Canada
10	2016	Faulkner, White, Riazi, Latimer-Cheung, Tremblay	Canadian 24-Hour Movement Guidelines for Children and Youth: Exploring the perceptions of stakeholders regarding their acceptability, barriers to uptake, and dissemination
11	2009	Sherar , Gyurcsik, Humbert, Dyck, Fowler-Kerry, Baxter-Jones	Activity and Barriers in Girls(8-16 yr) Based on Grade and Maturity Status

12	2010	Loebach & Gilliland	Child-Led Tours to Uncover Children's
			Perceptions and Use of Neighborhood
			Environments
13	2002	Humpel, Owen, Leslie	Environmental factors associated with
			adult's participation in Physical Activity:
			a review
14	2005	Mota, Almeida, Santos,	Perceived Neighborhood Environments
		Riberio	and physical activity in adolescents
15	2006	Galloway	Obesity Rates Among Rural Ontario
		, j	Schoolchildren
16	2008	Bruner, Lawson, Pickett,	Rural Canadian adolescents are more
		Boyce, Janssen	likely to be obese compared with urban
			adolescents
17	1999	Pate, Trost, DowdaOtt,	Tracking of Physical Activity, Physical
		Ward, Saunders, Felton	Inactivity, and Health-Related Physical
			Fitness in Rural Youth
18	2004	Bauman	Updating the evidence that physical
			activity is good for health: an
			epidemiological review 2000-2003
19	2006	Warburton, Nicol, Bredin	Health benefits of physical activity: the
			evidence
20	2008	Grow, Saelens, Kerr,	Where are youth activeÉ Roles of
		Durant, Norman, Sallis	proximity, active transport and built
			environment
21	2015	Babey, Tan, Wolstein,	Neighbourhood, family and individual
		Diamant	characteristics related to adolescent park-
			based physical activity
22	2010	Aarts, Wendel-Vos, van	Environmental Determinants of Outdoor
		Oers, van de Goor, Schuit	Play in Children A Large-Scale Cross-
			Sectional Study
23	2005	Burdette, Whitaker	Resurrecting Free Play in Young
			Children: looking beyond fitness and
			fatness to attention, affiliation, and affect
24	2017	Larouche, Garriguet,	Outdoor time, physical activity and
		Tremblay	sedentary time among young children:
			The 2012–2013 Canadian Health
			Measures Survey
25	2015	Grey, Gibbons, Larouche,	What Is the Relationship between
		Sandseter, Bienenstock,	Outdoor Time and Physical Activity,
		BrussoniPower	Sedentary Behaviour, and Physical
			Fitness in Children? A Systematic
			Review
26	2015	Tremblay, Gray, Babcock,	Position statement on Active Outdoor
		Barnes, Bradstreet,	Play
		CarrHerrington	

27	2016	Larouche, Garriguet,	Outdoor time, physical activity,
		Gunnell, Goldfield,	sedentary time, and health
		Tremblay	indicators at ages 7 to 14: 2012/2013
			Canadian Health
			Measures Survey
28	2012	Reed, Hooker	Where are youth physically active - a
			descriptive examination of 45 parks in a
			southeastern community
29	2011	Kacztnski, Walch, Stanis,	Variations in Observed Park Physical
		Besenyi	Activity Intensity Level by Gender,
			Race, and Age: Individual and Joint
			Effects
30	2016	Clark, Scott	Barriers to Walking: An invesitigation of
			adults in Hamilton (Ontario Canada)
31	2016	Mitchell, Clark, Gilliland	Built Environment Influences of
			Children's PhysicalActivity: Examining
			Differences by NeighbourhoodSize and
			Sex
32	2016	Ou, Levy, Peters,	A walk in the park: influence of urban
		Bongiovanni, Garcia-Soto,	parks and community violence on
		Medina, Scammell	physical activity in Chelsea MA
33	2014	Fitzpatrick	Examining Children's Perceptions and
			Use of Their Neighbourhood Built
			Environments: A Novel Participatory
			Mapping Approach
34	2009	Ries Voorhees, Roche,	A Quantitative Examination of Park
		Gittelsohn, Yan, Astone	Characteristics Related to Park Use and
			Physical Activity Among Urban Youth
35	2013	Alexander, Huber, Piper,	The association between recreational
		Tanner	parks, facilities and childhood obesity: A
			cross-sectional study of the 2007
			National Survey of Children's Health
36	2012	Gililand, Rangel, Healy,	Linking Childhood Obesity to the Built
		Tucker, Leoback,	Environment: A Multi-level Analysis of
		HessWilk	Home and School Neighbourhood
			Factors Associated With Body Mass
			Index
37	2010	Carson, Kuhle, Spence,	Parents' Perception of Neighbourhood
		Veugelers	Environment as a Determinant of Screen
			Time, Physical Activity and Active
			Transport
38	2011	Larson, Green, Cordell	Children's Time Outdoors: Results and
			Implications of the
			National Kids Survey

39	2017	Nykiforuk, Atkey, Brown,	Promotion of Physical Activity in Rural,
		Caldwell, Galloway,	Remote, and Northern Settings: A
		GillilandRaine	Canadian Call to Action
40	2013	Participaction	2013 Active Healthy Kids Canada Report
		-	Card on Physical Activity for Children
			and Youth
41	2015	Canadian Rural	2015 State of Rural Canada report
		Revitalization Foundation	L
42	2012	Walia, Liepert	Perceived facilitators and barriers to
			physical activity for rural youth: an
			exploratory study using photovoice.
43	2006	DesMeules, Pong, Lagacé,	How Healthy Are Rural Canadians? An
		Heng, Manuel,	Assessment of Their Health Status and
		PitbladoKoren	Health Determinants
44	2009	Yousefian, Ziller, Swartz,	Active Living for Rural Youth:
		Hartley	Addressing Physical Inactivity in Rural
			Communities
45	2013	Moore, Brinkley, Crawford,	Association of the built environment with
		Evenson, Brownson	physical activity and adiposity in rural
			and urban youth
46	1997	Trost, Pate, Saunders,	A prospective study of the determinants
		Ward, Dowda, Felton	of physical activity in rural fifth-grade
			children
47	2000	Matthews, Taylor,	Growing-up in the countryside: children
		Sherwood, Tucker, Limb	and the rural idyll.
48	2011	Public Health Agency of	Physical Activity
		Canada	
49	2014	Smith, Troped, McDonouh,	Youth perceptions of how neighborhood
		DeFreese	physical environment and peers affect
1			physical activity: a focus group study

Appendix B: Literature Review Results Demonstrating Similarities and Differences
Children's Perceptions of Barriers and Facilitators to Physical Activity between the
Urban and Rural Children (2 pages)

			Intr	apersona	l			
	Perceived		Preference for		Perceived Aesthetics		Preference	
	Saf	ety	Wea	ther	/Quality		for	
		-			-		Activities	
	+	-/NS	+	-/NS	+	-/NS	+	-
								/NS
Facilitators	23,				14, 3,		8, 20,	
n= 12	44, 3,				12, 23		42	
	12,							
	20, 21							
Barriers	1, 9,	37	7, 11,		14		38,	
n= 19	12,		38, 39				42	
	20,							
	23,							
	36,							
	38,							
	44,							
	39, 6,							
	21							
			Interperso	nal				
	Social	Factors	Access	sibility	Par	rents		
			(transpo	rtation)	Perceptions/Rules			
	+	-/NS	+	-/NS	+	-/NS		
Facilitators	14,		42, 20,		42			
n= 14	23,		36, 37,					
	32,							
	42, 5,							
	6, 12,							
	22							
Barriers	14,		4, 38,		4, 9, 12,			
n= 17	32,		42, 44,		42, 44,			
	42, 39		39, 32		8, 32		J	
	Institutional							
Socio-			Planned/S	tructured				
	econ	omic	Activ	vities				
	Circum	stances						
	+	-/NS	+	-/NS				
Facilitators	42		42, 44, 5					
n= 4								

Barriers	12,		2, 12,			
n= 7	42, 44		42, 44			
			Communi	ity		
	Facilit	ties &	Ci	City		proximity/
	Amer	nities	Planning	/Design	distance/a	vailability)
	+	-/NS	+	-/NS	+	-/NS
Facilitators	14,		3, 37, 42		20, 21,	32,
n= 13	44, 3,				42	
	5, 6,					
	20					
Barriers	12,		9, 44,		2, 8, 4,	32
n= 22	14,		39, 32,		11, 12,	
	20,		36		20, 38,	
	44,				42, 39,	
	39, 8				32, 44	

Appendix C: Research Ethics Approval Forms for Use of Human Participants STEAM Southwestern Ontario (Redacted)



This is an official document. Please retain the original in your files.

The University of Western Ontario

Office of Research Ethics Support Services Building Room 5150 • London, Ontario • CANADA - N6G 1G9 PH: 519-661-3036 • F: 519-850-2466 • ethics@uwo.ca • www.uwo.ca/research/ethics

Appendix D: Research Ethics Approval Forms for Use of Human Participants STEAM Northwestern Ontario (Redacted)

Research Ethics

Western Research

Western University Non-Medical Research Ethics Board NMREB Annual Continuing Ethics Approval Notice

Date: July 26, 2017 Principal Investigator: Dr. Jason Gilliland Department & Institution: Social Science/Geography,Western University

NMREB File Number: 108029 Study Title: Spatial Temporal Environment and Activity Monitoring Project: Understanding the environmental influences of health in rural and remote communities

NMREB Renewal Due Date & NMREB Expiry Date: Renewal Due -2018/07/31 Expiry Date -2018/08/16

The Western University Non-Medical Research Ethics Board (NMREB) has reviewed the Continuing Ethics Review (CER) form and is re-issuing approval for the above noted study.

The Western University NMREB operates in compliance with the Tri-Council Policy Statement Ethical Conduct for Research Involving Humans (TCPS2), Part 4 of the Natural Health Product Regulations, the Ontario Freedom of Information and Protection of Privacy Act (FIPPA, 1990), the Ontario Personal Health Information Protection Act (PHIPA, 2004), and the applicable laws and regulations of Ontario.

who are named as Investigators in research studies do not participate in discussions related to, nor vote on such studies when they are presented to the REB.

The NMREB is registered with the U.S. Department of Health & Human Services under the IRB registration

EO: Erika Basile___Grace Kelly___Katelyn Harris___Nicola Morphet___Karen Gopaul___Patricia Sargeant ___

Western University, Research, Support Services Bldg., Rm. 5150 London, ON, Canada N6G1G9 t. 519.661.3036 t. 519.850.2466 www.uwo.ca/research/ethics

Appendix E Research Ethics Letter of Information for Parents (3 pages) and Research Ethics Parent Consent Form (1 page)



Examining the Influence of the Neighbourhood Environment on Children's Health and Well-Being

Dear parent or guardian,

We would like to invite you and your child to participate in a study aimed at understanding how the neighbourhood environment around your child's school affects his or her health. The study involves grade 5, 6 and 7 classes at elementary schools across South Western Ontario.

What is being studied?

Our research team is studying the various places or facilities in their neighbourhood that children use (or intentionally don't use) on a regular basis for recreational or physical activities, including the way they travel to these places – for example, how they travel to and from school each day. We are also interested in looking at some of their eating patterns, especially the locations in their neighbourhoods where they might eat or purchase food. In addition, we'd like to learn more about how children feel about their local environments, and how this may affect the activities they do, or how and where they travel around their neighbourhood.

What will happen in this study?

If you and your child agree to participate in our project, your child will be asked to:

- 1. Complete the Healthy Neighbourhoods Survey for Youth. This survey primarily asks children about how they feel about their neighbourhood environments, the local facilities (such as parks) that they use for activities, places they may go to eat or buy food, and how they travel around their neighbourhood. Surveys usually take about 20-30 minutes to fill out and will be done in their classroom at a time decided by their teacher. (Note: students not filling out the survey will be given quiet activities by their teacher to do at their desks). Our research team will be on hand to help children fill out their surveys and to answer questions. All children will be given as much time as they need to complete the survey.
- 2. Wear two small pieces of equipment each day during the hours they are awake, for two 7-day periods about 6 months apart once during their Gr.5 or 6 year (Spring) and again in their Gr.6 or 7 year (in the Fall). The lightweight 'GPS Logger', worn on a collapsible neck strap, maps out the general places the child visits in the neighbourhood and they routes taken to get from place to place. The tiny 'Accelerometer', worn on a thin elastic belt around the student's waist (can be worn under clothes), is like a pedometer that counts steps but it can also tell how 'intense' the activity of the student is. These tools will help us to see patterns in children's neighbourhood activities and travel. Because students will wear the tools during two different weeks we can also better understand how children's activities change over time.
- Complete a short activity diary for each day they wear the 2 pieces of equipment, briefly telling us about their activities and any food purchases that day.
- 4. OPTIONAL If they would like, participating students can also meet together with the researchers and classmates for a group discussion to talk more about how they feel about their neighbourhood and how their local environment helps or prevents them from enjoying the recreational activities they like, or easily buying the foods they want, or travelling easily around the neighbourhood. There will be about 10-20 students in a group. The discussion will take place either at lunch recess or outside school hours, and will last about 1 hour. It will be held at the school or another community location. Participation is <u>completely voluntary</u>; a child can decide <u>not</u> to join in the group discussion and still be allowed to join in Steps 1-3.

As the child's parent/guardian, you will be asked to:

- OPTIONAL Complete the Healthy Neighbourhoods Survey for Parents. The survey asks many of the same questions as the Youth survey, as well as questions about your home and family and your own feelings about your neighbourhood. It usually takes about 15-25 minutes to fill out. The Parent Survey is <u>completely</u> <u>voluntary – your child can still join the study themselves even if you decide not to fill it out</u>. However, it gives us valuable information from the point of view of parents so we would really appreciate your participation.
- 2. OPTIONAL Parents of participating students will also have a chance to meet together with the researchers and other local parents for a group discussion about your neighbourhood environment and how it helps or prevents you and your family from doing the activities you like, buying the foods you need, or travelling easily around your neighbourhood. There will be about 8-12 parents in each group. The discussion will take place at a time outside school hours, and will last about 1-1.5 hours. It will be held at the school or another nearby community location. Participation in the group discussion is <u>completely voluntary</u>; a parent can decide <u>not</u> to participate and their child will still be allowed to participate in their own part of the study as outlined above.

Do we have to participate in this study?

Your participation in this study is completely voluntary. You and your child do <u>not</u> have to participate. You can each refuse to answer any survey questions, and can choose to leave the study at any time. Your decision will not affect your child's academic record in any way.

What are the benefits and risks if my child participates?

Recent research shows that our health is not only related to our personal lifestyle, such as the food we eat or physical activity we undertake, but also to the characteristics of the neighbourhood(s) we live in. This study will help us to better understand the links between our neighbourhoods, our activities, and our health. The results may help local municipal planners and school boards make decisions that will help plan healthier local communities.

There are no costs to you or your child for participating in this study. However, during each 7-day periods in which they participate, your child will receive \$2 each day from the research team when your child hands in their completed activity diary for the previous day and data from their equipment is collected. Your child will receive an <u>additional</u> \$1 on the last day when they return all their equipment. The total for EACH completed 7-day period is \$15.

The equipment in this study is easy to use, and the researchers will spend time with your child to make sure they understand how to use and care for the equipment. But, if any pieces of equipment break or become lost during the time your child is using them, we will give them a replacement unit right away without any cost to you or your child.

There may be risks to your child if he/she participates in this study. Getting tired or becoming disinterested in continuing with the project for the full 7 days are considered the largest risks. However, each piece of equipment weighs less than 60g (0.12 pounds) and should not be difficult for a child to carry. And a child can decide to quit the project at any time. The height and weight of your child will also need to be measured before they start in order to properly set up the accelerometer. These measurements will be taken in a private room at the child's school in the presence of a trusted adult (e.g. school nurse or teacher); no other children or people outside of the research team will be present. The equipment used to measure a child's weight also has no visible display - the measurements are automatically sent wirelessly to a laptop and so will not be visible to either your child or anyone else in the room.

There is little risk that you or your child will be identified or identifiable in any of the documents related to the study. All of the information collected in this study is kept strictly confidential. You and your child will be assigned a unique identification code – your name or personal information will not appear on any materials or data files. Also, materials and data files will ONLY be viewed by members of the research team and will be stored in a locked filing cabinet or on a password protected computer in a secure room at the University of Western Ontario. Parents and children who participate in the group discussions will be asked to keep everything they hear confidential and not to discuss it outside of the meeting. However, we cannot guarantee that confidentiality will be maintained by other participants in the focus group. Children can ask to see the maps of their own travel patterns and to change any information that feel is incorrect. However, to protect the privacy of each child, parents will not be able to view children's data or maps.

If you or your child decides to leave the study at any time (even up to 30 days AFTER the study has been completed), any of personal data collected from you or your child will be immediately destroyed and excluded from the study analysis.

You do not waive any of the legal rights you would otherwise have as a participant in a research study.

Follow Up

As the study involves a second round of participation this coming Fall (approximately 6 months after the first round this Spring), we may need to contact you at your home by phone or email in order to find out if your child changed schools between Spring and Fall. We would therefore ask that you include one or both of these pieces of information on the attached consent form.

Who do I contact if I have any other questions?

Should you have any questions or concerns about participating in this project, you can contact the lead researcher, Dr. Jason Gilliland, at the University of Western Ontario. Phone: (519) 661-2111 ext. 81239 or email: jgillila@uwo.ca

If you have any further questions regarding your rights as a study participant, please contact the Office of Research Ethics at 661-3036 or at: ethics@uwo.ca

Research Team

- Dr. Jason Gilliland, Department of Geography, University of Western Ontario
- Dr. William Avison, Department of Sociology, University of Western Ontario
- Dr. Harry Prapavessis, Department of Health and Rehabilitation Sciences, University of Western Ontario
- Dr. Paul Hess, Department of Geography and Planning, University of Toronto
- Dr. Kathy Speechley, Department of Paediatrics, University of Western Ontario
- Dr. Piotr Wilk, Department of Epidemiology, University of Western Ontario
- Dr. Colleen Gobert, Division of Food & Nutrition Sciences, Brescia University College

Mr. John Fleming, Director of Planning, City of London

This letter is for you to keep. Please return the attached Parent/Guardian consent form. You will also be given a copy of this consent form once it has been signed.



Examining the Influence of the Neighbourhood Environment on Children's Health and Well-Being

Parent / Guardian Consent Form

<u>Regardless</u> of whether you are consenting to let your child to participate in this study, we would ask that you return this form to school with your child, sealed in the envelope provided. Envelopes will be collected by your child's teacher. Thank you!

Consent: I, ______ (name of parent/guardian- please print), have read this letter and have been given the opportunity to ask questions. Any questions I had have been answered to my satisfaction. (Check all boxes that apply):

- □ I agree to participate by completing the *Healthy Neighbourhoods Survey for Parents* (*optional*; *if yes, please seal the survey in the envelope provided and return <u>with</u> signed consent form)*
- I am interested in being contacted about participating in a group discussion for parents
 (optional; if yes, please provide either phone or email contact information below)

Please select one of the following 2 options:

I agree to let my child ________ (child's full name – please print) participate in the full 14 days (two 7-day periods within the next 6-8 months) of the project as outlined. REQUIRED: My child has health issues which restrict their ability to walk/exercise or otherwise participate in this study YES NO

<u>OR</u> if your child is not interested in the full project but would still like to participate in the survey

I agree to let my child ________ (child's full name – please print) participate ONLY by way of completing the *Healthy Neighbourhoods Survey for Youth* (to be administered at child's school) rather than the full study.

Parent / Guardian's signature

Date

If your child IS participating, please provide a phone and/or email address (both is preferable) so that we may contact you this Fall to confirm whether or not your child has changed schools since the Spring. This information will be kept strictly confidential.

Parent/Guardian Email Address

Home or Cell Phone

Appendix F: Research Ethics Child Assent Form



How Healthy is the Environment in Your Neighbourhood?

Heffol We are researchers from the University of Western Ontario and we are doing a study in <u>your</u> neighbourhood! We need students in Grades 6, like you, to help us with this project!

What are we going to study?

We all know that getting lots of exercise and eating the right foods can help keep us healthy. We'd like to know if the places or services that you have and use in your neighbourhood also help to keep you healthy.

What would you have to do?

If you agree to be in the study there are 4 things we would like you to do:

1. Wear 2 small pieces of equipment every day for a week this Spring. A small GPS unit will help to make a map of all the places you visit every day. You would also wear a 'loonie'-sized piece of equipment on an elastic band around your waist that will tell us when you are doing physical activity, like running or playing sports. Both pieces of equipment are very light and easy to use. We will also come to your school every day in case you need help.



- 2. Fill out a short 1-page diary everyday about the activities you did that day.
- 3. Fill out a short survey on what you think about your neighbourhood. You will fill this out one day at school with your classmates. It takes about 20-30 minutes to finish but you can take as much time as you need.
- **4.** Then you would wear the equipment and fill out the diary <u>again</u> for a week later this Fall when you are in grade 7 (even if you are then going to a different school).

After both weeks are done, you could also join in a group discussion with some of your classmates to talk to us about where you like to go in your neighbourhood and the activities you like to do. You do not have to join in this group activity. The talk will take place at your school. We would like to audio record our talk.

To work some of the equipment we'll need to measure your height and weight. We'll do this in a private room at your school. Your teacher can be in the room. We won't share the information with anyone else.

Do you have to join this project?

No – you will only join if you would like to. You can also decide <u>at any time</u> that you would like to stop. We will never share your information with anyone else, even your parents, but you can ask to see it at any time. You can ALWAYS talk to your teacher or the researchers if you have any questions or worries.

I want to participate in this study!

If you would like to join this study in some way, choose one of the following two options:

I want to participate in the full 2 week stu	dy OR	I <u>only</u> want to complete the in-class survey
Print First and Last Name of Child		
Signature of Child Ag	e of Child	Date

Signature of Person Obtaining Assent

Date

Appendix G: Relevant Survey Questions from STEAM Project Assessing Perceptions used in this Research (2 pages)

J : Barriers to <u>activity</u> in your neighbourhood parks/playgrounds

Please tell us whether this stops you from going to a park/playground in your neighbourhood.	No	Sometimes No	Sometimes Yes	Yes
 It is too far from my house or takes too much time to get there 				
 There is no or not enough equipment or activities I like 				
 There is not enough room from the activities I like to do 				
4. There are no other kids to play with there				
5. There are no adults there to supervise				
 It feels unsafe there because of crime (ex: strangers, gangs, drugs) 				
7. I get bullied or teased when I go there				
8. I have nobody to go there with				
There are too many people there / feels too crowded				
10. There is too much garbage or graffiti				
11. Other reason?				

K : Streets in my neighbourhood

		I strongly disagree	I disagree a little bit	I agree a little bit	I strongly agree
1. The stre	ere are enough sidewalks on the eet in my neighbourhood.				
2. The neig to.	ere are walking trails in or near my ghbourhood that are easy to get				
3. The nea to <u>c</u>	ere are bicycle lanes or trails in or ir my neighbourhood that are easy get to.				
4. The stre	ere are lots of trees along the eets in my neighbourhood.				
5. I kr neig	now a lot of people in my ghbourhood.				

L : Safety in my neighbourhood

		I strongly disagree	I disagree a little bit	I agree a little bit	I strongly agree
1. There is so streets nea difficult or u	much traffic along <u>the</u> <u>r my home</u> that it is unpleasant to <u>walk</u> .				
2. There is so <u>streets nea</u> difficult to r <u>street</u>	much traffic along <u>the</u> <u>r my home</u> that it is ride my <u>bike or play on the</u>				
3. Most driver in our neigl	s go too fast while driving hbourhood.				
4. There is a l neighbourh	ot of crime in my ood.				
5. It feels uns around my <u>day</u> .	afe to walk <u>by myself</u> neighbourhood <u>during the</u>				
6. It feels uns <u>siblings</u> aro <u>during the</u>	afe to walk <u>with friends or</u> und my neighbourhood <u>day</u> .				
7. I am worrie <u>by myself</u> in local street being taker	ed about being or walking n my neighbourhood and s because I am afraid of n or hurt by a stranger				

Curriculum Vitae

Name:	Leah Taylor
Post-secondary Education and Degrees:	Western University London, Ontario, Canada 2016-2018 M.Sc. Geography
	Western University London, Ontario, Canada 2012-2016 BHSc (Honours Specialization, Health Promotion) <i>with Distinction</i>
Honours and Awards:	Children's Health Research Institute Fellowship 2017
	Western University Graduate Research Award 2018
	Canadian Institute for Health Research, Community Support Grant 2018
	London Health Research Day, Poster Competition Award 2018
Related Work Experience	Research Associate Human Environments Analysis Laboratory 2016 – Present
	Youth Review Committee Member Open Access Journal: Youth Engagement in Health Promotion University of Toronto 2015 – Present
	Teaching Assistant Western University 2016-2018
	Health Promotion Student: Chronic Disease and Injury Prevention Niagara Region Public Health 2009 – 2016
Conference Presentations:	Oral Presentation Poster Presentation Taylor, L.G., Clark, A.F., Wilk, P., & Gilliland, J.A. Assessing relationships between Ontario children's environments, experience of

barriers, and moderate-to-vigorous physical activity: a structural equation modelling approach. Annual meeting of the Canadian Associations of Geographers Université Laval, August 6-10, 2018

Poster Presentation

Context Matters: Examining children's perceived barriers to physical activity across varying Canadian environments. Annual meeting of the Canadian Associations of Geographers Université Laval, August 6-10, 2018

Oral Presentation

Context Matters: Examining children's perceived barriers to physical activity across varying Canadian environments. Canadian Public Health Association, Public Health 2018 Montreal QB, May 29, 2018

Poster Presentation

Context Matters: Examining children's perceived barriers to physical activity across varying Canadian environments. 2nd Place: Poster Competition Award London Health Research Day London ON, May 10, 2018

Oral Presentation

Context Matters: Examining children's perceived barriers to physical activity across varying Canadian environments. Annual meeting of the Canadian Associations of Geographers Ontario Division Queen's University, October 20-21, 2017